# INVESTIGATING MOLECULAR ADAPTATIONS IN ADIPOSE TISSUE AND SKELETAL MUSCLE IN RESPONSE TO INTERMITTENT FASTING AND EXERCISE TRAINING

by

# **ROBIN ALVIN WILSON**

MSc (Biochemistry)

A thesis submitted in fulfilment of the requirements for the degree of Doctor of Philosophy

College of Health and Biomedicine Victoria University Melbourne, Australia

June 2018

# ABSTRACT

The rapid rise in obesity prevalence appears to be a reflection of the changes in dietary and behavioural patterns, with eating habits shifting to higher consumption of energy-dense foods which are rich in fats and sugars, while at the same time, levels of physical activity are decreasing. These differences in energy intake and expenditure, often referred to as energy balance, have direct implications for weight regulation, with even small deviations in daily energy balance resulting in large body weight changes over the long term. Diet and exercise interventions aiming to shift the energy balance towards negative by either decreasing caloric intake and/or increasing physical activity have shown to be effective for weight loss. Many iterations of such dietary and physical activity interventions have been proposed, but intermittent fasting (IF) and high intensity interval training (HIIT) have recently been purported as effective strategies. Despite their effectiveness, the molecular mechanisms by which these lifestyle interventions induce their effects are unclear. Therefore, the purpose of this thesis was to examine the effects of IF and HIIT, alone and in combination, on anthropometric and metabolic health parameters in a model of diet-induced obese mice. To elucidate possible mechanisms of action, we investigated the impact of both lifestyle interventions on mRNA-miRNA regulatory networks, but more importantly, how such changes may translate into exerciseinduced and diet-induced improvements in body composition and metabolic health. The findings from the thesis demonstrate that intermittent fasting with or without high intensity interval training resulted in significantly less weight gain in male mice despite concurrently consuming a high fat and sugar diet. The reduced weight gain was predominantly in the form of lower fat mass accumulation, with no significant loss in lean mass. These observations were supported by enhanced expression of adipose tissue genes

relating to fragmentation of unilocular lipid droplets, lipolysis, fatty acid oxidation and efflux. Moreover, lower expression levels of leptin, pro-inflammatory markers and markers of hypoxia were also observed. These changes were also reflected by changes in miRNA-24, -222, -145 and -143. Within the skeletal muscle, the combination of diet and exercise demonstrated minimal impact on mRNA and miRNA expression markers relating to energy metabolism, however IF alone displayed significantly lower expression of all markers compared to control mice or other intervention groups. Interestingly, the changes in body composition, glycaemic control, lipid panels and expression of mRNA and miRNA seem to be gender specific with different responses, independent of intervention, demonstrated in male mice compared to the female mice. In conclusion, the novel results from this thesis have demonstrated superior effects on body composition and lipid profiles following combined IF and HIIT compared to either diet or exercise intervention alone while concurrently consuming a high fat and sugar diet. These observations are likely due to the physiological and biochemical changes that occur within the adipose and skeletal muscle tissue when creating a negative energy balance shift. The gender specific responses to the same diet and/or exercise intervention could indicate potential hormonal differences influencing metabolic control/adaptation in mice. Identification of important regulatory miRNAs through this thesis could provide potential therapeutic targets for obesity treatment and management.

# DECLARATION

"I, Robin Alvin Wilson, declare that the PhD thesis entitled "The effect of intermittent fasting and exercise training on body composition and molecular markers of adipose and skeletal muscle tissue physiology in diet-induced obese mice" is no more than 100,000 words in length including quotes and exclusive of tables, figures, appendices, bibliography, references and footnotes. This thesis contains no material that has been submitted previously, in whole or in part, for the award of any other academic degree or diploma. Except where otherwise indicated, this thesis is my own work".



28-06-2018

Signature

Date

## ACKNOWLEDGMENT

A huge thanks to my supervisor, Dr Matthew B. Cooke, for his judicious guidance, keen supervision, immense inspiration, invaluable criticism, thoughtful suggestions and friendly attitude throughout my PhD candidature. Special thanks for making this long distance supervision work.

Special thanks to my supervisory team, Professor Alan Hayes for his immense support throughout my PhD, especially with animal work, to Dr Christos Stathis for his invaluable criticism on manuscript writing and Dr Alba Moreno-Asso for help with molecular work.

I am very grateful to Victoria University for the support provided via the Victoria University International Postgraduate Scholarship and my supervisors for providing additional financial support to fund my overtly ambitious project. A big thanks to William Deasey and Anne Luxford, for their support with animal work and friendship, Dr Sarah Fraser, Stacey Lloyd, Charmaine DiQuattro, Mary Marshal, Tracy Murray, Dr Sudinna Hewakapuge, and Varsha Lal for their immense help and support with my Lab work. My gratitude to Ben Harvey, Danielle Debruin, Cara timpani and James Sorenson for their timely help at WCHRE or Werribee.

Thanks to Manpreet, Anthony and Swati for your great friendship, comically philosophical discussions and support. Special thanks to Neelu, Mini, Jagbir, Rishi, Hitesh, Loveleen, Deepika and Harshad for their mental and financial support to peruse this dream.

A humungous thanks to my Daddy and my sister for their love, trust and support. Couldn't ask for anything better.

V

# COMMUNICATIONS

#### PEER REVIEWED (PUBLISHED)

- Wilson, R. A., Deasy, W., Hayes, A. & Cooke, M. B. 2017. High fat diet and associated changes in the expression of micro-RNAs in tissue: Lessons learned from animal studies. Molecular Nutrition and Food Research, 2017 Jun; 61(6). doi: 10.1002/mnfr.201600943
- Wilson, R. A., Deasy, W., Stathis, C. G., Hayes, A. & Cooke, M. B. 2018. Intermittent Fasting with or without Exercise Prevents Weight Gain and Improves Lipids in Diet-Induced Obese Mice. Nutrients, Mar 12;10 (3). pii: E346. doi: 10.3390/nu10030346.

#### CONFERENCE ASTRACT

- Robin A. Wilson, William Deasy, Ben Harvey, Christos Stathis, Alan Hayes and Matthew Cooke. Effect of intermittent fasting and high intensity interval training on body composition and metabolic health in diet induced obese mice. Victorian obesity consortium symposium, 13<sup>th</sup> December 2016, Victoria, Australia.
- 2. Robin A. Wilson, William Deasy, Ben Harvey, Christos Stathis, Alan Hayes and Matthew Cooke. Effect of intermittent fasting and high intensity interval training on body weight and metabolic health markers in diet induced obese mice. Victoria University College of health and biomedicine postgraduate research student conference, 1<sup>st</sup> December 2016, Victoria, Australia.

# **CONFERENCE PRESENTATION**

	TITLE	AUTHORS	FORUM
1	Effect of intermittent	Robin A. Wilson,	Victoria University/
	fasting and high intensity	William Deasy,	University of Texas El
	interval training on body	Christos Stathis,	Paso Health Research
	composition and	Alan Hayes and	Symposium, 9 <sup>th</sup> February
	metabolic health in diet-	Matthew Cooke.	2017, Melbourne,
	induced obese mice.		Victoria, Australia. (Oral
			presentation)
2	Effect of intermittent	Robin A. Wilson,	Victorian obesity
	fasting and high intensity	William Deasy,	consortium symposium,
	interval training on body	Ben Harvey,	13 <sup>th</sup> December 2016,
	composition and	Christos Stathis,	Victoria, Australia. (Oral
	metabolic health in diet	Alan Hayes and	presentation)
	induced obese mice.	Matthew Cooke.	
3	Effect of intermittent	Robin A. Wilson,	Victoria University
	fasting and high intensity	William Deasy,	College of health and
	interval training on body	Ben Harvey,	biomedicine
	weight and metabolic	Christos Stathis,	postgraduate research
	health markers in diet	Alan Hayes and	student conference, 1 <sup>st</sup>
	induced obese mice.	Matthew Cooke.	December 2016,
			Victoria, Australia. (Oral
			presentation)

# **TABLE OF CONTENTS**

ABSTRACT	ii
DECLARATION	iv
ACKNOWLEDGEMENTS	v
COMMUNICATION	vi
CONFERENCE PRESENTATIONS	vii
TABLE OF CONTENTS	viii
LIST OF FIGURES	xiv
LIST OF TABLES	xxiii
LIST OF ABBREVIATIONS	XXV
CHAPTER 1: INTRODUCTION	1
1.1 Aims and hypothesis	3
1.2 Thesis Outline	4
CHAPTER 2 LITERATURE REVIEW	5
2.1 Causes of obesity	5
2.1.1 Environmental	6
2.1.2 Genetic and epigenetic	7
2.2 Effect of obesity on Adipose tissue	7
2.2.1 Formation of unilocular lipid droplet and adipose	8
tissue expansion	
2.2.2 Hypoxia	11
2.2.3 Extracellular matrix and cytoskeletal remodelling	12
2.2.4 Metabolic dysregulation (reduced oxidative capacity and	13
insulin sensitivity)	
2.2.4.1 Lipogenesis	13
2.2.4.2 Lipolysis	15
2.2.4.3 Lipid handling	16
2.2.5 Inflammation	17
2.2.6 Endocrine changes	18
2.3 Effect of obesity on Muscle tissue	21
2.3.1 Fibre type switching	22
2.3.2 Incomplete Fat oxidation and insulin resistance	24
2.3.3 Insulin signalling	25
2.3.4 Master regulators-PGC-1α	26
2.3.5 Metabolic sensors- AMPK and SIRT1	28
2.3.6 Inflammation	31
2.4 MicroRNA	34
2.4.1 Biogenesis of miRNA	35
2.4.2 Regulation of gene expression by miRNA	37
2.4.2.1 mRNA Degradation	37
2.4.2.2 Blocking translation initiation	37
2.4.2.3 Translocation to processing bodies	38
2.4.3 MiRNA, energy homeostasis and obesity	38
2.4.4 miRNAs: the effect of diet and exercise	40
2.5 Lifestyle interventions	43
2.5 Lifestyle interventions	43

2.5.1 Intermittent Fasting	43
2.5.1.1 Effect of IF on Body composition	45
2.5.1.2 Effect of IF on glucose homeostasis	46
2.5.1.3 Effect of IF on lipid profiles	48
2.5.1.4 Molecular changes induced by IF	49
2.5.2 High Intensity Interval Training	50
2.5.2.1 Effect of HIIT on Body composition	51
2.5.2.2 Effect of HIIT on glucose homeostasis	53
2.5.2.3 Effect of HIIT on lipid profiles	54
2.5.2.4 Molecular changes induced by HIIT	54
CHAPTER 3 MATERIAL AND METHODS	57
3.1 Overview	57
3.2 Animals	57
3.2.1 Procurements and Housing:	57
3.2.2 Diet	59
3.2.3 Assessment of Caloric Intake	60
3.3 Interventions	60
3.3.1 Intermittent Fasting Protocol	60
3.3.2 High Intensity Interval Training Protocol	61
3.3.2.1 Acclimatization to treadmill exercise	61
3.3.2.2 Top Speed Test	62
3.3.2.3 HIIT Training	62
3.4 Body Weight and Composition Assessment	63
3.5 Culling and tissue collection	63
3.6 Isolation of RNA	64
3.6.1 Assessment of RNA quality	65
3.7 mRNA Reverse transcription and assessment of gene expression	66
3.7.1 Synthesis of cDNA-reverse transcription	66
3.7.2 Gene expression	66
3.8 MiRNA Reverse transcription and assessment of miRNA	67
expression	
3.8.1 Synthesis of cDNA-reverse transcription	67
3.8.2 MiRNA expression	67
3.9 Statistical Analysis	68
CHAPTER 4- INTERMITTENT FASTING WITH OR WITHOUT	69
EXERCISE PREVENTS WEIGHT GAIN AND IMPROVES	
LIPIDS IN DIET-INDUCED OBESE MICE	
4.1 Introduction	69
4.2 Materials and methods	72
4.2.1 Animals	72
4.2.2 Assessment of Caloric Intake	72
4.2.3 Interventions	72
4.2.4 Body Weight and Composition Assessment	72
4.2.5 Assessment of Glucose Tolerance (AUC)	73
4.2.6 Blood and organ collection	73
4.2.7 Assessment of Plasma Insulin, TAG, LDL and HDL	74
Levels	
4.2.7.1 Plasma insulin	74

4.2.7.2 Triglycerides	74	
4.2.7.3 Low density lipoproteins	75	
4.2.7.4 High density lipoproteins		
4.2.8 Homeostatic Model Assessment of Insulin Resistance	75	
(HOMA-IR)		
4.2.9 Assessment of Muscle Strength	76	
4 2 10 Statistical analysis	77	
4 3 Results	78	
4.3.1 Effects of 12-weeks of the HE/S diet on body	78	
weight of all animals	70	
4.3.2 Effects of IE and/or HIIT on body weight and	79	
composition	1)	
4.3.2.1 Rody weight and fat mass	81	
4.3.2.1 body weight and fat mass	01 Q1	
4.3.2.2 Lean mass	01	
4.5.5 Effects of IF on caloric linake	82 82	
4.5.4 Effect of IF and/or HIT on blood glucose, plasma	83	
insuin, HOMA-IR and AUC	02	
4.3.4.1 Fasting blood glucose	83	
4.3.4.2 Plasma Insulin	83	
4.3.5 Insulin Resistance (HOMA-IR) and Glucose	85	
Tolerance (AUC)		
4.3.6 Effect of IF and/or HIIT on lipid profiles	88	
4.3.7 Effect of IF and/or HIIT on muscle strength	89	
4.3.8 Effect of IF and/or HIIT on organ weight	90	
4.4 Discussion	91	
4.4.1 Effect of IF and/or HIIT on body composition	92	
4.4.2 Effect of IF and/or HIIT on glucose homeostasis	96	
4.4.3 Effect of IF and/or HIIT on plasma lipids	97	
4.4.4 Effect of IF and/or HIIT on muscle strength	98	
4.4.5 Effect of IF and/or HIIT on organ weight	98	
4.4.6 Sexually dimorphic response to IF and/or HIIT	99	
4.5 Conclusion	100	
CHAPTER 5- MOLECULAR BASIS OF REDUCED FAT MASS	102	
AND ASSOCIATED BENEFITS IN ADIPOSE TISSUE INDUCED		
BY INTERMITTENT FASTING WITH OR WITHOUT HIGH		
INTENSITY INTERVAL TRAINING IN DIET-INDUCED OBESE		
MICE		
5.1 Introduction	102	
5.2 Materials and methods	105	
5.2.1 Animals	105	
5.2.2 Culling and tissue collection	105	
5.2.3 Adipose tissue gene and miRNA expression	105	
5.2.4 Statistical analysis	107	
5.3 Results	108	
5.3.1 The effect of 12 weeks of IF and/or HIIT on WAT gene	108	
expression related to structure and function		
5.3.1.1 CIDEC	108	
5.3.1.2 Leptin	109	
	= 0 /	

5.3.1.3 PPARy	110
5.3.1.4 FOXO1	111
5.3.2 The effect of 12 weeks of IF and/or HIIT on WAT gene	112
expression related to Fatty acid oxidation and	
trafficking	
5.3.2.1 HADH	112
5.3.2.2 FABP4	113
5.3.2.3 Hormone sensitive lipase	114
5.3.3 The effect of 12 weeks of IF and/or HIIT on WAT gene	115
expression of SIRT1	
5.3.4 The effect of 12 weeks of IF and/or HIIT on WAT gene	116
expression related to inflammation	
5.3.4.1 TRPV4	116
5.3.5 The effect of 12 weeks of IF and/or HIIT on WAT gene	117
expression related to hypoxia	
5.3.5.1 HIF1α	117
5.3.6 The effect of 12 weeks of IF and/or HIIT on WAT gene	118
expression related to white adipose tissue beiging	
5.3.6.1 UCP1	118
5.3.7 The effect of 12 weeks of IF and/or HIIT on miRNA	119
expression relating to adipogenesis and glucose	
handling in adipose tissue	110
5.3.7.1  miR-2/a	119
5.3.7.2 miR-24	120
5.3.7.3 miR-222	121
5.3./.4 miR-143	122
5.3.7.3 miR-145	123
5.3.8 Correlation analysis between mRINA and	124
miRNA expression	106
5.4 Discussion $5.4.1$ A dipose tissue structure and function	120
5.4.2 Eatty acid oxidation and trafficking	120
5.4.2 Adipose tissue SIPT1 expression and systemic insulin	120
sensitivity	150
5.4.4 Inflammation	131
5.4.5 Hypoxia	131
5.4.6 Adjpose tissue browning	132
5.4.7 Expression of miRNAs	133
5.5 Conclusion	135
CHAPTER 6- INTERMITTENT FASTING AND HIGH	136
INTENSITY INTERVAL TRAINING DISPLAY OPPOSING	
EFFECTS ON MARKERS OF SKELETAL MUSCLE OXIDATIVE	
METABOLISM, INSULIN SIGNALLING, INFLAMMATION,	
AND ATROPHY IN DIET-INDUCED OBESE MICE	
6.1 Introduction	136
6.2 Materials and methods	
6.2.1 Animals	138
6.2.2 Culling and tissue collection	139

6.2.3 Myosin ATPase Fibre typing	139
6.2.4 Skeletal muscle gene and miRNA expression	140
6.2.5 Statistical analysis	141
6.3 Results	143
6.3.1 The effect of IF and/or HIIT on skeletal muscle mass	143
and morphology	
6.3.1.1 Muscle weight	143
6.3.1.2 Muscle fibre type	144
6.3.2 The effect of IF and/or HIIT on gene expression of AMPK-SIRT1-PGC1α axis	146
6.3.2.1 AMPK	146
6.3.2.2 SIRT1	147
6.3.2.3 PGC1α	148
6.3.3 The effect of IF and/or HIIT on gene expression related	149
to oxidative metabolism	
6.3.3.1 CPT1	149
6.3.3.2 Citrate Synthase	150
6.3.3.3 COX-IV	151
6.3.3.4 UCP3	152
6.3.4 The Effect of IF and/or HIIT on gene expression related	153
to Inflammation	
6.3.4.1 TNF	153
6.3.5 The effect of IF and/or HIIT on gene expression related	154
to Insulin signalling	
6.3.5.1 AS160	154
6.3.6 The effect of IF and/or HIIT on gene expression related	155
to muscle atrophy	
6.3.6.1 MAFbx	155
6.3.6.2 MURF1	156
6.3.7 The effect of IF and/or HIIT on expression of myomiRs	157
6.3.7.1 miR-696	157
6.3.7.2 miR-133a	158
6.3.7.3 miR-133b	159
6.3.8 Correlation analysis between mRNA and miRNA expression	160
6.4 Discussion	161
6.4.1 AMPK-SIRT1-PGC1α signalling axis	161
6.4.2 Oxidative metabolism	164
6.4.3 Insulin signalling	167
6.4.4 Inflammation	167
6.4.5 Muscle atrophy	168
6.4.6 Muscle mass and fibre type	169
6.4.7 MyomiR expression	170
6.5 Conclusion	172
CHAPTER 7- CONCLUSION, LIMITATIONS AND FUTURE	173
7 1 Very findings	170
1.1 Key maings	1/3

7.1.1 Combination of IF and HIIT prevents weight gain by less accumulation of fat mass and maintenance of lean mass	173
7.1.2 Combination of IF with HIIT leads to less fat accumulation due enhanced fatty acid oxidation and efflux.	174
7.1.3 Intermittent fasting and high intensity exercise had opposite effect on skeletal muscle molecular markers	175
7.1.4 Intermittent fasting perturbed the systemic glucose homeostasis	176
7.1.5 Sexually dimorphic response to IF and/or HIIT	176
7.2 Limitations	176
7.3 Future directions	177
REFERENCES	179

No	Figure title	Page no
Figure 2.1.	Obesity by age, United States, 1971-1974 through 2005-2006	6
	(CDC, 2008).	
Figure 2.2.	Mouse epididymal white adipose tissue (10x -H&E stained).	8
Figure 2.3.	Transverse section of soleus muscle showing type 1 (dark) and	22
	type 2a (light) fibres.	
Figure 2.4.	Insulin-signalling pathway for insulin-stimulated GLUT4	25
	translocation and glucose transport in skeletal muscle (Cartee, 2015).	
Figure 2.5.	Proposed mechanism of PGC1a activation by cooperation	31
	between AMPK and SIRT (Ruderman et al., 2010).	
Figure 2.6.	MiRNA biogenesis pathway. MiRNAs are transcribed by RNA	36
	polymerase II resulting in pri-miRNA transcripts. Pri-miRNAs	
	are processed by Drosha leaving Pre-miRNAs which are	
	exported from the nucleus via Exportin5. Pre-miRNAs are	
	processed by Dicer, the mature miRNA is then permanently	
	incorporated into the RNA-induced silencing complex and will	
	bind to the 3'-UTR of a target mRNA (Peng et al., 2014).	
Figure 2.7.	An example of a HIIT protocol (Cassidy et al., 2017).	50
Figure 3.1.	General outlay of the study.	58
Figure 3.2.	Animal diets-(A) chow diet and (B) High fat diet.	59
Figure 3.3.	Mice treadmill (A), Treadmill control (B), mouse on treadmill	61
	(C).	
Figure 3.4.	EchoMRI machine (A) and mouse tunnel (B).	63
Figure 3.5.	Representative image of agarose gel showing two rRNA bands.	65
Figure 4.1.	Grip strength apparatus (A) mouse performing grip test (B).	76

# **LIST OF FIGURES**

- Figure 4.2. Weight gain (g) over the 12-weeks high fat/sugar diet feeding 77 period in (A) male and (B) females. Data is presented as mean of each group and error bars represent standard deviation.
- Figure 4.3. Change in body weight over the 10-weeks intervention period 80 in (A) male and (B) females. Data is presented as mean of each group and error bars represent standard deviation.
- Figure 4.4. Change in body composition (g) in males (A), Females (B), after 81
  10-weeks of intervention. Bars indicate means and error bars represent standard deviation. \* significant change over time, €
  significantly different from CON, # significantly different from HIIT, § significantly different from IF, ‡ significantly different from IF+HIIT.
- Figure 4.5. Caloric intake over the 12-weeks intervention period in (A) 82 males and (B) females. Data is presented as mean of each group and error bars represent standard deviation.
- Figure 4.6. Change in (A) fasting glucose and (B) insulin over the 10 or 84 12-weeks period in males and females. Bars represent means, error bars represent standard deviation and \* indicates significant change over time.
- Figure 4.7. HOMA-IR changes from 'obese' baseline in males and females 85 after 12-weeks of diet and/or exercise intervention. Bars represent means, error bars represent standard deviation and \* indicates significant change over time.
- Figure 4.8. Glucose tolerance curves (A,C,E,G) and glucose are under the 86 curve (B,D,F,H) of different intervention groups in males before and after 10-weeks of intervention. Bars represent means, error bars represent standard deviation and \* indicates significant change over time.
- Figure 4.9. Glucose tolerance curves (A,C,E,G) and glucose are under the 87 curve (B,D,F,H) of different intervention groups in females before and after 10-weeks of intervention. Bars represent

means, error bars represent standard deviation and \* indicates significant change over time.

- Figure 4.10. Change in glucose area under the curve (AUC) in males and 88 females after 10-weeks of diet and/or exercise intervention.
  Bars represent means, error bars represent standard deviation and \* indicates significant change over time.
- Figure 5.1. Effect of IF, HIIT and IF+HIIT on *CIDEC* gene expression 108 (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *B2M* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*) and (\*\*) brackets represent significant difference between groups at p<0.05 and p<0.01 level of significance respectively.</li>
- Figure 5.2. Effect of IF, HIIT and IF+HIIT on *Lep* gene expression 109 (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *B2M* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*) and (\*\*) brackets represent significant difference between groups at p<0.05 and p<0.01 level of significance respectively.</li>
- Figure 5.3. Effect of IF, HIIT and IF+HIIT on *PPARγ* gene expression 110 (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *B2M* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*) bracket represent significant difference between groups at p<0.05 level of significance.</li>
- Figure 5.4. Effect of IF, HIIT and IF+HIIT on *FOXO1* gene expression 111 (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *B2M* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*\*) brackets represent significant difference between groups at p<0.01 level of significance.</li>

- Figure 5.5. Effect of IF, HIIT and IF+HIIT on *HADH* gene expression 112 (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *B2M* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*) and (\*\*) brackets represent significant difference between groups at p<0.05 and p<0.01 level of significance respectively.</li>
- Figure 5.6. Effect of IF, HIIT and IF+HIIT on *FABP4* gene expression 113 (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *B2M* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*\*) brackets represent significant difference between groups at p<0.01 level of significance.</li>
- Figure 5.7. Effect of IF, HIIT and IF+HIIT on HSL gene expression 114 (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to B2M gene expression. Bars represent means, error bars represent standard deviation.
- Figure 5.8. Effect of IF, HIIT and IF+HIIT on SIRT1 gene expression 115 (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to B2M gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*\*) brackets represent significant difference between groups at p<0.01 level of significance.</li>
- Figure 5.9. Effect of IF, HIIT and IF+HIIT on *TRPV4* gene expression 116 (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *B2M* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*\*) brackets represent significant difference between groups at p<0.01 level of significance.</li>
- Figure 5.10. Effect of IF, HIIT and IF+HIIT on  $HIF1\alpha$  gene expression 117 (mRNA fold change) relative to CON group in (A) males and

(B) females. Gene (mRNA) expression is normalized to *B2M* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*\*) bracket represent significant difference between groups at p<0.01 of significance.

- Figure 5.11. Effect of IF, HIIT and IF+HIIT on UCP1 gene expression 118 (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to B2M gene expression. Bars represent means, error bars represent standard deviation.
- Figure 5.12. Effect of IF, HIIT and IF+HIIT on miR-27a expression (fold 119 change) relative to CON group in (A) males and (B) females.MiRNA expression is normalized to *RNU6* expression. Bars represent means, error bars represent standard deviation.
- Figure 5.13. Effect of IF, HIIT and IF+HIIT on miR-24 expression (fold 120 change) relative to CON group in (A) males and (B) females.
  MiRNA expression is normalized to *RNU6* expression. Bars represent means, error bars represent standard deviation.
  Asterisk (\*\*) brackets represent significant difference between groups at p<0.01 level of significance.</li>
- Figure 5.14. Effect of IF, HIIT and IF+HIIT on miR-222 expression (fold 121 change) relative to CON group in (A) males and (B) females.
  MiRNA expression is normalized to *RNU6* expression. Bars represent means, error bars represent standard deviation.
  Asterisk (\*\*) bracket represent significant difference between groups at p<0.01 level of significance.</li>
- Figure 5.15. Effect of IF, HIIT and IF+HIIT on miR-143 expression (fold 122 change) relative to CON group in (A) males and (B) females.
  MiRNA expression is normalized to *RNU6* expression. Bars represent means, error bars represent standard deviation.
  Asterisk (\*\*) brackets represent significant difference between groups at p<0.01 level of significance.</li>

- Figure 5.16. Effect of IF, HIIT and IF+HIIT on miR-145 expression (fold 123 change) relative to CON group in (A) males and (B) females.
  MiRNA expression is normalized to *RNU6* expression. Bars represent means, error bars represent standard deviation.
  Asterisk (\*) and (\*\*) brackets represent significant difference between groups at p<0.05 and p<0.01 level of significance respectively.</li>
- Figure 5.17. Correlation between (A) HIF1α and miR-222a in HIIT group 125 (R=-0.68, p<0.05) and (B) FOXO1 and miR-145a in males from IF+HIIT group (R=-0.81, p<0.016) in males.</p>
- Figure 6.1. Effect of IF, HIIT and IF+HIIT on weight (mg) of soleus 143 muscle in (A) males and (B) females and EDL muscle in (C) males and (D) females. Bars represent means, error bars represent standard deviation.
- Figure 6.2. Effect of IF, HIIT and IF+HIIT on soleus muscle type I fibre 144 in (A) males and (B) females and type IIa fibre in (C) males and (D) females. Bars represent means, error bars represent standard deviation.
- Figure 6.3. Representative images of (A) soleus and (B) EDL muscle 145 showing fibre type distribution. Effect of IF, HIIT and IF+HIIT on ED muscle type IIa fibre in (C) males and (D) females and type IIb fibre in (E) males and (F) females. Bars represent means, error bars represent standard deviation.
- Figure 6.4. Effect of IF, HIIT and IF+HIIT on AMPK gene expression 146 (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *ppia* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*) bracket represent significant difference between group at p<0.05 level of significance.</li>
- Figure 6.5. Effect of IF, HIIT and IF+HIIT on *SIRT1* gene expression 147 (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *ppia*

gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*\*) bracket represent significant difference between group at p<0.01 level of significance.

- Figure 6.6. Effect of IF, HIIT and IF+HIIT on PGC1α gene expression 148 (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *ppia* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*) bracket represent significant difference between group at p<0.05 level of significance.</li>
- Figure 6.7. Effect of IF, HIIT and IF+HIIT on *CPT1* gene expression 149 (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *ppia* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*\*) bracket represent significant difference between group at p<0.01 level of significance.</li>
- Figure 6.8. Effect of IF, HIIT and IF+HIIT on citrate synthase (CS) gene 150 expression (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *ppia* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*) bracket represent significant difference between group at p<0.05 level of significance.</li>
- Figure 6.9. Effect of IF, HIIT and IF+HIIT on COX-IV gene expression 151 (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *ppia* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*) bracket represent significant difference between group at p<0.05 level of significance.</li>
- Figure 6.10. Effect of IF, HIIT and IF+HIIT on UCP3 gene expression 152 (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *ppia* gene expression. Bars represent means, error bars represent

standard deviation. Asterisk (\*) and (\*\*) bracket represent significant difference between group at p<0.05 and p<0.01 level of significance respectively.

- Figure 6.11. Effect of IF, HIIT and IF+HIIT on *TNFα* gene expression 153 (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *ppia* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*\*) bracket represent significant difference between group at p<0.01 level of significance.</li>
- Figure 6.12. Effect of IF, HIIT and IF+HIIT on AS160 gene expression 154 (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *ppia* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*\*) bracket represent significant difference between group at p<0.01 level of significance.</li>
- Figure 6.13. Effect of IF, HIIT and IF+HIIT on *MAFbx* gene expression 155 (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *ppia* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*\*) bracket represent significant difference between group at p<0.01 level of significance.</li>
- Figure 6.14. Effect of IF, HIIT and IF+HIIT on *MURF1* gene expression 156 (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *ppia* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*) and (\*\*) bracket represent significant difference between group at p<0.05 and p<0.01 level of significance respectively.</li>
- Figure 6.15. Effect of IF, HIIT and IF+HIIT on miR-696 expression (fold 157 change) relative to CON group in (A) males and (B) females.
  MiRNA expression is normalized to *RNU6* expression. Bars represent means, error bars represent standard deviation.

Asterisk (\*) bracket represent significant difference between group at p<0.05 level of significance.

- Figure 6.16. Effect of IF, HIIT and IF+HIIT on miR-133a expression (fold 158 change) relative to CON group in (A) males and (B) females.
  MiRNA expression is normalized to *RNU6* expression. Bars represent means, error bars represent standard deviation.
  Asterisk (\*) bracket represent significant difference between group at p<0.05 level of significance.</li>
- Figure 6.17. Effect of IF, HIIT and IF+HIIT on miR-133b expression (fold 159 change) relative to CON group in (A) males and (B) females.
  MiRNA expression is normalized to *RNU6* expression. Bars represent means, error bars represent standard deviation.
  Asterisk (\*) bracket represent significant difference between group at p<0.05 level of significance.</li>

LIST	OF	TABI	LES

Table No	Table title	Page No.
Table 2.1.	Factors secreted by adipose tissue into the bloodstream and	20
	respective function/effect in their targets	
Table 3.1.	Calculated nutritional parameters of diets	59
Table 4.1.	The effect of 12-weeks of intermittent fasting and high	89
	intensity exercise combined or individual on plasma TAG,	
	HDL and LDL in male and female mice	
Table 4.2.	The effect of 10-weeks of intermittent fasting and high	90
	intensity exercise combined or individual on absolute and	
	relative muscle strength in both male and female mice	
Table 4.3.	The effect of 12-weeks of intermittent fasting and high	91
	intensity exercise combined or individual on liver and	
	pancreas weight in both male and female mice.	
Table 5.1.	List of targeted genes in adipose tissue with full official	106
	names, NCBI reference sequence numbers and Qiagen	
	catalogue number	
Table 5.2.	List of miRNAs analysed in adipose tissue, miRbase	106
	accession numbers and Qiagen miScript Primer Assay	
	Catalog Numbers	
Table 5.3.	Correlation between mRNA and miRNA expression in	124
	skeletal muscle in males.	
Table 5.4.	Correlation between mRNA and miRNA expression in	124
	skeletal muscle in females.	
Table 6.1.	List of targeted genes in skeletal muscle with full official	141
	names, NCBI reference sequence numbers and Qiagen	
	catalogue number	
Table 6.2.	List of miRNAs analysed in skeletal muscle, miRbase	141
	accession numbers and Qiagen miScript Primer Assay	
	Catalog Numbers	

- Table 6.3.Correlation between mRNA and miRNA expression in160skeletal muscle in males.
- Table 6.4.Correlation between mRNA and miRNA expression in160skeletal muscle in females.

# LIST OF ABBREVIATIONS

ADF	Alternate day fasting
AMP	Adenosine monophosphate
AMPK	Protein kinase AMP-activated catalytic subunit alpha 2
ANOVA	Analysis of variance
AS160	Akt substrate of 160 kDa
ATGL	Adipose tissue triglyceride lipase
ATP	Adenosine triphosphate
AUC	Area under the curve
B2M	Beta-2 microglobulin
BMI	Body mass index
CaCl <sub>2</sub>	Calcium chloride
cDNA	Complementary deoxyribonucleic acid
CIDE	Cell Death Inducing DFFA like effector
CIDEC	Cell death-inducing DFFA-like effector c
CoCl <sub>2</sub>	Cobalt chloride
CON	Control group
COX-IV	Cytochrome c oxidase subunit 4 isoform 2
CPT1	Carnitine palmitoyltransferase Ib
CR	Calorie restriction
CS	Citrate synthase
CT	Cycle threshold values
DNA	Deoxyribonucleic acid
DPX	Dibutylphthalate Polystyrene Xylene
ECM	Extracellular matrix
EDL	Extensor digitorum longus
EDTA	Ethylenediaminetetraacetic acid
ELISA	Enzyme linked immunosorbent assay
FABP4	Fatty acid binding protein 4, adipocyte
FABPs	Fatty acid binding proteins
FAT/CD36	Fatty acid translocase

Fatty acid transport proteins
Free fatty acid
Forkhead box O1
Fat specific protein 27
Glucose transporter type 4
Genome wide association studies
Hydroxyacyl-Coenzyme A dehydrogenase
High density lipoproteins
High fat-high sugar
High fat diet
High intensity exercise
Hypoxia inducible factor 1, alpha subunit
High intensity interval training
Homeostatic model assessment
Hormone sensitive lipase
Heat shock protein 70
Intermittent fasting
Intermittent fasting with High intensity interval training
Insulin like growth factor-1
Intraperitoneal glucose tolerance test
Interleukin-6
Inter-myocellular or intermuscular adipose tissue
insulin receptor substrate
c-JUN N-terminal kinase
c-Jun N-terminal kinase
Low density lipoproteins
Leptin
Lipase, hormone sensitive
F-box protein 32
Macrophage chemo attractant protein-1
microRNA
MicroRNA

mTOR	Mammalian target of rapamycin
MuRF1	Muscle RING finger 1
MyHC	Myosin heavy chain
NADH	Nicotinamide adenine dinucleotide
NF-ĸB	Nuclear factor-kappa B
OBC	Obese baseline control
OCT	Optimal cutting temperature compound
PGC1a	Peroxisome proliferator-activated receptor gamma coactivator $1$ - $\alpha$
РКА	Protein kinase A
PMAT	Perimuscular adipose tissue
PPARγ	Peroxisome proliferator-activated receptor-y
PPARγ	Peroxisome proliferator activated receptor gamma
Ppia	Peptidylprolyl isomerase A
Pri miRNA	Primary miRNA
RNA	Ribonucleic acid
rRNA	Ribosomal ribonucleic acid
RT- PCR	Real time polymerase chain reaction
SD	Standard deviation
Sec	Second
SIRT1	Sirtuin 1
SIRT1	Sirtuin (silent mating type information regulation 2 homolog) 1
SOD	Superoxide dismutase
SREBP1	Sterol regulatory element-binding protein 1
TAE	Tris-acetate-EDTA buffer
TAG	Triacylglyceride
TNFα	Tumour necrosis factor-a
TNFα	Tumor necrosis factor-alpha
TRIS-HCl	Trisaminomethane hydroxychloride
TRPV4	Transient receptor potential cation channel, subfamily V, member 4
UCP1	Uncoupling protein 1 (mitochondrial, proton carrier)
UCP3	Uncoupling protein 3
VEGF	Vascular endothelial growth factor

VLDL Very low density lipoproteins

WAT White adipose tissue

# **CHAPTER 1**

#### INTRODUCTION

Obesity is a growing health issue around the globe and has recently been classified as a disease by the American Medical Association (Kushner, 2014). Since 1975, global obesity has approximately tripled, with more than 1.9 billion adults (18 years and older) currently classified as overweight and obese. Moreover, more than 41 million children below the age of 5 years and 340 million children and adolescents aged 5-19, are classified as overweight or obese in 2016. Overall, 13% of the world's adult population is obese (WHO 2017). In Australia, 63% (11.2 million) of population aged 18 and above was overweight or obese in 2014-15. Approximately, 36% (1 in 3 adults) and ~28% (1 in 4 adults) of population is overweight and obese respectively (Australian Institute of Health and Welfare 2017). Such figures impose a substantial economic burden on the Australian health care system. Obesity is associated with various comorbidities such as type 2 diabetes, hypertension, dyslipidemia, and certain types of cancers (Kim et al., 2014a) and the incidence of such diseases have also risen worldwide.

Although genetics, physiological and environmental factors contribute to weight gain and obesity (Wright and Aronne, 2012), the combination of high calorie intake and low physical activity, is a major contributing factor (Meldrum et al., 2017, Moore et al., 2012). Minor deviations in energy balance as a result of high energy intake and low energy expenditure can manifests itself into large increases in body weight over time (Romieu et al., 2017). Increased weight is also accompanied by various other changes at systemic level and cellular level in most of the metabolically important tissues like adipose, skeletal muscle, liver, brain etc. (Kim, 2016). Obesity alters various regulatory molecular signals at transcriptional and translational levels which contributes to the changes at systemic level. Recent addition to these molecular signalling molecules is microRNAs which regulate various cellular functions at transcriptional or translational levels (Cui et al., 2017).

Lifestyle interventions aimed at shifting the energy balance towards negative side are the first line of defence for prevention and cure of obesity (Wilson et al., 2018). Intermittent fasting (IF) has recently gained attention, supported by several human and animal studies (Patterson and Sears, 2017). Intermittent fasting refers to fasting (100-75%) reduced caloric intake) for 2-3 alternate days in a week with ad libitum food intake on non-fasting days (Mattson et al., 2018). Intermittent fasting has been suggested to be effective not only for improving body composition but also improves glucose homeostasis, plasma lipid profiles (Patterson and Sears, 2017) and cognitive function (Shojaie et al., 2017). High intensity interval training (HIIT) has gained popularity as an alternative to moderate intensity exercise, which is currently recommended by American college of sports medicine (ACSM) for weight loss (Wewege et al., 2017). High intensity interval training involves, small bouts of high intensity exercise interspersed by active rest and is considered to improve body composition, cardiopulmonary fitness (Türk et al., 2017), reduces plasma lipids (Ouerghi et al., 2017) and improves glucose homeostasis (Batacan et al., 2017). This mode of exercise is much less time intensive and is therefore more preferred by clinical population (Ribeiro et al., 2017).

Combination of dietary and exercise interventions has been reported to be more effective than either of them alone (Foster-Schubert et al., 2012). Few studies have reported the effects of IF or HIIT individually on obese human population or animal models. However, combined effect of these two less intensive but efficient interventions has not been explored and warrants investigation. In addition, very few studies have reported the effects of IF or HIIT in both males and female cohorts and it is unknown if there is any sexual dimorphic response to these lifestyle interventions. Moreover, given changes in body composition reported by previous studies are mainly due to changes in adipose tissue mass, very few studies, both on IF and HIIT, have investigated the wide range of molecular changes at mRNA and especially at miRNA level in adipose tissue that underpin the systemic changes induced by these interventions. Since obesity has various detrimental effects on skeletal muscle function, it is important to know if IF and/or HIIT can reverse those damages. Finally, diet restriction has been reported to cause loss in lean mass (Heymsfield et al., 2014). It is unknown how IF affects molecular markers of muscle atrophy or growth and such knowledge will provide a foundation for its application in future research and clinical population.

### **1.1 AIMS AND HYPOTHESIS**

The aims and hypothesis of the thesis are:

- To investigate the effect of combining IF and HIIT on changes in body composition, systemic glucose homeostasis and plasma lipids in diet induced obese mice. It is hypothesised in this study that combination of IF with HIIT will have synergistic effects on these anthropometric or physiological parameters.
- 2. To explore molecular changes at mRNA and miRNA expression level induced by IF and/or HIIT on structural and functional aspects of adipose tissue like unilocular lipid droplet formation, fatty acid oxidation and trafficking, inflammation, hypoxia and beiging. It has been hypothesised that IF with HIIT will synergistically reduce fat accumulation by increasing fatty acid oxidation and efflux, prevent/reduce obesity induced inflammation and hypoxia.

3. To examine changes in fibre type distribution and molecular changes at mRNA expression levels induced by IF and/or HIIT related to skeletal muscle oxidative metabolism, insulin signalling, energy sensing molecules and muscle atrophy as well as alterations in muscle specific miRNAs. It is hypothesised that IF and/or HIIT will enhance oxidative metabolism, improve insulin signalling, energy sensing of energy sensing molecules and reduce inflammation and muscle atrophy.

#### **1.2 THESIS OUTLINE**

Chapter 2 provides the review of current scientific knowledge on the causes of obesity, its manifestations on adipose tissue and muscle as both of them are considered important for maintenance of systemic energy homeostasis. Also discussed, how lifestyle intervention, IF and HIIT, impact systemic and cellular homeostasis in obesity. A part of the review talks about miRNAs, from basic synthesis to their role in regulating diet and exercise induced changes at cellular level. Chapter 3 details the general materials and methods used in this thesis. Chapter 4 reports the effects of IF and/or HIIT on body composition, systemic glucose homeostasis and plasma lipids in diet induced obese mice. Chapter 5 presents the underpinning molecular adaptation that derived the reduction in adipose tissue mass induced by IF and/or HIIT including alterations in miRNAs. Chapter 6 describes the molecular changes induced by IF and/or HIIT on muscle energy sensors, oxidative metabolism, inflammation, insulin signalling, and muscle atrophy, including changes in muscle specific miRNAs. Chapter 7 presents the study conclusion, limitations and future directions.

# **CHAPTER 2**

## LITERATURE REVIEW

Obesity is a condition in which excess body fat has accumulated to an extent that may adversely affect the health of an individual. According to World Health Organization, individuals with a body mass index (BMI) greater than 25 and 30 are considered overweight and obese, respectively. There are multiple causes of obesity, for example genetic, physiological, alterations in neurotransmitter functions, changes in epigenome, dysbiosis of gut microbiota, effect of certain nutrients on mitochondrial function that promote fat accumulation, environmental, socio-economic, and psychological which either alone or in combination with another leads to obesity. While many factors may influence an individual's weight, overweight and obesity are due mainly to an imbalance of energy intake from the diet and energy expenditure (through physical activities and bodily functions). During onset and development of obesity, various changes occur at morphological, physiological, cellular and molecular levels both systemically and in metabolically active tissues like adipose tissue, muscle, liver, brain and pancreas etc. (Wilson et al., 2017, Rogge and Gautam, 2017).

#### 2.1. CAUSES OF OBESITY

The etiology of obesity is quite complex and involves many factors for example, genetic, physiological, alterations in neurotransmitter functions, changes in epigenome, dysbiosis of gut microbiota, effect of certain nutrients on mitochondrial function that promote fat accumulation, environmental, socio-economic, and psychological which either alone or in combination with another leads to obesity (Wright and Aronne, 2012). Although multifactorial in origin, it is evident from the literature that increased

availability of food supplies (increased energy intake) and a reduction in physical activity (decreased energy expenditure) have been the major contributor to the increased prevalence of obesity (Olivo-Marston et al., 2014).

## 2.1.1 Environmental

As per United States centre for disease control, incidence of obesity in USA increased after 1980's (Figure 2.1). This increase occurred within a decade of huge changes in lifestyle (Swinburn, 2008). Within the last few decades' advancement in technology has reduced the need for



Figure 2.1. Obesity by age, United States, 1971-1974 through 2005-2006 (CDC, 2008).

physical activity such as elevators, escalators and other labour saving devices, in addition to passive entertainment such as video games, television, online socialization and entertainment. It has been noticed that average daily physical activity has gone down in the last few decades around the globe (Moore et al., 2012). In conjunction to this, availability of inexpensive highly processed and sugar laden food has increased which is considered as an important factor responsible for soaring increase in obesity (Meldrum et al., 2017). The average consumption of sugar in 1800s was 4-6 pounds per year which has gone up to 150-170 pounds per year today. Sugar has been reported to be highly addictive with both short term and long term adverse health effects (Ifland et al., 2009, Davy and Jahren, 2016). Environmental and behavioural factors alone and in combination create what is known as "obesogenic environment" of modern times (Meldrum et al., 2017).

#### 2.1.2 Genetic and epigenetic

Despite the huge impact of environment on the recent surge in obesity, huge variation has been found among individual's susceptibility to obesity in the population. Heritability plays an important role in determining the susceptibility to obesity and accounts for approximately 40-70% to variation in susceptibility (Locke et al., 2015). Genome wide association studies (GWAS) have identified a number of genetic loci associated with obesity, however these genetic loci can explain only 20% of variation in susceptibility and large portion of it still remains unexplained (van Dijk et al., 2015, Locke et al., 2015).

Notwithstanding, genetic and environmental factors, emerging research suggests epigenetics may play an important role in development of overweight or obese phenotype (Milagro et al., 2013). Epigenetics is the heritable change in gene expression without changes in the DNA sequence. Epigenetic modifications are affected by various external and internal environmental factors and are now considered as important players in the etiology of many diseases like obesity, diabetes and cancers (Choi and Friso, 2010, Ehlert et al., 2013). Epigenetic changes include DNA methylation, histone modification, chromatin remodelling and most recently known microRNA expression (Choi and Friso, 2010).

#### 2.2. EFFECT OF OBESITY ON ADIPOSE TISSUE

Adipose tissue is found under the skin (subcutaneous), around internal organs (visceral) (figure 2.2), in muscle fibres (intramuscular) and in the breast tissue. Adipose tissue consists of adipocytes and non-adipocytes also called vascular-stromal fraction, which is made up of macrophages, fibroblasts, endothelial cells and pre-adipocytes (Hajer

7

et al., 2008). Classically, adipose tissue is considered important for providing insulation and cushion to the body and as a reservoir for storage of excessive energy after food intake and its release during fasting state (Hajer et al., 2008). However, over the last few decades, experimental evidences have shown that adipose tissue is an important endocrine organ, which secretes various bioactive



Figure 2.2. Mouse epididymal white adipose tissue (10x -H&E stained).

compounds to maintain systemic energy homeostasis (Coelho et al., 2013, Choe et al., 2016). Alterations in energy status induce dynamic changes in adipose tissue like numerical and functional changes in adipocyte and extracellular matrix, collectively called adipose tissue remodelling (Choe et al., 2016), manifested as adipose tissue hypoxia, decline in insulin sensitivity, enhanced autophagy, apoptosis and tissue inflammation. Consequently, adipose tissue releases certain signals in the form of adipokines, cells and metabolites which induce pro-inflammatory, diabetogenic and atherogenic serum profile. These signals in turn further promote adipose tissue inflammation and damage other organs like skeletal muscle, liver, brain, endothelium, vasculature and endocrine organs (Klöting and Blüher, 2014). Some important events that leads to adipose tissue dysfunction have been discussed underneath.

#### 2.2.1 Formation of unilocular lipid droplet and adipose tissue expansion

The excess lipids like free fatty acids and sterols are stored after esterification in lipid droplets as neutral lipids (Thiam et al., 2013). Although adipocytes in adipose tissue are the primary storage organs, lipid droplets are also found in other cells like hepatocytes, enterocytes, macrophages and adrenocortical cells (Krahmer et al., 2013). Structurally,
lipid droplet is composed of a non-polar core which is mainly triglycerides or sterol esters surrounded by a polar phospholipid monolayer (Bartz et al., 2007). Phospholipid monolayer is mainly composed of phosphatidylcholine, phosphatidylinositol, lysophosphatidylcholine and lyso-phosphatidylethanolamine (Krahmer et al., 2011, Guo et al., 2008, Fei et al., 2011, Szymanski et al., 2007). In addition to phospholipids, various proteins are also attached to the lipid droplet surface which helps regulating the lipid droplet size, number and lipid metabolism. These include PAT proteins (Perilipin 1, Perilipin2/adipophilin (ADRP), Perilipin3/TIP47, Perilipin4/S3-12, Perilipin 5 (OXPAT)), Cell Death Inducing DFFA like effector (CIDE) proteins, including CIDEA, CIDEB, CIDEC/FSP27) and several lipases. However, the lipid droplet proteome is large with around 100 proteins identified so far (Thiam et al., 2013, Krahmer et al., 2013). Adipose tissue has one large unilocular lipid droplet with a diameter of about 100µm, which is in contrast to other cell types where numerous small lipid droplets are found with diameter ranging from 100nm to 5µm. CIDE proteins promote the formation of unilocular lipid droplets (Krahmer et al., 2013). CIDEC also FSP27/regulate the triglyceride storage in lipid droplet and influence the development of metabolic syndrome (Krahmer et al., 2013). Perilipin1 also regulate triglyceride storage and is an important regulator of lipolysis. It protects the lipid droplets from basal lipolysis but regulates access of lipases to lipid droplets when the lipolysis is stimulated (Sztalryd et al., 2003, Zhai et al., 2010). In state of excess energy intake, owing to the massive accumulation of lipid droplets, the expression of lipid droplet proteins changes which in turn influence the pathological state. For instance, in human adipose tissue expression, of FSP27/CIDEC inversely correlate with insulin resistance (Puri et al., 2008). Higher expression of perilipin1 and CIDEC have been found in adipose tissue of insulin sensitive subjects in comparison to insulin resistant subject of same weight indicating that elevated expression of these proteins enhance triglyceride storage in adipose tissue and protect from lipotoxicity. Polymorphism in FSP27/CIDEC has also been reported to influence metabolic risk (Zhang et al., 2008, Dahlman et al., 2005). However, these human findings are in contrast to the findings in mice/animals. Mice lacking FSP27 have higher energy expenditure and exhibit protection from diet induced obesity and insulin resistance (Nishino et al., 2008). Likewise, perilipin1 knockout mice also remained lean and protected from high fat diet induced obesity (Tansey et al., 2001, Martinez-Botas et al., 2000). These finding indicate the difficulty in translating the animal findings into human pathologies.

The expansion of adipocytes due to enlargement of lipid droplets results in adipose tissue expansion. Initially the adipose tissue expands by increasing the size of existing adipocytes to accommodate incoming triglycerides, a phenomenon called hypertrophy. This is followed by increase in the number of adipocytes, a phenomenon called hyperplasia (Sun and Scherer, 2010). Although previously hypertrophy was considered as the sole mechanism but later it was established that hyperplasia also contributes to the adipose tissue expansion (de Ferranti and Mozaffarian, 2008). During hyperplasia, initially number of pre-adipocytes increases which is followed by differentiation of pre-adipocytes into mature adipocytes. Through some animal studies, it has been revealed that hypertrophy is followed by hyperplasia and the latter is associated with increased severity and reduced irreversibility of metabolic consequences (Hirsch et al., 1989, Bjorntorp et al., 1982), however this has yet to be demonstrated in humans. Although the factors regulating adipose tissue expansion (hypertrophy and hyperplasia) have still remained elusive but it has been seen that insulin and glucocorticoid stimulate differentiation of pre- adipocytes (Avram et al., 2007). Hypertrophied adipocytes have also been reported to secrete tumour necrosis factor- $\alpha$  (TNF $\alpha$ ) and insulin like growth factor-1 (IGF1), both of which stimulate hyperplasia (Avram et al., 2007). In addition to this, Peroxisome proliferator-activated receptor- $\gamma$  (PPAR $\gamma$ ), an important transcription factor has also been reported to stimulate hyperplasia (Tontonoz et al., 1994).

#### 2.2.2 <u>Hypoxia</u>

Adipocytes can expand beyond 100µm in size whereas, the diffusion capacity of oxygen is only 100µm (Trayhurn, 2013). Lack of oxygen supply causes hypoxia in adipocytes at cellular level which leads to induction of Hypoxia-inducible factor 1a (HIF1 $\alpha$ ) (Ban Jae-Jun et al., 2014). HIF $\alpha$  is a transcription factor induced by hypoxia and is ones of the adipose tissue hypoxia marker genes (He et al., 2011). HIF1a is constantly expressed in the cell and increase in mRNA level is not necessary for elevation of HIF1a protein level (He et al., 2011). HIF1a rapidly degrades under normoxic conditions (Rutkowski et al., 2015) however under hypoxic conditions degradation is inhibited (He et al., 2011) leading to elevated level of HIF1a protein. In adipocytes, elevated HIF1a fails to induce pro-angiogenic response, albeit the underpinning reason is not completely understood. Several studies have demonstrated that even though the HIF1 $\alpha$  level is effectively stabilized under hypoxia, obese adipose tissue lack other co-factors like vascular endothelial growth factor (VEGF) necessary for vascularization (Sun and Scherer, 2010). Selective inhibition of HIF1a improves adipose tissue metabolism indicating that HIF1α responsive genes are more important for adipose tissue dysfunction than vascularization (Sun et al., 2013a). In addition to this, deletion of HIF1 $\alpha$  in adipocytes protect against inflammation, insulin resistance (Lee et al., 2014). These effects have been related to the protective signalling of estrogen receptor  $\alpha$  which limits hypoxia and fibrosis by HIF1a ubiquitination and deactivation (Kim et al., 2014b). It has

been suggested that inhibition of HIF1 $\alpha$  has a therapeutic potential to prevent the detrimental effect of obesity (Ban Jae-Jun et al., 2014). Overexpression of HIF1 $\alpha$  in adipocytes did not exhibit pro-angiogenic response per se rather it induced another transcriptional panoply involved in remodelling of extracellular matrix (Halberg et al., 2009).

## 2.2.3 Extracellular matrix and cytoskeletal remodelling

A unique extracellular matrix (ECM) surrounds adipocytes which provides mechanical support to the cells and also participates in variety of signalling events (Khan et al., 2009). ECM surrounding adipose tissue is an isotropic matrix of collagen and elastic fibres with extracellular fibronectin and laminin forming networks with collagen fibres (Pope et al., 2016). The components of ECM are regulated by matrix metalloproteinases, a family of neutral endopeptidase that cleaves ECM components and facilitates remodelling of ECM (Sun et al., 2011). The health of adipocyte is regulated by interaction with ECM and matrix composition per se (Rutkowski et al., 2015). Flexibility of ECM is important for healthy expansion of adipocytes. However, during the course of development of obesity, ECM loses its flexibility and plasticity due to enhanced interstitial fibrosis in white adipose tissue, resulting in adipose tissue dysfunction. Excessive synthesis of fibrillar components of ECM like collagen I, III and VI and reduced breakdown of these components leads to their accumulation and adipose tissue fibrosis (Sun et al., 2013b). Adipocytes in fibrotic ECM lose proper function and exhibit increased rate of fibrosis. Once the necrotic cell dies, the large lipid droplets remain in the tissue. The surrounding dysfunctional live adipocytes induce infiltration of macrophages, neutrophils, lymphocytes, and mast cells, initiating a local proinflammatory environment (Sun et al., 2013b). Destabilised ECM due to lack of an

12

important component, Collagen VI, does not affect the phenotype of mice in comparison to wild type. On challenging these mice with high fat diet or with *ob/ob* mutation, despite increase in adipocyte size, the tissue metabolic health and systemic metabolic phenotype of mice radically improved. Which indicates that, expansion of adipocytes itself does not leads to adipose tissue dysfunction and ECM plays an important role in adipose tissue dysfunction rather than merely an epiphenomenon associated with obesity (Khan et al., 2009, Sun and Scherer, 2010). HIF1 $\alpha$  has been found to play a critical role in inducing the fibrotic response. Enhanced expression of HIF1 $\alpha$  under expanding hypoxic adipocytes induce increased transcription of ECM components leading to development of fibrosis (Sun et al., 2013b). HIF1 $\alpha$  also alters cellular redox state which affects enzymes associated with collagen crosslinking and stabilization like lysyl oxidase and prolyl-4hydroxylase (Mariman and Wang, 2010). PPAR $\gamma$  has also been found to exhibit antifibrotic activity by supressing the transcription of collagen (Khan et al., 2009).

The cytoskeleton of adipocytes is also remodelled during adipogenesis which alters the mechanical properties of the tissue. During the expansion of lipid droplet, the vimentin (type III intermediate filament protein) expression increases to increase the size of vimentin cage surrounding each lipid droplet, whereas the expression of actin and tubulin decreases (Pope et al., 2016).

# 2.2.4 <u>Metabolic dysregulation (reduced oxidative capacity and insulin</u> <u>sensitivity)</u>

# 2.2.4.1. Lipogenesis

The extra energy stored in adipose tissue as triglycerides are synthesized from fatty acids either by *de novo* lipogenesis from non-lipid precursors or taken up from circulating triglycerides in chylomicron and very low density lipoproteins (VLDL) (Lafontan, 2008, Luo and Liu, 2016). For *de no* lipogenesis, glucose provides acetyl-

13

coenzyme A (acetyl CoA), induces the rate limiting enzyme acetyl-CoA carboxylase and also stimulates the release of insulin from pancreatic  $\beta$  cells which boosts lipogenesis (Luo and Liu, 2016). Insulin stimulated uptake of glucose by adipocyte, activate glycolytic and lipogenic enzymes and induce the expression of transcription factor sterol regulatory element-binding protein 1 (SREBP1) which regulates the expression of fatty acids, triglycerides, phospholipids and cholesterol synthesis genes (Assimacopoulos-Jeannet et al., 1995, Ferre and Foufelle, 2007). In addition to SREBP1, carbohydrate response element-binding protein (ChREBP) also stimulates *de novo* lipogenesis and regulates glucose and lipid metabolism in adipocytes and systemic insulin sensitivity (Eissing et al., 2013, Herman et al., 2012). In adipose tissue, the *de no* lipogenesis is low in comparison to liver under normal conditions in rodents and is even lower in humans (Letexier et al., 2003, Swierczynski et al., 2000). However this does not rule out the possibility of enhanced *de novo* lipogenesis under high carbohydrate intake state (Lafontan, 2008).

The majority of fatty acids incorporated into triglyceride stored in adipose tissue are supplied by circulating lipids (Luo and Liu, 2016). Lipoprotein lipase (LPL), secreted by adipocytes, translocated to the lumen of adipose tissue capillaries and catalyses hydrolysis of one fatty acid from circulating triglycerides (Frayn, 2002, Kersten, 2014). LPL activity is regulated by insulin and Angiopoietin-like 4 (Angptl4) (Luo and Liu, 2016). During adipocyte hypertrophy the LPL activation is impaired (Lafontan, 2008), which can lead to increased availability of fatty acids in non-adipose tissues and ectopic lipid accumulation (Mittendorfer, 2011).

# 2.2.4.2. Lipolysis

During the period of energy demand triglycerides in adipose tissue are hydrolysed by lipolysis, which supplied free fatty acids as fluids for other organs and glycerol for hepatic gluconeogenesis (Kuriyama et al., 2002). Adipose tissue triglyceride lipase (ATGL) and hormone sensitive lipase (HSL) are the major lipases besides 20 other lipases (Saponaro et al., 2015). ATGL catalyses the first step of conversion of triglycerides to diglycerides and a free fatty acid. ATGL belongs to patatin like phospholipase family, also called PNPLA2. Besides adipose tissue, ATGL is also present in other tissue forms like liver, muscle and heart, however its function is slightly different types of tissue. ATGL is activated by fasting, glucocorticoids, PPAR agonists and act in the presence of coactivator protein named comparative gene identification-58 (CGI-58) (Haemmerle et al., 2006, Shaw et al., 2013, Smirnova et al., 2006). HSL although can act on triglycerides, diglycerides, monoglycerides and cholesteryl esters but is mainly involved in hydrolysis of diglycerides to monoglycerides and monoglycerides to free fatty acid and glycerol (Kraemer and Shen, 2002). During fasting, elevated glucagon level activates cAMP dependent protein kinase A (PKA) which phosphorylates perilipin which in turn translocates HSL to the lipid droplet for lipolysis (Marcinkiewicz et al., 2006, Lafontan and Langin, 2009, Brasaemle, 2007). Insulin promotes lipogenesis and inhibit lipolysis (Gastaldelli, 2011). Insulin resistant individuals exhibit higher level of plasma free fatty acids in comparison to insulin sensitive subjects at the same plasma insulin level. Elevated plasma free fatty acids inhibit insulin signalling, reducing basal and plasma stimulated glucose uptake (Boden et al., 2001). Free fatty acids decline muscle ATP synthesis (Brehm et al., 2006), and nitric oxide production (Wang et al., 2006), and impair insulinstimulated activation of phosphoinositol-3 kinase (PI3K), pyruvate dehydrogenase

kinase-isozyme1(PDK1), RAC-alpha serine/threonine-protein kinase (also known as proto-oncogene c-AKT), and endothelial nitric oxide synthase (eNOS) (Wang et al., 2006). Also, elevated level of plasma free fatty acids is related to higher level of cellular diglycerides, the first step of triglyceride synthesis. In addition to this, excess free fatty acids also increases insulin secretion and impedes  $\beta$ - cell function (Gastaldelli, 2011, Kashyap et al., 2003).

# 2.2.4.3. Lipid handling

The increased plasma free fatty acid concentration under obese or insulin resistant state could be due to increased adipose tissue lipolysis and free fatty acid secretion, increased hepatic VLDL triglyceride synthesis and secretion, and reduced peripheral free fatty acid uptake (Gertow et al., 2004). The cellular uptake of free fatty acids is facilitated by membrane associated fatty acid transport proteins (FATPs) and fatty acid translocase (FAT/CD36) (Rutkowski et al., 2015, Gertow et al., 2004). Intracellularly, the free fatty acids transport is facilitated by a family of lipid chaperone protein called fatty acid binding proteins (FABPs) (Furuhashi et al., 2014). FABPs have been suggested to play an important role in lipid trafficking with in the cell facilitating fatty acid transport to mitochondria and peroxisomes for oxidation, to nucleus for fatty acid mediated transcriptional regulation, to endoplasmic reticulum for signalling and membrane synthesis, and to lipid droplets for storage (Zimmer et al., 2004, Spiegelman et al., 1983). The level of FABPs in the cell reflects the rate of fatty acid handling and metabolism (Veerkamp and van Moerkerk, 1993, Furuhashi et al., 2014). FABPs have several isoforms namely FABP1, FABP2, FABP3, FABP4, FABP5, FABP6, FABP7, FABP8 and FABP9. FABP4 also known as aP2, is abundantly expressed in adipose tissue and constitutes 1% of all soluble proteins in adipocytes (Baxa et al., 1989, Spiegelman et al.,

16

1983). FABP4 is transcriptionally regulated by PPARγ, insulin, fatty acids and dexamethasone (Furuhashi et al., 2014). FABP4 deficient mice exhibit enhanced body weight and reduced insulin resistance in both high-fat diet fed and genetically obese mice but its effect on insulin sensitivity was not observed in lean mice (Hotamisligil et al., 1996, Uysal et al., 2000). FABP4 has been reported to physically interact with HSL, stimulates its activity and regulates lipolysis (Hertzel et al., 2002). FABP4 deficient mice exhibit impaired lipolysis in vitro and vivo (Scheja et al., 1999). FABP4 deficient adipocytes (*in vitro*) exhibit reduced lipolysis, indicating its role in FA efflux from adipocytes (Hertzel et al., 2002, Furuhashi et al., 2014). Loss of FABP4 is compensated by FABP5 which is mainly expressed in epidermal cells but also expressed in other tissues including adipose. Although, the amount of FABP5 is 100 times lower than the FABP4, both of them contribute to the development of insulin resistance and atherosclerosis (Furuhashi et al., 2014).

## 2.2.5 Inflammation

Adipose tissue remodelling under positive energy balance is accompanied by infiltration of immune cells into adipose tissue resulting in chronic low grade adipose tissue inflammation (Wang and Ye, 2015). Various immune cells like macrophages, granulocytes, and lymphocytes gets accumulated in the stromal vascular fraction of adipose tissue (Esser et al., 2014). Although enlarged adipocytes secrete proinflammatory cytokines and chemokines, macrophages have been reported to make significant contribution to the adipose tissue derived pro-inflammatory cytokines (Skurk et al., 2007, Esser et al., 2014). Macrophages in adipose tissue are of two types namely M1 and M2 macrophages. M1 macrophages secrete TNF- $\alpha$ , interleukin-6 (IL-6) and thus enhance inflammation whereas M2 macrophages secrete IL-10 which is an antiinflammatory cytokine (Chawla et al., 2011). Macrophage infiltration correlates directly with adiposity, systemic inflammation, insulin resistance (Weisberg et al., 2003) and metabolic syndrome (Esser et al., 2013). Large adipocytes release saturated free fatty acids which binds to toll like receptors 4 (TLR4) on macrophages, activating nuclear factor-kappa B (NF- $\kappa$ B) and enhancing production of TNF- $\alpha$ . TNF- $\alpha$  in turn activate human adipocytes, inducing lipolysis and enhancing expression of genes like IL-6, intracellular adhesion molecule-1 (ICAM1) and macrophage chemo attractant protein-1 (MCP1). ICAM1 and MCP1 facilitates diapedesis of monocytes from blood to adipose tissue and their differentiation into macrophages (Hajer et al., 2008).

These immune cells release pro-inflammatory cytokines including TNF- $\alpha$ , IL-1, IL-6 (Wang and Ye, 2015), which act in autocrine or paracrine manner, activating c-JUN N-terminal kinase (JNK) and nuclear factor-kappa B (NF- $\kappa$ B) pathways in peripheral tissues and promote insulin resistance by interfering in insulin signalling (Shoelson et al., 2006)

## 2.2.6 Endocrine changes

Apart from storing energy, adipose tissue releases various molecules, called adipokines (Table.1), which acts as endocrine, paracrine and autocrine signals, controlling various cellular functions in several tissues like hypothalamus, pancreas, liver, skeletal muscle, kidneys, endothelium, and the immune system (Coelho et al., 2013, Booth et al., 2016, Leal and Mafra, 2013). It has been suggested that most adipokines secreted by adipose tissue are produced by cells other than adipocytes. Leptin, an adipokine released by white adipose tissue is the first one discovered in 1994 (Zhang et al., 1994). Leptin controls adipose tissue growth through its effect on central nervous system. Leptin targets  $\gamma$ -amino butyric acid (GABA)-nergic neurons to reduce appetite

and enhance energy expenditure via its receptors (Vong et al., 2011, Leal and Mafra, 2013). Leptin receptors are located in various hypothalamic sites (Berglund et al., 2012), liver, kidney, lung, pancreas and adipose tissue (Hoggard et al., 1997, Lollmann et al., 1997, Leal and Mafra, 2013). Under obese conditions, hyperphagia and increased adipose tissue mass with hyperleptinemia indicates leptin resistance (Vazquez-Vela et al., 2008, Coppari and Bjorbaek, 2012). Leptin mRNA expression and plasma concentration has been found to be directly proportional to the severity of obesity (Considine et al., 1996, Vidal et al., 1996). Hyperleptinemia is associated with insulin resistance (Howard and Flier, 2006), inflammation (Loffreda et al., 1998), activation of macrophages, TNF $\alpha$ , reactive oxygen species production, MCP-1 expression and endothelial cell proliferation and migration (Loffreda et al., 1998, Konstantinides et al., 2001, Cooke and Oka, 2002).

Adiponectin, another adipokine secreted by adipose tissue is considered as "good adipokine" as it improves insulin sensitivity through activation of AMPK in liver and skeletal muscle (Yamauchi et al., 2002). Adiponectin lowers gluconeogenesis enzyme expression (Kadowaki and Yamauchi, 2005), reduce inflammation by inhibition of TNF $\alpha$ induced activation of Nuclear factor- $\kappa$ B (Ouchi et al., 2000). Adiponectin concentration decreases with increase in adiposity (Yatagai et al., 2003, Turer et al., 2011) and level comes to normal with weight loss.

As mentioned in the previous section, obesity is associated with low grade inflammation along with activation of macrophages and enhanced production of proinflammatory cytokines like TNF $\alpha$  and IL-6 (Lumeng et al., 2007). TNF $\alpha$  is considered as an important player in activation and recruitment of inflammatory cells (Clark, 2007). It has also been associated with insulin resistance in skeletal muscle and adipose tissue by abnormal phosphorylation of IRS1 and declining glucose transporter type 4 (GLUT4) signal transduction and translocation (Lorenzo et al., 2008). TNF $\alpha$  also activates Nuclear factor- $\kappa$ B (NF- $\kappa$ B) (Collins et al., 1995), which in turn is vital transcriptional factor of an array of inflammatory genes (Antunes and Han, 2009). IL-6 is another proinflammatory factor produced by monocytes, fibroblasts and the stromal vascular fraction of visceral white adipose tissue (Fain et al., 2004). Plasma levels of IL-6 increases with adiposity and is stimulated by TNF $\alpha$  and IL-1. Increased circulatory levels are associated with increase of coronary artery disease, atherosclerosis and instable angina (Coelho et al., 2013). Il-6 inhibits insulin signalling by activating Suppressor of cytokine signalling 3 (SOC3) expression, which in turn hampers insulin induced insulin receptor and insulin receptor substrate-1 phosphorylation in adipocytes and hepatocytes (Coelho et al., 2013). Both TNF $\alpha$  and IL-6 are involved in regulation of other adipokines and both reduce the expression of adiponectin (Bruun et al., 2003). However, TNF $\alpha$ stimulates the production of leptin and chemerin (Wang and Trayhurn, 2006, Sell et al., 2009).

Table 2.1. Factors secreted by adipose tissue into the bloodstream and resp	ective
function/effect in their targets (Coelho et al., 2013).	

Molecule	Function/effect
Leptin	Signals to the brain about body fat stores. Regulation of appetite and energy
	expenditure. Wide variety of physiological functions
Adiponectin	Plays a protective role in the pathogenesis of type 2 diabetes and cardiovascular
	disease
Resistin	Hypothetical role in insulin resistance
TNF-α	Affects insulin receptor signalling, possible cause of the development of insulin
	resistance in obesity
IL-6	Pro-inflammatory, lipid and glucose metabolism, regulation of body weight
PAI-1	Inhibitor of the fibrinolytic system by inhibition of activation of plasminogen
Angiotensinogen	Precursor of angiotensin II; regulator of blood pressure and electrolyte homeostasis
FFA	Oxidized in tissues to produce local energy. Serve as a substrate for triglyceride and
	structural molecular synthesis. Involved in the development of insulin resistance
ASP	Influences the rate of triacylglycerol synthesis in adipose tissue
VEGF	Stimulation of angiogenesis
Adipsin	Potential relation between the complement pathway and adipose tissue metabolism
Glycerol	Structural component of the major classes of biological lipids and gluconeogenic
	precursor
IGF-1	Stimulates proliferation of a wide variety of cells and mediates many cells and many
	of the effects of growth hormone

#### 2.3. EFFECT OF OBESITY ON MUSCLE TISSUE

Skeletal Muscle constitutes approximately 40% of body weight of a lean nonobese individual (male or female) (Pedersen and Febbraio, 2012). It is an important metabolic organ, plays an important role in energy expenditure, physical strength, locomotion (Pattanakuhar et al., 2016) and is a highly plastic organ (Loenneke and Loprinzi, 2016). Skeletal muscle is composed of different types of muscle fibres and ratio between these different types of fibres is determined by various genetic and environmental factors (Simoneau and Bouchard, 1995). In resting conditions, lipids contribute 90% of energy requirement of skeletal muscle, indicating that skeletal muscle determines the fate of circulating lipids (Bonen et al., 2006). Under positive energy balance, level of circulatory lipids increases, resulting in accumulation if lipids in adipose tissue, liver and skeletal muscle (Stern et al., 2016). Accumulation of lipids in skeletal muscle is associated with lower lipid oxidation as it was found in obese individuals before and after weight loss (Kelley et al., 1999, Kim et al., 2000). Lower lipid oxidation is attributed to lower activities of enzymes involve in fatty acid oxidation (Kim et al., 2000). Accumulation of lipids and their metabolites perturbs the metabolic functions of muscle which may leads to muscle insulin resistance in obesity and development of metabolic syndrome (Wells et al., 2014). Increasing evidence have indicated that skeletal muscle is an important organ regulating the systemic glucose homeostasis (Baskin et al., 2015). Disturbance in muscle metabolic function due to lipid accumulation in muscle is a mediator or marker of insulin resistance remains questionable (Goodpaster and Kelley, 2002). Nevertheless, it is clear that positive energy balance leads to structural and functional changes in skeletal muscle. Some aspects of obesity induced changes in skeletal muscle relevant to this study are discussed in this section.

# 2.3.1 <u>Fibre type switching</u>

Skeletal muscle is composed of different types of muscle fibres, broadly classified as slow twitch or type I fibre and fast twitch or type II fibre. Based on the expression of myosin heavy chain (MyHC), muscle fibres are further classified as type I, IIa, IIx and IIb (Figure 2.2) expressing MyHC7, MyHC2, MyHC1 and MyHC4



Figure 2.3. Transverse section of soleus muscle showing type 1 (dark) and type 2a (light) fibres.

respectively. Type IIb fibres are only found in rodents and not in humans as humans do not express MyHC4 (Talbot and Maves, 2016). Slow twitch fibres exhibit low fatigability, high oxidative capacity and preference for fatty acids as substrate for ATP production. On the other hand, fast twitch fibres exhibit moderate (IIa) to high (IIb) fatigability, high or medium (IIa) to low (IIb) oxidative capacity and preference for glucose as substrate for ATP production (Bassel-Duby and Olson, 2006, Baskin et al., 2015). Both genetics and environmental factors are responsible for the ratio of oxidative to glycolytic muscle fibre in the body (Simoneau and Bouchard, 1995). In the state of positive energy balance, characteristic of obesity and diabetes, muscle fibres switch from slow twitch to fast twitch which lowers the oxidative capacity of muscle (Mootha et al., 2003).

Two different mechanisms have been hypothesized for obesity induced alterations in muscle fibre type (Pattanakuhar et al., 2016). One of the mechanism, also called metabolic hypothesis proposes that, the obesity induced changes in muscle fibre type are induced by metabolic changes in muscle fibres due to oxidative stress and inflammation. Muscle fibre type phenotype is controlled by several proteins and transcriptional factors involved in various signalling pathways like p38 mitogen-activated protein kinase (MAPK) (Murgia et al., 2000), calcium/calmodulin-dependent protein kinase IV (CaMKIV) (Wu et al., 2002), calcineurin (Naya et al., 2000) and peroxisome proliferatoractivated receptor gamma coactivator 1-alpha (PGC-1a)-peroxisome proliferatoractivated receptor delta (PPAR $\delta$ ) pathway (Wang et al., 2004, Schuler et al., 2006b, Rockl et al., 2007), however, the PGC-1 $\alpha$ -PPAR $\delta$  pathway is the considerably supported by several experimental evidences (Stuart et al., 2013, Rockl et al., 2007, Kaneko et al., 2011, Couturier et al., 2013, Ringseis et al., 2013, Shortreed et al., 2009). PGC-1a is a coactivator interacting with all forms of PPAR isoforms and more details of its role in metabolic regulation will be discussed in the later section 2.2.3. As for muscle phenotype, genetic modification studies in mice have indicated that PGC-1 $\alpha$  is a key regulator of muscle fibre type alterations (Wang et al., 2004, Schuler et al., 2006a). Myocyte enhancing factor 2 (MEF2) is the main downstream transcription factor of PGC-1 $\alpha$ responsible for oxidative type muscle fibre transformation (Rockl et al., 2008). A systemic oxidative stress and inflammation reduces PGC-1a expression in skeletal muscle, which results in relatively reduced oxidative muscle fibres (Liang and Ward, 2006).

Second mechanism, also called mechanical hypothesis, relates muscle fibre type to workload. Weight gain during obesity increases the muscle workload particularly in weight bearing muscles (Kaneko et al., 2011). Increased muscle work load promotes glycolytic muscle fibre phenotype (Scott et al., 2001). Oxidative muscle fibres are associated with walking ability (Zierath and Hawley, 2004) which is reduced in obese individuals owing to reduced oxidative fibres and more glycolytic fibres induced by increased muscle work load (Healy et al., 2011). This hypothesis suggests that fibre type switching is an indirect repercussion of obesity. However long term clinical studies are required to prove this hypothesis.

# 2.3.2 Incomplete Fat oxidation and insulin resistance

Obesity is associated with development of muscle insulin resistance. Although the exact mechanism remains elusive but according to the model hypothesised by Muoio et al. (2008) fatty acid influx and β-oxidation increases in muscle mitochondria owing to both increase in transcriptional regulation and fuel oversupply under obese conditions. However, in state of positive energy (over nutrition and low physical activity), there is no co-ordinated increase in downstream TCA cycle and electron transport chain to fully oxidize acetyl-CoA. Downstream cycles are not only inactivated at transcriptional level but are also hampered due to high energy redox state (increasing NADH/NAD and acetyl-CoA/free CoA ratios). As a result, the by-products of incomplete fatty acid oxidation (acylcarnitines, ketones and reactive oxygen species) accumulate in muscle, which leads to accumulation of long chain-CoA species and subsequent production of other lipid derivatives like diacylglyceroles, ketones, ceramides, intramuscular triacylglycerols and yet some unidentified metabolites. Accumulation of these lipid species affects insulin signal transduction and sensitivity, impeding glucose uptake and metabolism (Muoio et al., 2008).

Besides accumulated lipid derivatives affecting insulin signalling, growing number of evidence suggest that inflammation of skeletal muscle in obesity adversely affect muscle metabolism and contributes to insulin sensitivity (Wu and Ballantyne, 2017). This aspect will be discussed later in inflammation section.

24

# 2.3.3 <u>Insulin signalling</u>

After a meal insulin is released into the blood stream. Insulin binds to insulin receptor, a dimeric protein also called insulin receptor kinase. This bonding leads to closing of the protein dimer and cross-phosphorylation of tyrosine residues of insulin receptors changing its conformation and activates insulin receptor kinase (Di Meo et al., 2017). One of the phosphorylated tyrosine residue attract a protein called insulin receptor substrate (IRS), which has two isoforms (IRS1 and IRS2). Upon binding to insulin receptor, IRS also gets phosphorylated at tyrosine residues. IRS acts as an attachment point for PI3K (Cheatham et al., 1994), which phosphorylates Phosphatidylinositol 4,5-bisphosphate (PIP2) to Phosphatidylinositol (3,4,5)-trisphosphate (PIP3) that travels along membrane to activate phosphoinositide-dependent protein kinase-1 (PDK1). PDK1 is not membrane bound and travels around the cell and stimulate a cascade of signalling events that results in the movement of glucose transporter 4 to cell membrane to facilitate glucose transport (White, 2002) (Figure 2.4).



Figure 2.4. Insulin-signalling pathway for insulin-stimulated GLUT4 translocation and glucose transport in skeletal muscle (Cartee, 2015).

In obese and type 2 diabetes subjects, IRS1 tyrosine phosphorylation and PI3K activity has been found to be reduced (Goodyear et al., 1995, Björnholm et al., 1997). Reduced IRS1 tyrosine phosphorylation has been related to enhanced serine threonine phosphorylation(Paz et al., 1997) by serine/threonine kinases including IkB kinase (IKK), c-JUN N-terminal kinase (JNK) and mammalian target of rapamycin (mTOR) (Gao et al., 2002, Aguirre et al., 2002, Li et al., 1999). Consequently, level of PI3K decreases, affecting the downstream targets leading to lower activity of Protein kinase B (AKT) (Kim et al., 1999), atypical Protein kinase C (PKC) (Kim et al., 2003) and lower glucose uptake probably due to reduced GLUT activity/translocation (Shulman, 2000). It is important to mention that a number of studies have provided evidence to support dysfunctionality of IRS in insulin resistance, however there have been some inconsistencies in the model and it has also been observed that insulin resistance can occur independent of IRS (Hoehn et al., 2008).

## 2.3.4 <u>Master regulators-PGC-1α</u>

PGC-1 $\alpha$  has been extensively described as a master regulator of metabolic programming in skeletal muscle, both in health and disease (Chan and Arany, 2014). In addition to fibre type switching mentioned above, PGC-1 $\alpha$  regulates the expression of various genes related to lipid oxidation and mitochondrial metabolism (Canto and Auwerx, 2009). Besides muscle, PGC-1 $\alpha$  is highly expressed in tissues with high oxidative capacity lie heart, liver, brown adipose tissue and brain (Puigserver et al., 1998, Mootha et al., 2004). PGC-1 $\alpha$  increases energy expenditure by co-ordinately enhancing mitochondrial biogenesis, respiration rates and substrate uptake and utilization for energy production (Chan and Arany, 2014, Wu et al., 1999, St-Pierre et al., 2003, Lehman et al., 2000). These wide range of functions of PGC-1 $\alpha$  are exerted by directly co-activating multiple transcription factors such as PPARs (Vega et al., 2000, Wang et al., 2003), myocyte enhancer factor 2 (MEF2) and the family of fork head o-box (FOXO) and nuclear receptors such as thyroid hormone receptor (Puigserver et al., 1998), glucocorticoid receptors (Vega et al., 2000), estrogen receptor (Vega et al., 2000, Puigserver et al., 1998) and estrogen related receptor (Schreiber et al., 2003, Huss et al., 2002). Several studies have demonstrated using genetically modified organisms to demonstrate the role of PGC-1a at systemic and skeletal muscle level. Muscle specific PGC-1a knock out mice displayed abnormal glucose homeostasis owing to reduced oxidative muscle fibre type I, decreased endurance capacity and mitochondrial gene expression (Handschin et al., 2007). Contrary to this, mice with muscle specific overexpression of PGC-1a exhibit increased insulin resistance in response to high fat diet. However, same PGC-1 $\alpha$  overexpressing mice exhibit improvement in insulin sensitivity in response to exercise in comparison to exercising wild type mice, indicating the insulin sensitizing effect of PGC-1 $\alpha$ , though the mechanism is still unclear (Finley et al., 2012). PGC-1a expression was found to be lower in humans with metabolic syndromes in conjunction with lower oxidative capacities and reduced expression of metabolic genes (Richardson et al., 2005, Mootha et al., 2003, Tanner et al., 2002). However, PGC-1a expression is restored after weight loss or interventions to maintain glucose homeostasis [36]. Expression of PGC-1 $\alpha$  is not only controlled at transcriptional level but its activity is affected by various post-translational modifications like phosphorylation, acetylation and methylation etc. (Canto and Auwerx, 2009, Mensink et al., 2007). However how these various modifications regulate PGC-1 $\alpha$  activity at any time point is still unclear.

Owing to the role of PGC-1 $\alpha$  as master regulator of metabolism, its activity is regulated by cellular mechanisms sensing metabolic perturbations and substrate availability. In response to these metabolic changes, these sensors determine which enzymatic and transcriptional pathways needs to be tweaked to adapt to the changing environment. Two such metabolic sensors, AMP-activated protein kinase (AMPK) and Sirtuin 1, also known as NAD-dependent deacetylase sirtuin-1 (SIRT1), play a major role in metabolic regulation and have also been reported to regulate the activity of PGC-1 $\alpha$ (Canto and Auwerx, 2009).

#### 2.3.5 Metabolic sensors- AMPK and SIRT1

AMPK and SIRT1 are energy-sensing molecules present in all eukaryotes and probably coexisted thorough out evolution. It has recently become apparent that AMPK and SIRT1 have common targets like PGC-1 $\alpha$ , regulate cellular metabolism in similar way and surprisingly regulate each other (Ruderman et al., 2010). The expression and activity of both AMPK and SIRT1 is affected under metabolic syndromes (Pfluger et al., 2008, O'Neill et al., 2013).

AMPK is a serine/threonine kinase that plays a key role in maintaining skeletal muscle and systemic energy homeostasis (Ferraro et al., 2014). AMPK is a heterotrimer, consisting of catalytic  $\alpha$  subunit and regulatory  $\beta$  and  $\gamma$  subunits. The  $\gamma$  subunit has several cystathione  $\beta$ -synthase domains that under normal conditions primarily bind ATP. However, glucose deprivation, fasting, exercise, cellular and oxidative stress depletes energy stores and increases AMP/ATP and creatine/phosphocreatine (Cr/PCr) ratios and AMP is replaced by ATP on two of these domains (Kahn et al., 2005, Hardie, 2011, Winder and Hardie, 1996, Wojtaszewski et al., 2000). It leads to a conformational change in AMPK, which increase AMPK activity by two to ten folds and phosphorylation of  $\alpha$ 

subunit at Threonine<sup>172</sup>, which increases the activity by several folds. In muscle, an upstream kinase, serine threonine liver kinase B1 (LKB1) is required for phosphorylation of AMPK (Carling et al., 2008). However, LKB1 has been reported to be constitutively expressed in skeletal muscle and AMPK activation under energy deficit conditions is proposed to be a result of conformational changes that makes it resistant to protein phosphatases (Sanders et al., 2007, Ruderman et al., 2010). Nevertheless, to restore energy stores, activated AMPK induce catabolic pathways by upregulating the transcription of enzymes involved in glucose, lipid and mitochondrial oxidative metabolism and mitochondrial biogenesis (Jager et al., 2007, Bergeron et al., 2001, Long et al., 2005, Winder and Hardie, 1996). In addition to this, AMPK inhibits anabolic processes like glycogen synthesis and protein synthesis to spare ATP (Carling and Hardie, 1989).

SIRT1 is an enzyme that catalyses deacetylation of wide range of substrate (Lin et al., 2004) and its activity is tightly regulated by NAD<sup>+</sup>. NAD<sup>+</sup> is an essential coenzyme in all cellular organisms. It participates in wide array of redox reactions which involve electron transfer and quickly change from electron accepting form (oxidising) NAD<sup>+</sup> to electron donating form (reducing) NADH and is therefore an important indicator of cellular energy status (Houtkooper et al., 2010, Bordone and Guarente, 2005). Several studies have indicated that like AMPK, SIRT1 senses changes in NAD<sup>+</sup> availability which changes with nutrient availability (Bordone and Guarente, 2005, Cohen et al., 2004, Nemoto et al., 2004) and energy expenditure (Canto et al., 2009, Suwa et al., 2008). SIRT1 activity increases under low energy conditions that increases cellular NAD<sup>+</sup> level like fasting and calorie restriction and exercise (Rodgers et al., 2005, Chen et al., 2008b, Hayashida et al., 2010, Chabi et al., 2009, Canto et al., 2009, Canto et al., 2010). However,

activity decreases with high fat diet feeding and acute inflammation due to decrease in cellular NAD<sup>+</sup> level (Yoshino et al., 2011, Kim et al., 2011, Kendrick et al., 2011, Tao et al., 2011). Besides NAD<sup>+</sup> level, SIRT activity is also regulated by other factors that affect cellular NAD<sup>+</sup> level, for example, concentration of nicotinamide (NAM) (Ruderman et al., 2010) and activity of NAM phosphoribosyltransferase (Nampt), the rate-limiting enzyme in the salvage pathway of NAD<sup>+</sup> biosynthesis (Li, 2013).

AMPK and SIRT1 regulate each other and are proposed to be the components of a regulatory cycle (Ruderman et al., 2010). It has been suggested that activated AMPK phosphorylates PGC-1 $\alpha$  and activates SIRT1, later in turn deacetylate PGC-1 $\alpha$ , an important step for its activation (Figure 2.5). Although several in vitro and in vivo studies have provided the evidence for the link between the AMPK and SIRT 1, existence of a regulatory cycle has yet to be proven and many gaps needs to be filled. One such gap is link between the action of AMPK and SIRT one as AMPK activation is an early event in contracting muscle and SIRT1 activation is a let event. Nevertheless, it has been hypothesised that in muscle, late activation of SIRT1 helps to maintain the activation of AMPK (Ruderman et al., 2010).



Figure 2.5. Proposed mechanism of PGC1 $\alpha$  activation by cooperation between AMPK and SIRT (Ruderman et al., 2010).

Several human and animal studies have reported that AMPK and SIRT 1 activity decreases in obese state and this dysregulation is related to pathological changes like altered lipid metabolism, hyperinsulinemia, insulin resistance, ectopic lipid accumulation, inflammation, oxidative stress and mitochondrial dysfunction. However, dysregulation of AMPK and SIRT1 accompany or precedes these pathological changes remains elusive (Ruderman et al., 2010).

# 2.3.6 Inflammation

Obesity induced inflammation is well documented in adipose tissue, however number of studies have suggested that inflammation also occurs in skeletal muscle (Warfel et al., 2016, Hommelberg et al., 2010, Kewalramani et al., 2010). Increasing evidence suggest that obesity is associated with immune cell infiltration into skeletal muscle that contribute to the predominant inflammatory cells in muscle (Khan et al., 2015, Lackey and Olefsky, 2016, Olefsky and Glass, 2010). Density of macrophages and T cells have been reported to increase after high fat diet feeding in both humans (Boon et al., 2015, Tam et al., 2014) and mice (Khan et al., 2015, Patsouris et al., 2014, Olefsky and Glass, 2010). Some immune cells like mast cells and eosinophils displayed no change in muscle with obesity (Fink et al., 2014, Altintas et al., 2012), however others like neutrophils, B cells, NK cells and invariant NKT cells have not been reported in SM in obesity (Wu and Ballantyne, 2017).

In muscle, adipose tissue is found between the myocytes called inter-myocellular or intermuscular adipose tissue (IMAT) and surrounding muscle called perimuscular adipose tissue (PMAT) (Wu and Ballantyne, 2017). Both IMAT and PMAT are different from subcutaneous adipose tissue and their amount in muscle increases with obesity and decreases with weight loss (Santanasto et al., 2015). Additionally, they are highly correlated to insulin resistance and inflammation (Kelley and Goodpaster, 2001, Albu et al., 2005, Goodpaster et al., 2000, Haam et al., 2016, Scott et al., 2015). Histologically, macrophages and T cells reside both in muscle adipose tissue (IMAT/PMAT) and between myofibres, however, their density is much higher in muscle adipose tissue (Togashi et al., 2000, de Alvaro et al., 2004, Fink et al., 2014) than between myofibres (Fink et al., 2014, Khan et al., 2015, Patsouris et al., 2014). Obesity linked changes in immune cells in skeletal muscle is much higher in adipose tissue depots than in muscle, which helps to explain no alteration in immune cell number in muscle after weight loss in some studies (Tam et al., 2012, Bruun et al., 2006).

Although the exact trigger for skeletal muscle inflammation remains elusive, fatty acids, particularly dietary saturated fatty acids have been proposed to induce expression of inflammatory molecules like chemokines in both skeletal muscle and adipose tissue (Ciaraldi et al., 2016, Togashi et al., 2000). Myocytes secrete number of cytokines like IL-6, IL-8, IL-15 and other molecules like FGF21, irisin, myonectin, myostatin, also known as myokines (Pedersen and Febbraio, 2012, Eckardt et al., 2014). In comparison to adipokines which are considered as proinfammatory and are involved in development of metabolic syndromes, myokines have beneficial effects on glucose and lipid metabolism and counteract the detrimental effects of adipokines (Lumeng and Saltiel, 2011, Osborn and Olefsky, 2012, Olefsky and Glass, 2010, Eckardt et al., 2014). Myokines can act in autocrine and paracrine manner affecting myocytes and immune cells and can also act in endocrine manner affecting other cells like adipocytes and hepatocytes (Wu and Ballantyne, 2017). IL-6 is most extensively studied myokine. Its secretion and plasma concentration increases with exercise and muscle contraction (Pedersen and Febbraio, 2012, Eckardt et al., 2014, Fischer, 2006). Although considered important but its function is highly arguable in muscle inflammation owing to the inconsistency in results reported by different studies. After acute exercise or acute treatment of myocytes or acute infusion of IL-6 in healthy humans enhance basal and insulin stimulated glucose uptake and systemic insulin sensitivity (Carey et al., 2006, Pedersen and Febbraio, 2012). Positive effect of IL-6 is attributed to phosphorylation of AS160 through AMPK (Kewalramani et al., 2010). IL-6 also enhances lipolysis, fatty acid oxidation, induce expression of anti-inflammatory cytokines such as IL-10 and inhibit proinflammatory cytokines TNF1 $\alpha$  (19, 26). However under chronic conditions like obesity, it acts as proinflammatory and promotes insulin resistance (Wu and Ballantyne, 2017). Negative

effects of IL-6 are attributed to activation of JNK1/2, suppression of cytokine signalling 3 expression and activation of IRS1 tyrosine phosphatase and protein tyrosine phosphatase-1B (Kewalramani et al., 2010). Inconsistencies does not only lie among the acute and chronic studies, even among chronic studies contradictory results have been reported. For instance, IL-6 expression decreased in skeletal muscle of obese rats (Shin et al., 2015), however, its expression level increased in cultured skeletal muscle myocytes of obese type 2 diabetic humans in comparison to healthy cohort (Green et al., 2011). Also, increase in IL-6 expression have been reported in obese type 2 diabetes subjects in comparison to healthy cohort (Reyna et al., 2008, Corpeleijn et al., 2005, Ciaraldi et al., 2016). Taken together, the exact role of IL-6 in muscle insulin sensitivity needs further investigation.

TNF $\alpha$  is a proinflammatory cytokine, its expression increases in obesity with insulin resistance or with type 2 diabetes, inhibits insulin sensitivity and insulin stimulated glucose uptake and impairs mitochondrial structure and function (Wu and Ballantyne, 2017). TNF $\alpha$  manifests these effects through number of mechanisms including activation of extracellular signal regulated kinase 1/2 (ERK1/2), JNK or the NF $\kappa\beta$  pathway (Kewalramani et al., 2010). TNF $\alpha$  also activates sphingomyelinase which produces ceramides (Haus et al., 2009) and reduces expression of sortilin (Kaddai et al., 2009) that apparently controls GLUT4 sorting (Shi and Kandror, 2005).

#### 2.4. MICRORNA

MicroRNAs (miRNA) are noncoding, endogenous, highly conserved (in most instances), 18-25 nucleotide RNAs (Hilton et al., 2013, Flowers et al., 2013, Palmer et al., 2014). MiRNA binds with specific mRNA guided by Watson-Crick complementary base pairing rules and regulate gene expression at post transcriptional level either by

cleavage of mRNA or blocking the translation process. MiRNAs control various cellular processes like proliferation, differentiation and apoptosis (McGregor et al., 2014). In humans, more than 1000 miRNAs have been identified which regulate 60% of human genome (Pasiakos and McClung, 2013). Expression of miRNA is developmental stage, cell and tissue specific (Sood et al., 2006).

## 2.4.1 Biogenesis of miRNA

The miRNAs are transcribed from miRNA genes, introns of protein coding genes or from polycistronic transcripts that encode multiple, closely related miRNAs (Peng et al., 2014). Biogenesis begins with synthesis of long RNA transcript which is generally several thousand nucleotide long stem loop and is called primary miRNA transcript (pri miRNA) (Figure 2.6) (Hilton et al., 2013). Primary miRNA is cleaved near the base of pri miRNA stem loop into 60-70 nucleotide intermediates known as miRNA precursor or pre miRNA, with the help of enzyme Drosha and RNase III endonuclease. Cleavage of primary miRNA by Drosha is staggered, typical of RNase III endonuclease, leaving a 5' end and approximately 2 nucleotide overhanging at the 3' end, defining one end of the miRNA (Basyuk et al., 2003, Lee et al., 2003). These pre miRNAs are actively transported by RAN-GTP and export receptor, Exportin-5 from nucleus to cytoplasm (Bartel, 2004).

In the cytoplasm, pre miRNA is cleaved, two helical turns away from the loop by Dicer (an RNase III endonuclease) with a staggered cut, leaving a 5' end and approximately 2 nucleotide overhanging at the 3' end, defining other end of the miRNA (Bartel, 2004). This results in a double stranded RNA, with about 2 nucleotide overhang at each 3' end. One of the strand from the duplex is called "guide strand (miRNA)" will

35



Figure 2.6. MiRNA biogenesis pathway. MiRNAs are transcribed by RNA polymerase II resulting in pri-miRNA transcripts. Pri-miRNAs are processed by Drosha leaving Pre-miRNAs which are exported from the nucleus via Exportin5. Pre-miRNAs are processed by Dicer, the mature miRNA is then permanently incorporated into the RNA-induced silencing complex and will bind to the 3'-UTR of a target mRNA (Peng et al., 2014).

act as a mature miRNA and the complementary strand is called "passenger strand (miRNA<sup>\*</sup>)". Out of the two complementary strands of miRNA, the strand which acts as guide strand is the one whose 5′ end is less tightly paired (Khvorova et al., 2003, Schwarz et al., 2003). Helicase like enzyme (yet to be identified) separates the guide strand from the passenger strand and the latter is degraded by DiGeorge syndrome critical region gene 8 (DGCR8) protein, commonly called Pasha (Yeom et al., 2006). Now the single stranded mature miRNA is incorporated into ribonucleoprotein complex known as ribonucleoprotein miRNA induced silencing complex (RISC).

RISC is a complex ribonucleoprotein consisting of miRNA, Argonaute protein, DICER and human immune deficiency virus-1 transactivating response element RNA binding protein (McDaneld, 2009). The Argonaute protein plays a major function in RISC by acting as catalytic component of endonucleolytic cleavage (Liu et al., 2004).

## 2.4.2 <u>Regulation of gene expression by miRNA</u>

There are three mechanisms by which the gene expression is regulated by miRNA namely mRNA degradation, blocking translation initiation and translocation to processing bodies.

#### 2.4.2.1. mRNA Degradation

RISC is directed by miRNA to target mRNA by base pairing interactions. The miRNA has 7-8 nucleotides sequence called seed sequence, at the 5' end which binds with mRNA (Heneghan et al., 2010, McDaneld, 2009). The binding of miRNA to mRNA, induce endonucleolytic cleavage between the nucleotide pairing to residue 10 and 11 of miRNA (Elbashir et al., 2001). The cleaved mRNA is then recognized and degraded by general cellular degradation machinery (Valencia-Sanchez et al., 2006). Additionally, miRNA deadenylates poly A tail followed by decapping of mRNA, resulting in unstable mRNA sequence which is susceptible to degradation by cellular mRNA degradation machinery (Giraldez et al., 2006, Wu et al., 2006).

# 2.4.2.2. Blocking translation initiation

Gene regulation by miRNA does not always results in decrease in the level of mRNA indicating that besides degradation/cleavage or deadenylation, miRNAs regulate gene expression by other alternative mechanisms. It has been reported that in certain cases, interaction of miRNA with target mRNA blocks translation initiation by binding to the 5' cap of mRNA instead of inducing cleavage (Chendrimada et al., 2007, Kiriakidou

et al., 2007). Some Argonaute proteins in RISC have similar binding domains as translation initiation factors, competing with the latter for binding with 5' cap of mRNA, blocking translation initiation (McDaneld, 2009). Thus translation initiation is halted, lowering the level of protein without altering the level of mRNA.

## 2.4.2.3. Translocation to processing bodies

Processing bodies (P bodies) are aggregation of translationally repressed mRNA and mRNA decay machinery in cytoplasmic foci (Parker and Sheth, 2007). It has been hypothesized that miRNA-mRNA complex is translocated to P bodies. There are some evidences that support this hypothesis like the presence of enzyme and factors for translational repression. However, certain evidences also suggest that p bodies may only serve as temporary storage units for mRNA because disrupting P bodies does not disrupt the miRNA pathway or degree of translational repression (McDaneld, 2009).

# 2.4.3 MiRNA, energy homeostasis and obesity

MiRNA play important role in energy homeostasis by regulating the expression of various key players of regulatory cascade involved in glucose, lipid and protein metabolism. They are expressed in various tissues involved in energy homeostasis for example adipose tissue, liver, muscle and pancreas and have been implicated to regulate pancreatic islet development,  $\beta$ -cell differentiation, insulin secretion, insulin signalling, insulin sensitivity in muscle, adipose tissue and liver (Dumortier et al., 2013).

Several miRNAs have been reported to play regulatory role in lipid metabolism like miR-14, miR-33, miR-122, miR-370, miR-378/378<sup>\*</sup>, miR-302a, miR-613, miR-168a, miR-758, miR-10b, miR-224 (Peng et al., 2014). MiR-122 is besides being liver specific, is the most abundant miRNA in liver and is the first miRNA identified, which directly regulates lipid metabolism (Rottiers and Näär, 2012, Esau et al., 2006).

Downregulation of miR-122 leads to decreased plasma cholesterol and triglyceride levels associated with altered cholesterol biosynthesis and increased fatty acid oxidation (Dumortier et al., 2013). Although, several genes have been reported to be regulated by miR-122, target validation needs further investigation. Number of studies have shown that bifunctional SREBP2-miR-33a locus regulates lipid metabolism (Rottiers and Näär, 2012). The *Srebf2* gene encodes both SREBP2 which regulated cholesterogenic gene expression and miR-33a, located in the intronic region of the gene. Likewise, *Srebf1* express SREBP1a/c and miR-33b, found in the intronic region of the gene. SPREBP1a/c regulates lipogenic gene expression. Interestingly, Mir-33a and b regulate cholesterol efflux and fatty acid degradation. Taken together, SREBP proteins and miR-33 together regulate the cholesterol and lipid homeostasis (Dumortier et al., 2013).

Glucose homeostasis is maintained by the action of two antagonistic hormones, glucagon and insulin, secreted by  $\alpha$  and  $\beta$  cells of pancreas respectively. Although there haven't been much reported about the miRNA regulation of glucagon secretion, number of miRNAs have been implicated in insulin secretion (Özcan, 2014). For instance, miR-375 have been implicated in regulating  $\alpha$  and  $\beta$  cell growth and proliferation and also influences the physiology of  $\beta$  cells (Poy et al., 2009). In addition to pancreas, miRNAs regulate glucose homeostasis at systemic level. The Lin28/let-7 axis regulates glucose metabolism by targeting various components of insulin signalling pathway (Zhu et al., 2011). Let-7 over expression induce insulin resistance and impaired glucose tolerance (Frost and Olson, 2011). Similarly, downregulation of miR-103/107, improves insulin action and glucose homeostasis (Trajkovski et al., 2011). Improvement in glucose homeostasis by these miRNAs has been attributed to the inhibition of caveolin-1 by these

miRNAs. Likewise, miR-143/145 cluster has been implicated to regulate glucose homeostasis through activation of insulin stimulated Akt (Jordan et al., 2011). Recently, miR-802 has been found to decrease insulin action and hampers glucose tolerance by inhibiting Hnf1b in liver (Kornfeld et al., 2013).

MiRNAs also regulate amino acid metabolism (Dumortier et al., 2013). MiR-29b control amino acid catabolism by targeting mRNA of dihydrolipoamide branched chain acyltransferase component of branched-chain keto acid dehydrogenase, which catalyses first step in branched-chain amino acid breakdown. Branched-chain amino acid homeostasis is important since they contribute 20% of the amino acid found in proteins, regulate hormone secretion (e.g. insulin), protein synthesis, and autophagy, through modulation of the mTOR pathway (Mersey et al., 2005).

# 2.4.4 miRNAs: the effect of diet and exercise

Recent studies have suggested that dietary intake can alter the expression of miRNAs (McGregor et al., 2014). Our recent review has reported that diet rich in fats alters the expression of miRNAs in various tissues like adipose, liver, skeletal muscle, intestine, nervous tissue, heart and corporal tissue (Wilson et al., 2017). For instance, expression of miR-143 has been reported to be elevated in adipose tissue of mice fed high fat diet (45% of total calories from fat) in comparison to the group on standard chow for 8 weeks (Takanabe et al., 2008). The expression of miR-143 was found to be positively correlated with body weight, mesenteric fat weight, plasma leptin concentration levels, peroxisome proliferator-activated receptor- $\gamma$  (PPAR- $\gamma$ ) expression and adipocyte fatty acid-binding protein (aP2). PPAR- $\gamma$  is a nuclear receptor which controls adipocyte differentiation and hypertrophy, while aP2 regulates systemic glucose and lipid metabolism and is highly expressed in obesity (Maeda et al., 2005). High dietary fat intake

40

also upregulates, miR-221 and 222 in adipose tissue (Chartoumpekis et al., 2012). miR-222 targets hypoxia-inducible factor 1-alpha (Hif- $1\alpha$ ) and tumour necrosis factor alpha (TNF- $\alpha$ ), which have been reported to regulate cell differentiation and may also be linked to inflammatory processes (Ortega et al., 2010, Xie et al., 2009b, Xu et al., 1999, Yun et al., 2002) and also associated with arterial smooth muscle angiogenesis (Felli et al., 2005). In skeletal muscle, high fat diet feeding down-regulated expression of miR-1, miR-133a, miR-133b and miR-206, known as the myomiRs and upregulated expression of whereas miR-144 and miR-106b (Chen et al., 2012). Subsequent signalling mapping revealed 19 significant signalling pathways related to insulin resistance. These included signal transduction pathways related to cell proliferation [(Mitogen-activated protein kinases (MAPK), Wnt, neurotrophic, hedgehog, vascular endothelial growth factor (VEGF), and transforming growth factor beta (TGF- $\beta$ )], metabolism (lysine degradation, heparin sulphate biosynthesis and glycan degradation), protein degradation (ubiquitin mediated proteolysis) and oncogenic (colorectal) pathways. Few of these pathways, for instance MAPK have already been implicated in skeletal muscle insulin resistance (Gehart et al., 2010) and key proteins within the MAPK pathway such as p38 and pJUK have been found to be up-regulated in high fat diet-induced diabetic mice (Chen et al., 2012). Downregulation of miR-1, miR-206, and miR-133 levels has also been reported in adipose tissue of high fat diet-induced obese model (Chartoumpekis et al., 2012) and in the vastus lateralis muscle of type 2 diabetic patients (Gallagher et al., 2010). Also, expression of miR-1 was negatively correlated with glycemia in high fat diet-induced diabetic mice (Frias Fde et al., 2016) and thus could be a prospective target for management of skeletal muscle insulin resistance.

In terms of exercise, several studies have reported alteration in miRNA expression in response to resistance versus cardiorespiratory training, acute versus prolonged exercise and acute changes versus chronic changes in miRNA expression in response to long term training in both animal and human studies (Flowers et al., 2015). Resistance training has been reported to result in differential expression of miR-378, miR-299, miR-269 and miR-451 in skeletal muscles of low and high responding human males. Downregulation of these miRNAs in low responders correlated with failure to activate muscle growth and remodelling genes (Davidsen et al., 2011). Acute resistance exercise has also been reported to alter muscle specific primary miRNA (miR-1, miR-133a and miR-206), implicated to regulate muscle cell differentiation and development (Koutsoulidou et al., 2011)), pre-miRNA expression and miRNA biogenesis enzyme, mRNA expression in skeletal muscles of young and old adults (Drummond et al., 2008). Same study also reported that miR-1 expression reduced in young adults but remained unaltered in older adults. Similar increase in expression of miR-1, miR-133a, miR-133b and miR-181a following acute endurance exercise bout have been reported in human skeletal muscle. Increased expression of miR-1 and miR-29b following short term endurance training has also been reported by the same study (Russell et al., 2013). Likewise, another study examining the effect of endurance exercise on miRNA profile of human skeletal muscles revealed alterations in profile of miR-1 and miR-133 with endurance exercise and its reversal to the basal level 14 days after training period (Nielsen et al., 2010). However, no negative correlation between miRNAs and protein involved in endurance adaptation (Transforming growth factor beta (TGFB) and mitogen activated protein kinase (MAPK)) was observed in this study. In C57BL/6J mice, endurance exercise increased the expression miR-1, miR107and mi-181 and decreased expression of miR-23. Decreased expression of miR-23 has been concluded to negatively regulate PGC-1a (Peroxisome proliferator-activated receptor gamma coactivator 1-alpha) (Safdar et al., 2009) which is a transcriptional activator of energy metabolism genes. Similar study on C57BL/6 mice demonstrated that expression of miR-696 is dependent on physical activity and is involved in regulation of PGC-1 alpha and skeletal muscle metabolism (Aoi et al., 2010). In addition to skeletal muscle, plasma miR-486 associated with energy metabolism have been reported to decline by both acute and chronic exercise in young human males (Aoi et al., 2013). In a human study, different circulatory/plasma miRNA were reported to alter their expression profiles in response to different exercise regimes. Acute exhaustive exercise before and after sustained aerobic training upregulate miR-146a and miR-222 associated with anti-inflammation and anti-angiogenesis respectively. The expression of miR-21 (associated with anti-apoptosis, anti-proliferation and antiinflammation) and miR221 (associated with anti-angiogenesis) was increased by acute exhaustive exercise before but not after sustained aerobic training. Plasma level of miR-20a (associated with anti-angiogenesis) was not affected by Acute exhaustive exercise before but increased after sustained aerobic training only (Baggish et al., 2011). Besides skeletal muscle and plasma, impact of exercise on miRNA expression in nervous and cardiac tissue have also been addressed in few animal studies as these kind of experiments are highly invasive and therefore only animal models have been used (Denham et al., 2014).

## **2.2 LIFESTYLE INTERVENTIONS**

## 2.4.5 Intermittent Fasting

The concept of intermittent fasting (IF) or more commonly known as 5:2 diet was popularized by two British authors, Michael Mosley and Mimi Spencer. Broadly, IF refers to calorie restriction by 75%-100% on 1-3 consecutive or alternate days, called fast days, per week and ad libitum food consumption on other days (Patterson et al., 2015, Tinsley and La Bounty, 2015). More precisely IF can be categorized into 3 groups: alternate day fasting, whole day fasting and time restricted eating. Alternate day fasting (ADF) refers to alternating fasting days with only 25% calorie intake and ad libitum feeding days. Whole day fasting refers to 1-2 complete fasting days and ad libitum feeding on other days of the week. However, some protocols allow 25% calorie intake and are therefore referred to as modified fasting (Brown et al., 2013). Time restricted eating included a daily routine with a restricted period called feeding window of 4-8 hours devoted to eating with remaining hours of no calorie intake called fasting window (Rothschild et al., 2014). Although time restricted feeding comes under the umbrella of IF, this section of the review mainly focuses on whole day modified fasting and alternate day fasting, as this study is based on whole day fasting.

Although only few scientific (preclinical and clinical) studies have been reported on IF, still it has gained a lot of attention with in the last decade owing to its health benefits like improvement in body composition, glucose homeostasis and cardiovascular health and delayed aging. Besides that, IF has also been reported to reduce inflammation, delays onset of neuronal dysfunction and degeneration (Mattson et al., 2016). Several animal and human studied have revealed that fasting is effective to reduce weight loss, delay aging and optimize metabolism. In humans, fasting (12-24 hours) decline the serum glucose level by 20% or more and depletes hepatic glycogen. This is accompanied by "metabolic switch" to a mode where non hepatic glucose, ketone bodies and free fatty acids are used as energy source. Ketone bodies are derived from acetyl CoA generated from  $\beta$ -oxidation of fatty acids released into the blood stream from adipocyte and also by
the conversion of ketogenic amino acids (Longo and Mattson, 2014). Impairment of this metabolic switch is a key dysfunction of metabolic syndromes like obesity and diabetes (Storlien et al., 2004).

## 2.4.5.1. Effect of IF on Body composition

Several animal and human studied have suggested that IF is effective for weight loss and beneficial changes in body composition. Weight loss with Intermittent fasting and alternate day fasting have been reported in several human studies. (Klempel et al., 2012, Harvie et al., 2011) (Heilbronn et al., 2005b, Eshghinia and Mohammadzadeh, 2013, Johnson et al., 2007, Varady et al., 2009, Klempel et al., 2013, Bhutani et al., 2013, Varady et al., 2013). These studies suggest that, IF resulted in 3-7% of body weight reduction where as ADF resulted in 4-8% body weight reduction. Slightly higher decrease in ADF than IF could be attributed to more fasting days in ADF as compared to IF. On an average, IF and ADF resulted in 0.25 and 0.75 kg weekly reduction in body weight. Among animal studies, only one study has reported 3% weight loss (Varady et al., 2007) in mice after four weeks of ADF. However, most studies have reported reduced weight gain by 0-40% in comparison to ad libitum control group (Higashida et al., 2013, Karbowska and Kochan, 2012, Li et al., 2013, Beigy et al., 2013, Varady et al., 2010).

Weight loss due to IF or ADF is primarily attributed to decline in fat mass. In humans, IF for 8-24 weeks has been reported to decrease body fat, measured as change in waist circumference, by 1.4-2.3% (Klempel et al., 2012, Harvie et al., 2011). Alternate day fasting also showed similar decline in fat mass, with 3-12 weeks of ADF resulting in 1.3 to 2.4% decline in fat mass (Eshghinia and Mohammadzadeh, 2013, Varady et al., 2009, Bhutani et al., 2013, Klempel et al., 2013, Varady et al., 2013). Among animal studies, change in total fat mass following intermittent or alternate day fasting was not reported by any study, however, decline in intra-abdominal fat mass by 40% (Higashida et al., 2013), white adipose tissue by 46% (Karbowska and Kochan, 2012), visceral fat by 18% (Varady et al., 2010) and no change in inguinal and epididymal fat mass (Varady et al., 2007).

Weight loss interventions are aimed to reduce body weight by loss in fat mass and conserving lean muscle mass. In addition to changes in fat mass, lean muscle mass change has also been reported by various studies on IF. Intermittent fasting in overweight humans for 24 weeks resulted in 1.3% decline in muscle mass (Harvie et al., 2011). Alternate day fasting for 3-12 weeks has also been reported to decrease muscle mass by 1-2.1% (Heilbronn et al., 2005b, Eshghinia and Mohammadzadeh, 2013, Bhutani et al., 2013, Klempel et al., 2013, Varady et al., 2013, Varady et al., 2009). In comparison to calorie restriction, IF has been reported to cause less muscle mass loss (Harvie and Howell, 2017). Weight loss in form of muscle mass has been reported to be higher in lean non-obese individuals in comparison to obese individuals (Varady et al., 2009, Varady et al., 2013). Contrary to this, some studies comparing IF with calorie restriction, also reported equivalent loss of lean muscle mass (Harvie et al., 2011, Hill et al., 1989).

## 2.4.5.2. Effect of IF on glucose homeostasis

In humans, intermittent fasting has been consistently reported to improve glucose homeostasis (Harvie and Howell, 2017, Barnosky et al., 2014). Intermittent fasting for 8-24 weeks in overweight and obese individuals have been reported to cause 2-3% decline in fasting blood glucose (Klempel et al., 2012, Harvie et al., 2011). Alternate day fasting for 3-12 weeks resulted in 1-6% decline in fasting blood glucose (Heilbronn et al., 2005a, Eshghinia and Mohammadzadeh, 2013, Johnson et al., 2007, Varady et al., 2009, Varady et al., 2013, Klempel et al., 2013, Bhutani et al., 2013). In particular, three studies have reported the effect of these fasting interventions in prediabetic individuals (Klempel et al., 2012, Varady et al., 2009, Varady et al., 2013) with a decline of 3-6% in fasting glucose. The extent of decline was found to be directly proportional to the duration of the study with highest decline of 6% reported after 12 weeks of ADF (Varady et al., 2013).

In animal models, decline in fasting blood glucose by 17% has been reported in NMRI mice after six weeks of intermittent fasting (Beigy et al., 2013). However, few studies have reported no significant change in fasting blood glucose after 7-10 weeks of IF in CD-1 wild type mice (Li et al., 2013) and male Wistar rats (Higashida et al., 2013).

In humans, although several studies have reported the effect of IF and ADF on plasma insulin levels but only three studies have reported the effect on prediabetic individuals (Klempel et al., 2012, Varady et al., 2009, Varady et al., 2013). These studies reported 20 – 31% decline in plasma insulin after 8-12 weeks of IF/ADF. Remarkably, two separate studies reported similar decline in insulin level by 20 and 21% after 8 weeks of IF (Klempel et al., 2012) and ADF (Varady et al., 2009) respectively. The decline was comparable despite more number of fasting days implemented in ADF study (3-4 fast days) in comparison to IF (1 fast day). However, interestingly, both interventions prescribed the same level of energy restriction (35-40%), indicating that degree of restriction is a stronger predictor of insulin lowering in comparison to the number of fasting days. In addition to this, duration of the intervention also determines the extent of decrease, for instance, 31% decline in plasma insulin was reported after 12 weeks of (Varady et al., 2013), indicating that longer duration of fasting intervention may be more beneficial (Barnosky et al., 2014).

In male Wistar rats, ADF for 6 weeks resulted in 46% decline in plasma insulin (Higashida et al., 2013). However, contradictory to this and most human studies, 48 days

of ADF in male Wistar rats have been reported to elevate plasma insulin levels by 712% (Karbowska and Kochan, 2012). This indicates that long term effects of IF/ADF on insulin regulation needs further investigation.

Intermittent fasting and ADF studies in humans have constantly reported improvements in insulin sensitivity with these interventions (Barnosky et al., 2014). Consistent improvements in insulin sensitivity (measured in terms of homeostatic model assessment (HOMA-IR)) after 3-4 weeks of IF/ADF have been reported in normo-glycemic and prediabetic individuals. The improvement in insulin sensitivity has been found to be directly proportional to the amount of weight loss. For example, 8% weight loss after 8 weeks of ADF showed 33% improvement in insulin sensitivity (Johnson et al., 2007) however, 4% weight loss after 12 weeks of ADF resulted in 9% improvement in insulin sensitivity (Bhutani et al., 2013). Intermittent fasting studies also reported in improvement in insulin sensitivity by 23% and 27% with weight loss of 4% and 7% after 8 weeks (Klempel et al., 2012) and 24 weeks (Harvie et al., 2011) respectively.

## 2.4.5.3. Effect of IF on lipid profiles

Few studies have reported the impact IF/ADF on plasma lipids. Weight loss have been found to be not always accompanied by beneficial changes in plasma lipids. A recent 8 weeks ADF study reported 9% weight loss accompanied by significant reduction in LDL cholesterol and fasting TAG (Catenacci et al., 2016). Likewise, significant reduction in plasma LDL and TAG after 1.6% reduction in body weight have been reported after 8 weeks of ADF (Varady et al., 2009). However, another 8 weeks ADF study reported significant decline in TAG but no significant change in LDL despite 1.8% decline in body weight (Johnson et al., 2007). Few studies also reported no change in lipid profile despite weight loss after ADF in humans (Bhutani et al., 2013, Eshghinia and Mohammadzadeh, 2013). Among animal studies, two separate studies reported decline in blood lipid levels following 6 weeks (Higashida et al., 2013) and 11 months (Li et al., 2013) of ADF in CD-1 wild type mice and male Wistar rats respectively.

## 2.4.5.4. Molecular changes induced by IF

Several studies have investigated the impact of IF on body composition and physiological parameters but very few studies have investigated the molecular adaptations in metabolically important tissues following IF regime. In a study on male Wistar rats, IF 48 days (3 days fasting, 3 days feeding) resulted in increased mRNA expression of FSP27 in white adipose tissue, indicating that despite low calorie intake under IF, fat deposition in adipose tissue is promoted by increased expression of genes involved on lipid deposition (Karbowska and Kochan, 2012). Even though the expression of these genes is higher in IF group, animals in this group still weighed less and had less adipose tissue mass in comparison to ad libitum control group as well as the calorie restricted animals (food intake reduced to 80% of ad libitum control), indicating that molecular adaptation to IF is different from calorie restriction. Increase in FSP27 was also accompanied by increased mRNA expression of FSP27 in white adipose tissue.

Another recent study in mice reported that IF is able to induce favourable health adaptations despite consuming equivalent amount of calories as the ad libitum group indicating that changing feeding pattern without decreasing the calorie intake can induce metabolic adaptations and prevent metabolic syndromes (Kim et al., 2017). According to this study, IF leads to increased expression of beige/brown adipose markers (i.e., *Ppargc1a, Cidea and Ucp1*) and browning of adipose tissue. In addition to this, IF induced elevated expression of M2 macrophage marker genes (e.g., *CLEC10A and IL10*), indicating the activation of adipose tissue macrophages. Most importantly, IF induced vascular endothelial growth factor (VEGF) specifically in adipose tissue and is responsible for metabolic benefits induced by IF.

## 2.4.6 <u>High Intensity Interval Training</u>

World health organisation and American college of sports medicine recommend at least 150-250 minutes per week of moderate intensity physical activity for moderate weight loss. For clinically significant weight loss, more than 250 minutes per week of moderate intensity exercise is recommended (World Health Organization, 2010) (American College of Sports Medicine, 2013). Despite the established therapeutic potential, this type of exercise regimes is less favoured by healthy and clinical population due to lengthier time commitments (Batacan et al., 2017). High intensity

exercise/training (HIE/HIIT) has been concluded as time efficient strategy for inducing adaptations generally associated with long term low or moderate intensity exercise. HIIT refers to brief bouts of

intermittent/interval



Figure 2.7. An example of a HIIT protocol (Cassidy et al., 2017)

vigorous activity (80-90% VO<sub>2 max</sub>) interspersed by relatively longer bouts of low intensity exercise (Gibala et al., 2012, Trapp et al., 2008). There is also a subcategory of HIIT know as sprint interval training which involves brief bouts of all out sprint (more than 100% VO<sub>2 max</sub>) interspersed by longer bouts of low intensity exercise (Figure 2.7) (Cassidy et al., 2017). In comparison to moderate intensity exercise, HIIT has been

reported to significantly improve aerobic capacity (Karstoft et al., 2014, Schrauwen-Hinderling et al., 2007, Ritov et al., 2010, Mootha et al., 2003), reduce metabolic syndrome associated risks including blood pressure (Mootha et al., 2003), insulin action and lipogenesis (Schrauwen-Hinderling et al., 2007) in wide range of clinical population (Batacan et al., 2017).

## 2.4.6.1. Effect of HIIT on Body composition

Although HIIT has been extensively studied in context to muscle physiology, limited studies have been carried out to evaluate the effect of HIIT in obese / overweight individuals. Weight loss with HIIT has been reported by several studies in overweight and obese humans (Earnest et al., 2013, Racil et al., 2013, Sartor et al., 2010, Tjonna et al., 2013, Whyte et al., 2013, Zhang et al., 2015, Fisher et al., 2015, Kemmler et al., 2014, Nybo et al., 2010, Shepherd et al., 2015, Martins et al., 2016, Sawyer et al., 2016, Sijie et al., 2012, Sim et al., 2015). Although several studies also reported non-significant change in body weight with HIIT after 5-12 weeks of training (Gillen et al., 2013, Keating et al., 2014, Kong et al., 2016, Sawyer et al., 2016), there are studies which reported weight loss ranging from 1% following 2 weeks of HIIT to 8.4% after 12 weeks of training (Sijie et al., 2012) in humans. There has been a huge variation among the data even within the studies reporting same study duration in overweight and obese individuals, for instance, seven studies have reported 12 weeks HIIT with change in body mass ranging from no change (Gillen et al., 2013, Keating et al., 2014) to up-to 8.4% weight loss (Earnest et al., 2013, Nybo et al., 2010, Sijie et al., 2012, Sim et al., 2015, Zhang et al., 2015). The difference in the results could be attributed to the difference in the HIIT protocol and also the ethnicity of the participants. Among animal studies very few studies have reported the effect of HIIT in obese animals. Two studies have reported 16% and 5% decline in weight gain in comparison to sedentary control mice after 4 (Cho et al., 2015) and 6 weeks (Lund et al., 2015) of HIIT 5 days per week.

High intensity interval training has been reported to reduce fat mass (Wewege et al., 2017). Similar to change in body weight, the reduction to fat mass ranges from nonsignificant change reported after five weeks of HIIT (Kong et al., 2016) to 9.7% after 12 weeks of training (Zhang et al., 2015). Consistent with the weight loss, change in fat mass did not correlate with duration of study and studies with similar time duration reported wide range of difference in fat mass reduction. Several studies reporting 12-week intervention period reported fat mass decline ranging from 0.5% to 9.7% (Earnest et al., 2013, Nybo et al., 2010, Racil et al., 2013, Sijie et al., 2012, Sim et al., 2015, Zhang et al., 2015).

Even in the absence of change in body mass, HIIT has been reported to reduce waist circumference (Wewege et al., 2017). Reduction in waist circumference reported ranges from a 2.2% after 16 weeks of HIIT (Kemmler et al., 2014) to 5.8% after 12 weeks of training (Zhang et al., 2015). In addition to waist, reduction in hip circumference by 1.4cm (Martins et al., 2016) and 4.5cm (Zhang et al., 2015) has also been reported after 12 weeks of training. High intensity exercise has also been reported to reduce body mass index (BMI) (Fisher et al., 2015, Racil et al., 2013, Shepherd et al., 2015, Sijie et al., 2012, Sim et al., 2015, Zhang et al., 2015). Reduction in BMI has been reported by several but not all studies and it varies from no change (Kong et al., 2016, Sawyer et al., 2016) after 5-8 weeks of HIIT to a minimum of 0.86% (Fisher et al., 2015) to a maximum of 7.8% decline following 12 weeks of training .

### 2.4.6.2. Effect of HIIT on glucose homeostasis

Single session of HIIT. In comparison to no exercise has been reported to improve glucose homeostasis (Little et al., 2014). Some studies looking at the long term effect of HIIT reported no change (Kemmler et al., 2014, Racil et al., 2013, Shepherd et al., 2015, Tjonna et al., 2013), while others reported reduced fasting glucose (Tjonna et al., 2008, Fex et al., 2015, Jeppesen and Madsen, 2015, Tjonna et al., 2011, Earnest et al., 2013). Fasting glucose is primarily a marker of hepatic insulin sensitivity (Cassidy et al., 2017). The decline in fasting glucose following HIIT training is generally less than 14% (Fex et al., 2015, Mitranun et al., 2014, Tjonna et al., 2011, Jeppesen and Madsen, 2015, Tjonna et al., 2008), indicating that HIIT by itself, is not potent enough to trigger a significant change on fasting glucose. Although it has been shown that HIIT reduces liver fat but without any significant change in fasting glucose (Cassidy et al., 2016). High intensity interval training for 8 weeks has also been reported to reduce fasting glucose by 29% in high fat diet induced obese mice (Cho et al., 2015), however similar study on HIIT for 10 weeks reported nonsignificant change in fasting glucose in same animal model (Lund et al., 2015), which could be due to difference in the intensity of exercise.

High intensity interval training has also been reported to reduce fasting plasma insulin in both humans and animals. In humans, huge variation has been found between studies in context to change in plasma insulin. Some studies reported no change (Nybo et al., 2010, Sawyer et al., 2016) where as others have reported change varying from 8.1% to 27% (Karstoft et al., 2013). In addition to fasting insulin, some studies have also reported approximately 20% improvement in insulin sensitivity measures as HOMA-IR following HIIT (Mitranun et al., 2014, Jeppesen and Madsen, 2015, Tjonna et al., 2008, Earnest et al., 2013), whereas some studies have shown no improvement (Shaban et al., 2014, Hollekim-Strand et al., 2014). Likewise, other indicators of glucose homeostasis such as oral glucose tolerance test and glucose area under the curve also showed similar inconsistences among the literature. The major reasons behind these inconsistences are study populations, exercise protocols and the degree of volunteer supervision during exercise but the most plausible is the variation in time of post-intervention measures relative to the last bout of exercise (Cassidy et al., 2017).

## 2.4.6.3. Effect of HIIT on lipid profiles

High intensity interval training has reported to reduce blood lipids. There are several studies reporting no change in blood lipid profile after HIIT (Kong et al., 2016, Tjonna et al., 2013). Some studies have reported decline in TAG ranging from 2.2% to 11.8% after 6-16 weeks of HIIT (Fisher et al., 2015, Keating et al., 2014, Kemmler et al., 2014, Racil et al., 2013, Sawyer et al., 2016, Shepherd et al., 2015, Zhang et al., 2015). Besides TAG, HIIT has also been reported to reduce total cholesterol in humans (Fisher et al., 2015, Nybo et al., 2010, Racil et al., 2013, Shepherd et al., 2015, Zhang et al., 2015). Reduction in TAG, free fatty acids and total cholesterol following 8 weeks of HIIT has also been reported in high fat diet induced obese mice (Cho et al., 2015).

## 2.4.6.4. Molecular changes induced by HIIT

Skeletal muscle plays a major role in regulating metabolism and manifest number of changes following acute or chronic HIIT. GLUT-4 content and translocation to plasma membrane is important for glucose homeostasis (Huang and Czech, 2007). In humans, six sessions of HIIT resulted in 369% increase in GLUT-4 content in vastus lateralis muscle of type-2 diabetes patients (Little et al., 2011). Although low expression of GLUT-4 is not related to insulin resistance, its higher expression is related to improved glucose uptake in the muscle (Ren et al., 1994). Consistent with this study, another 16 week HIIT study reported higher mRNA content of GLUT-4 and higher membrane bound GLUT-4 in type-2 diabetes patients, however there was no overall increase in total GLUT-4 protein content (Karstoft et al., 2014).

In addition to GLUT-4 content, HIIT has been found to increase mitochondrial content, function and markers of mitochondrial biogenesis, all of which have been found to be reduced in individuals with metabolic syndromes (Cassidy et al., 2017). Significant increase in mitochondrial capacity evidenced by an increase in citrate synthase activity and raised content of electron transport chain complexes have been found after 2 weeks of HIIT (90% maximum heart rate) in type-2 diabetes individuals (Little et al., 2011). However, another 16-weeks study reported only increase in citrate synthase mRNA content but not in citrate synthase activity after HIIT (70% VO<sub>2</sub> max) in type-2 diabetes individuals (Karstoft et al., 2014). This discrepancy in results could be due to lower intensity of intervals in this study and more over the muscle biopsies were taken 5–6 days post-training.

High intensity interval training has also been reported to increase both mRNA and protein content of PGC-1 $\alpha$  in muscle, which regulates mitochondrial biogenesis (Wu et al., 1999, Little et al., 2010). It has been suggested that ATP turnover during HIIT, differs from usual steady state condition of ATP production and could be responsible for activating the signalling cascade that leads to increase in PGC-1 $\alpha$  after HIIT (Daussin et al., 2008).

Calcium ion  $(Ca^{2+})$  handling by sarcoplasmic reticulum plays an important role in muscle fatigue (Cassidy et al., 2017). Enhanced ca2+ reuptake has been reported

following HIIT but not moderate intensity interval training in individuals with metabolic syndrome (Tjonna et al., 2008). Increased ca2+ reuptake by 56-60% after HIIT significantly improve the work capacity of muscle and contributes to the improvement in fitness due to this type of exercise training. Although this study (Tjonna et al., 2008) concluded that HIIT provide more potent stimulus than moderate intensity exercise for skeletal muscle adaptations, definite conclusion cannot be drawn on the basis of single study with sample size of 32 participants and therefore needs further investigations.

# **CHAPTER 3**

# **MATERIAL AND METHODS**

## **3.1. OVERVIEW**

This chapter describes general methods used in three studies presented in this thesis. The central aim of the thesis was to investigate the effect of combined IF and HIIT on changes in body composition, metabolic health and molecular changes in metabolically important tissue namely adipose and skeletal muscle in high fat and sugar diet induced obese mice. The first study is designed to investigate the anti-obesity effect of IF and HIIT and its combination and their effect on glucose homeostasis and blood lipid markers. The second and third study uses various molecular tools to investigate the changes at molecular level (mRNA and miRNA) in adipose and muscle tissue induced by IF, HIIT and combination of these two interventions. General layout of the study is given in figure 3.1. Details of methods specific to the three studies are described in the respective chapters.

#### 3.2. ANIMALS

#### 3.2.1 Procurements and Housing:

One hundred, 8-weeks old, C57BL/6 mice (Wt: 22.2±2.5g Males, 17.6±1.7g Females) were acquired from the Animal Resource Centre (Murdoch, WA, Australia) and housed in animal holding facility at Werribee campus, Victoria University. Mice were housed in groups of 5, at a constant temperature (22°C), under a 12:12-h light-dark photoperiod with *ad libitum* access to food and water (other than fasting periods) in accordance with ethical approval obtained from the Victoria University Animal Ethics

Committee, conforming to the Australian code of practice for the care and use of animal for scientific purposes.



Figure 3.1. General outlay of the study.

# 3.2.2 <u>Diet</u>

Before starting the high fat diet, animals were given one week of adaptation to animal house and were fed mice chow (Figure 3.2(A)) (meat free rat and mouse diet, Specialty Feeds, Glen Forrest, WA, Australia) and potable water. Animals were gradually introduced to the special diets over a period of one week by increasing the proportion of special diet mixed with chow, starting with 1:3 (high fat diet: chow diet) for first 2 days, followed by 1:1 for next 2 days and 3:1 for next 3 days and eventually moving to complete high fat diet. Diet induced obese group had *ad libitum* access to high fat diet (HF) (Figure 3.2(B)) (59% energy from fat, SF03-002, speciality feeds, WA, Australia) and high sugar (S) water (30% sugar (weight/volume)) for rest of the study period. Nutritional parameters of both high fat and chow diet are given in table 3.1



Figure 3.2. Animal diets-(A) chow diet and (B) High fat diet.

Nutritional Parameters	High fat diet (SF03-002)	Chow diet
Protein	19.4%	20%
Total Fat	36.0%	4.8%
Total Digestible Carbohydrate	-	59.4%
Crude Fibre	4.7%	4.8%
AD Fibre	4.7%	7.6%
Digestible energy	22.8 MJ/Kg	14.0 MJ/Kg
% Total calculated digestible energy from lipids	59.0%	12%
% Total calculated digestible energy from protein	15.0%	23%

Table 3.1. Calculated nutritional parameters of diets

After 12-weeks of HF/S feeding, mice were divided into five groups: obese baseline control (OBC; Males=8, Females=10), no intervention (CON; Males=7, Females=9); intermittent fasting (IF; Males=8, Females=10), high intensity intermittent exercise (HIIT; Males=8, Females=10) and a combination of the dietary and exercise intervention (IF+HIIT; Males=8, Females=10). Mice from OBC group were killed after 12 weeks of (HF/S) feeding period to establish baseline measures for some variables

compared to the remaining groups. The remaining mice were maintained on the HF/S diet for an additional 12 weeks. After 24 weeks of HF/S feeding and intervention, all remaining mice were culled and tested as described below. Eleven male mice died during the dietary and exercise intervention period, presumably due to the complications of diabetes as a result of the HF/S diet. One female mouse did not gain weight in the control group and was excluded from the study.

#### 3.2.3 Assessment of Caloric Intake

Food and sugar water intake of animals from each cage was measured weekly and average caloric intake per animal per day was calculated. Due to housing constrains of ethical approval and to avoid added stress, animals from the non-fasting groups (control and HIIT) were housed together and animals from fasting groups (IF and IF+HIIT) were housed together. The average caloric intake of the non-fasting groups (control and HIIT animals) was compared to fasting groups (IF and IF + HIIT animals) to determine if intermittent fasting resulted in over compensation of food intake on non-fasting days over the 12-weeks intervention period.

## **3.3.** INTERVENTIONS

All intervention groups underwent an 8-day cycle instead of the 7-day weekly cycle. This was designed to give at least one day break between fasting day and exercise day for combination (IF+HIIT) group and ensure animal wellbeing. To ensure uniformity between the groups, IF and HIIT groups underwent the same 8-day cycle. However, the total period of interventions was restricted to 12 calendar weeks.

#### 3.3.1 Intermittent Fasting Protocol

Mice from IF and IF+HIIT group were deprived of food on two alternate days of an eight day cycle for 12-weeks. During the fasting period, mice were moved to clean cages with potable water available *ad libitum* and returned to their original cages 24 hours later. On non-fasting days, mice had *ad libitum* access to the HF/S diet (Castello et al., 2010, Varady et al., 2007). Mice were always fasted on day 2 and 7 of the 8-day cycle.

## 3.3.2 <u>High Intensity Interval Training Protocol</u>

#### 3.3.2.1. Acclimatization to treadmill exercise

All groups were acclimatized to treadmill running through five gradual treadmill training familiarization sessions over a period of 2 weeks prior to intervention commencement according standard to operating procedures for treadmill exercise and Victoria university animal ethics guidelines. For the first two acclimatization sessions, mouse was placed on the treadmill (Panlab, Barcelona, Spain) (Figure 3.3) for 5 min. The treadmill was then turned on at 3cm/sec for 5 min followed by gradual increase in speed over next 5 minutes from 3cm/sec to 12cm/sec. For



Figure 3.3. Mice treadmill (A), Treadmill control (B), mouse on treadmill (C).

third and fourth acclimatization sessions, treadmill speed was gradually increased over 5 minutes from 3cm/sec to 12cm/sec, followed by gradual increase in speed over next 10 minutes from 12cm/sec to 20cm/sec. During the final session, speed was gradually increased over 15 minutes from 3cm/sec to 25cm/sec. To minimise the effects of the training adaptation during the acclimatization sessions, all mice, irrespective of their group, underwent the 2-week familiarisation training.

## 3.3.2.2. Top Speed Test

The protocol for top speed determination was as follows: mice performed an incremental warm up period for 5 min reaching 8 cm/sec on mice treadmill. The speed was slowly increased from 8 to 35 cm/sec and was held at 35 cm/sec for 20 sec and then returned back to 8 cm/sec for 40 sec. Following this, the speed was slowly increased to 40 cm/sec and maintained for 20 sec followed by another 40 sec of active rest at 8 cm/sec. The 20 sec running and 40 sec active rest protocol was repeated with increasing running speed by 5 cm/sec until the mouse was unable to continue running. The running speed prior to this was recorded as the top speed. The average top speed for all mice was  $59\pm 2$  cm/sec.

## 3.3.2.3. HIIT Training

The HIIT protocol was a slightly modified protocol used by Tuazon et al. (2015) with the relatively low volume and high intensity shown to have positive effects on body composition and metabolic health (Lund et al., 2015). Briefly, the HIIT training sessions consisted of a 5 min warm up period at 8 cm/sec, followed by 6-8, 20 sec bouts of treadmill running at the previously determined top speed of the individual animal, interspersed by 40 sec active rest at 8 cm/s. To implement overload, the number of intervals was increased from 6 at the beginning of the training intervention to 8 at the end of the eighth HIIT session. HIIT was carried out on three non-consecutive days per 8-day cycle. To encourage the mice to run on the treadmill at their maximum speed, an air puff device was used if the mice appeared to slow down.

#### *3.4.* Body Weight and Composition Assessment

Body weight was measured weekly on laboratory weighing balance. Body composition (fat and lean mass) was determined using EchoMRI (EchoMRI- body

composition analyser, Echo Medical Systems, Houston, USA) (Figure 3.4) before and after 10-weeks of diet and/or exercise intervention. Briefly, machine was calibrated with canola oil. Once calibrated, mouse was inserted into the mouse tunnel (Figure 3.4 (B)) which was then placed into the EchoMRI machine. The measurement was then started through a computer software machine. attached to the The body composition (Fat mass, lean mass and water) appears on the screen in the form of a table. Once the measurement is complete, mouse was placed back into the cage.



Figure 3.4. EchoMRI machine (A) and mouse tunnel (B).

## **3.5.** CULLING AND TISSUE COLLECTION

To minimise any acute transient effects from either fasting and exercise treatment (Joslin et al., 2017), a period of 48 h was given between the last fasting period and exercise session and any physiological measurement or sample collection for all animals. Mice were anaesthetised with intraperitoneal injection of pentobarbitone (60 mg/kg). Once the mouse was fast asleep, extensor digitorum longus (EDL) muscle from the left leg and both soleus and EDL muscle from right leg were removed surgically. Both muscles from right leg were preserved in optimal cutting temperature (OCT) compound for histological analysis and EDL muscles from left leg were snap frozen in liquid nitrogen and stored in

-80°C freezer until further analysis. After muscle collection, the blood was collected by cardiac puncture and plasma was prepared using EDTA as anticoagulant and centrifugation at 2000g for 15 min. The abdominal cavity was opened and epididymal and periovarian adipose tissue was collected and snap frozen in liquid nitrogen and stored in -80°C freezer until further analysis. Liver, pancreas, heart, kidneys and spleen were removed and weighed on a laboratory scale.

# 3.6. ISOLATION OF RNA

Total RNA for mRNA and miRNA expression from adipose and muscle tissue was extracted using miRNeasy mini kit (Qiagen). Sample (adipose (100mg)/muscle (10mg)) was homogenised with 700µl of Qiazol lysis reagent in TissueLyser (Qiagen) for 5 min at 50Hz and incubated at room temperature for 5 min. To this homogenate 140µl of chloroform was added and vortexed for 15 sec followed by incubation at room temperature for 2-3 min. The homogenates were centrifuged at 12000g for 15 min at 4°C. Following centrifugation, the upper aqueous phase was transferred to a new collection tube and 525µl of ethanol was added to it and mixed thoroughly. Up to 700µl of this mixture was transferred to RNeasy mini column in a 2ml collection tube followed by centrifugation at 8000g for 15sec at room temperature and flow-through was discarded. This step was repeated with the rest of the mixture and flow- through was again discarded. To the column, 700µl of RWT buffer was added followed by centrifugation at 8000g for 15s at room temperature and flow-through was discarded. After this, 500µl of RPE buffer was added to the column and centrifuged at 8000g for 15s at room temperature and flowthrough was discarded. This step was repeated once more. The column was then placed in a fresh 1.5ml collection tube and 30µl of RNase free water was added to the column followed by centrifugation at 8000g for 1 minute at room temperature. This step was repeated with same collection tube. The RNA content was measured using nano-drop spectrophotometer (DeNovix DS-11, DeNovix, USA). RNA extract was aliquoted and stored in -80°C freezer until further analysis.

## 3.6.1 Assessment of RNA quality

Agarose gel (1%) was prepared in Tris-acetate-EDTA (TAE) buffer (1X), mixed with SYBR<sup>™</sup> Green II RNA gel stain (Invitrogen) and casted in a horizontal gel casting tray with a 15 well comb. RNA (400ng from adipose tissue or 300ng from muscle), mixed with loading dye in 3:1 ratio, were heated at 70°C for 10 min and cooled down on ice for 3 min. Agarose gel was placed in a horizontal gel running tank and filled with TAE buffer (1X). Samples (RNA + loading dye) were loaded into wells in duplicate and gel was run at 80 volts for 45 min. After this, gel was taken out of the gel tray and imaged in gel image and analysis system (ChemiDoc XRS system, Biorad). The 28s and 18s rRNA band ratio was calculated for each sample and ratio between 1.3 to 2 was taken as measure for good RNA quality. RNA extraction was repeated for samples with 28s:18s band ratio less than 1.3. A representative image of RNA agarose gel is shown in Figure 3.5.



Figure 3.5. Representative image of agarose gel showing two rRNA bands.

# 3.7. MRNA REVERSE TRANSCRIPTION AND ASSESSMENT OF GENE EXPRESSION

## 3.7.1 Synthesis of cDNA-reverse transcription

RT<sup>2</sup> first strand kit (Qiagen) was used to synthesize complementary deoxyribonucleic acid (cDNA). RNA sample (400ng), mixed with genomic DNA elimination buffer (2µl) and water (to make up the volume to 10µl) in a 1.5ml tube, was incubated at 42°C for 5 min and immediately placed on ice for 1 minute. To each tube, 10µl of reverse transcription mix (5X buffer BC3, control P2, RE3 reverse transcriptase mix and nuclease free water in a ratio of 4:1:2:3µl) was added and mixed with pipetting up and down. The tubes were incubated at 42°C for 5 min. The tubes were then transferred to ice and 91µl of water was added to each tube and mixed properly. The cDNA samples were aliquoted and stored in -20°C freezer until further analysis.

#### 3.7.2 <u>Gene expression</u>

Gene expression was determined using Custom  $RT^2$  PCR array (Qiagen). Each column of 384- well plate consist of 11 genes of interest, 2 housekeeping genes, one genomic DNA control, one positive PCR control and one reverse transcription control. Primers for both genes of interest and housekeeping genes were synthesized by the manufacturer (Qiagen) based on the list and gene symbols supplied by us. Details of genes and primers are provided separately for adipose and muscle tissue in chapter 5 and 6 respectively. Primers were absorbed on the bottom of 384 well optical micro-plate with one primer pair per well. To each well, 10µl PCR mix ( $RT^2$  SYBR green master mix with ROX (passive normalization dye), cDNA and water in a ratio of 5:0.8:4.2µl) was added. The plate was sealed with optical adhesive film and centrifuged at 1000g for 1 min to remove bubbles. The plate was placed in real time thermal cycler (QuantStudio 7 Flex Real-Time PCR System) with cycling conditions consisting of 1 cycle of 10 min at 95°C to activate *Taq* polymerase and 40 cycles of 15 sec at 95°C for denaturation and 1 minute at 60°C for annealing. The ramp rate was adjusted at 1°C/sec. The fluorescence was measured after each cycle and baseline was defined using automated baseline option in thermal cycler. Cycle threshold ( $C_T$ ) values generated by the thermal cycler were exported to excel file and change in gene expression expressed as fold change was calculated as:

 $\Delta C_T = C_{T(\text{gene of interest})} \text{-} C_{T(\text{Housekeeping gene)}}$ 

 $\Delta \Delta C_T = \Delta C_{T(intervention)} - \Delta C_{T(control)}$ 

Fold change =  $2^{-\Delta\Delta CT}$ 

# 3.8. MIRNA REVERSE TRANSCRIPTION AND ASSESSMENT OF MIRNA EXPRESSION

#### 3.8.1 Synthesis of cDNA-reverse transcription

MiScript II RT kit (Qiagen) was used to synthesize cDNA. RNA sample (250ng) was mixed with 4µl of 5x miScript HiSpec Buffer, 2µl of 10x miScript Nucleics mix, 2µl of miScript reverse transcriptase mix and water (to make up the volume to 20µl) in a 1.5ml tube. The tubes were incubated at 37°C for exactly 60 min and the reaction was stopped immediately by incubating at 95°C for 5 min. The tubes were then transferred to ice and 200µl of water was added to each tube and mixed properly. The cDNA samples were aliquoted and stored in -20°C freezer until further analysis.

## 3.8.2 MiRNA expression

MiRNA expression was determined using Custom RT<sup>2</sup> PCR array (Qiagen). Each column of 384- well plate consist of 5 adipose tissue miRNAs of interest, 3 skeletal muscle miRNAs of interest, 2 housekeeping miRNAs for each tissue, one positive PCR

control and one reverse transcription control for each tissue. Primers for both, miRNAs of interest and housekeeping miRNAs were synthesized by the manufacturer (Qiagen) based on the list and gene symbols supplied by us. Details of miRNAs and primers are provided separately for adipose and muscle tissue in chapter 5 and 6 respectively. Primers were absorbed on the bottom of 384 well optical micro-plate with one primer pair per well. To each well, 10µl PCR mix (2x QuantiTech SYBR Green PCR master mix, 10x miScript Universal Primer, RNAse free water and template cDNA, in a ratio of 5.5:1.1:3.4:1.0µl) was added. The plate was sealed with optical adhesive film and centrifuged at 1000g for 2 min to remove bubbles. The plate was placed in real time thermal cycler (QuantStudio 7 Flex Real-Time PCR System) with cycling conditions consisting of 1 cycle of 15 min at 95°C to activate HotStarTaq DNA polymerase and 40 cycles of 15 sec at 94°C for denaturation, 30 sec at 55°C for annealing and 30 sec at 70°C for extension. The ramp rate was adjusted at 1°C/sec. The fluorescence was measured after each cycle and baseline was defined using automated baseline option in thermal cycler. Cycle threshold  $(C_T)$  values generated by the thermal cycler were exported to excel file and change in miRNA expression expressed as fold change was calculated as:

$$\begin{split} \Delta C_T &= C_{T(miRNA \text{ of interest})} \text{-} C_{T(Housekeeping miRNA)} \\ \Delta \Delta C_T &= \Delta C_{T(intervention)} \text{-} \Delta C_{T(control)} \\ \text{Fold change} &= 2^{-\Delta \Delta CT} \end{split}$$

## **3.9.** STATISTICAL ANALYSIS

Data is presented as mean  $\pm$  standard deviation (SD) in tables and figures presented. Details of specific statistical analysis are given in individual chapters.

# **CHAPTER 4**

# INTERMITTENT FASTING WITH OR WITHOUT EXERCISE PREVENTS WEIGHT GAIN AND IMPROVES LIPIDS IN DIET-INDUCED OBESE MICE

Robin A. Wilson <sup>1,2</sup>, William Deasy <sup>1,2</sup>, Christos G. Stathis <sup>1</sup>, Alan Hayes <sup>1,2</sup> and Matthew B. Cooke <sup>1,3,\*</sup>

- <sup>1</sup> College of Health and Biomedicine, Victoria University, Melbourne, VIC 3000, Australia;
- <sup>2</sup> Australian Institute for Musculoskeletal Science (AIMSS), Western Health, Melbourne, VIC 3021, Australia
- <sup>3</sup> Department of Health and Medical Sciences, Swinburne University of Technology, Melbourne, VIC 3122, Australia

\* Correspondence: mbcooke@swin.com.au or mbcooke@swin.edu.au; Tel.: +61-3-9214-5560

Journal: Nutrients, Received: 2 February 2018; Accepted: 8 March 2018; Published: 12 March 2018

# **4.1.** INTRODUCTION

Obesity is a complex multifaceted disease resulting from the interplay between genetics and lifestyle, including economic growth, modernization, and urbanization (van Vliet-Ostaptchouk et al., 2012). The rapid rise in obesity prevalence appears to be a reflection of the changes in dietary and behavioural patterns, with eating habits shifting to greater consumption of energy-dense foods that are high in fats and sugars, while at the same time, levels of physical activity are decreasing. These differences in energy intake and expenditure, often referred to as energy balance, have direct implications for weight regulation, with even small deviations in daily energy balance resulting in large body weight changes over the long term (Romieu et al., 2017). Interventions that target and reverse these small deviations in energy balance may be an effective tool in reducing body weight and adiposity, but also help maintain a stable body weight over the longer period.

Diet and exercise interventions aiming to shift the energy balance towards negative by either decreasing calorie intake and/or increasing physical activity have shown to be effective for weight loss (Strasser et al., 2007). Many iterations of such dietary and physical activity interventions have been proposed, but intermittent fasting (IF) and high intensity interval training (HIIT) have recently been purported as effective strategies (Türk et al., 2017, Tinsley and La Bounty, 2015). Intermittent fasting or alternate day fasting (ADF) is a dietary regime that refers to short periods of intense energy restriction (75-100% reduced caloric intake on fasting day) followed by 'normal' eating on non-fasting days (Barnosky et al., 2014, Heilbronn et al., 2005b). Studies have shown that short to long term intermittent fasting can reduce body weight and fat mass, fasting glucose and insulin levels, and improve insulin sensitivity and lipid profiles (Barnosky et al., 2014, Azevedo et al., 2013). Although calorie restriction (CR) and IF demonstrate comparable benefits on body composition and other health markers, adherence to IF is typically greater compared to the conventional daily CR diets and thus could be more effective over the longer term (Azevedo et al., 2013).

High intensity interval training is promoted as a superior and time efficient method for reducing body and fat mass and other biomarkers of chronic diseases compared to moderate-intensity continuous training. HIIT refers to brief bouts of vigorous intensities (80-90% VO2 max) interspersed by relatively longer bouts of low intensity active recovery or passive recovery (Gibala et al., 2012, Trapp et al., 2008). The effectiveness of HIIT is its ability to produce rapid physiological adaptations that may promote greater energy deficit post-exercise, independent of the changes in total physical activity energy expenditure. Our lab previously reported higher plasma hypoxanthine accumulation and urinary purine base excretion (indirect markers of energy loss from the

muscle) following high intensity exercise compared to continuous exercise (Gerber et al., 2014). These losses could indicate the need for greater energy input for subsequent restorative processes (via intramuscular purine *de novo* replacement) during the recovery period and may in part contribute to the negative energy balance that leads to fat loss following long term high intensity training programs (Racil et al., 2013, Earnest et al., 2013). Our findings and others (Gillen et al., 2013, Sartor et al., 2010) have suggested that training programs that include high intensity and low volume exercise could be effective at creating metabolic disturbances that can result in enhanced weight loss in the form of reduced adipose tissue, but also promote health and fitness benefits normally seen over the longer term.

Manipulating the energy balance equation through exercise and diet are effective strategies to improve body composition and manage lifestyle-related metabolic diseases. However, protocols that are characterised by severe energy restriction or large energy expenditure such as IF and HIIT, respectively, may induce greater physiological changes and/or further stimulate already maximal adaptation rates, independent of changes in energy balance. The current study was designed to investigate the effect of IF and HIIT, alone and in combination, on anthropometric and metabolic health parameters in a model of diet-induced obese mice. We hypothesised that, the combined interventions will result greater prevention of weight gain in the form of fat mass, preserving lean muscle mass despite IF, reductions in fasting glucose levels and markers of insulin sensitivity and glucose tolerance, and improvement in lipid profile compared to either dietary and exercise intervention alone.

### 4.2. MATERIALS AND METHODS

## 4.2.1 Animals

Details of animal procurement, housing and diet are mentioned in chapter 3 section 3.2. This study compared five different groups namely: obese baseline control (OBC; Males=8, Females=10), no intervention control (CON; Males=7, Females=9); intermittent fasting (IF; Males=8, Females=10), high intensity intermittent exercise (HIIT; Males=8, Females=10) and a combination of the dietary and exercise intervention (IF+HIIT; Males=8, Females=10).

### 4.2.2 Assessment of Caloric Intake

Food and sugar water intake of animals from each cage was measured weekly using a laboratory scale and average caloric intake per animal per day was calculated. Due to ethical approval, animals from the control and HIIT groups were housed together and animals from IF and IF+HIIT groups were housed together. The average calorie intake from the CON + HIIT animal ("non-fasting" group) was compared to IF and IF+HIIT animals ("fasting" group) to determine if IF resulted in over compensation of food intake on non-fasting days over the 12-week intervention period.

#### 4.2.3 Interventions

Details of interventions are given in chapter 3, section 3.4.

## 4.2.4 Body Weight and Composition Assessment

Details of body composition measurement using Echo-MRI are given in chapter 3, section 3.5

## 4.2.5 Assessment of Glucose Tolerance (AUC)

Fasting blood glucose and glucose tolerance was analysed before and after 10weeks of diet and/or exercise intervention. Last measurement was done at week-10 instead of week-12 (experimental endpoint), to give a period of 2 weeks between the intraperitoneal glucose tolerance test (IGTT) and culling, since it was found in the preliminary studies that mice show distress after IGTT and take few days to recover. A day before IGTT, body weight and body composition was measured. At 7pm animals are transferred to new cages with potable water available *ad libitum*. Next morning, at 7am, Fasting glucose was measured using handheld glucometer (Accuckeck GO glucometer) from blood collected by snipping the tail tip (1-2mm) following an overnight fast (12hrs). Mice were given intraperitoneal injection of glucose (1.5g/kg body weight) using an insulin syringe (BD ultra-fine, 1ml, 29G) and blood was collected by massaging the tail from base towards the tip and reading was taken at 5, 15, 30, 45, 60, 90 and 120 min following the glucose administration. Glucose tolerance was calculated as glucose area under the curve (AUC) using trapezoidal method (Andrikopoulos et al., 2008).

## 4.2.6 Blood and organ collection

Mice were anaesthetised with intraperitoneal injection of pentobarbitone (60 mg/kg). Once the mouse was fast asleep, extensor digitorum longus (EDL) and soleus muscle from right leg were removed surgically and weighed. Blood for plasma measurements was collected by cardiac puncture. Plasma was prepared using EDTA as anticoagulant and centrifugation at 2000g for 15 min. The abdominal cavity was opened and liver, pancreas, heart, kidneys and spleen were removed and weighed on a laboratory scale. Details are given in chapter 3, section 3.7.

#### 4.2.7 Assessment of Plasma Insulin, TAG, LDL and HDL Levels

## 4.2.7.1. Plasma insulin

Plasma insulin was estimated using Ultra-sensitive mouse insulin ELISA kit (Crystal Chem, Inc, IL, USA). Frozen plasma samples were thawed on ice. To anti-insulin antibody-coated microplate, 95 $\mu$ l of sample diluent was added followed by 5 $\mu$ l of plasma sample or standard insulin (0.1 to 12.8ng/ml) in duplicate. The plate was covered with the cover and incubated for 2 hours at 4°C. After 2 hours, all the contents of the well were aspirated and plate was washed five times with wash buffer. After washing, 100 $\mu$ l of horse radish peroxidase-conjugated anti-insulin antibody was added. The plate was covered with the cover and incubated at room temperature. After 30 min, all the contents of the well were aspirated and plate was washed seven times with wash buffer. Immediately after washing, 100 $\mu$ l of the 3, 3′, 5, 5′-tetramethylbenzidine substrate solution was added and plate was incubated at room temperature for 40 min in dark. Enzyme reaction was stopped by adding 100 $\mu$ l of enzyme reaction stop solution. The absorbance was measure at 450nm and subtracted from absorbance at 630nm using a plate reader (SpectraMax i3x). Insulin concentration was calculated using the standard curve.

### 4.2.7.2. Triglycerides

Plasma TAG concentration was measured using colorimetric quantification assay kit (ABCAM, Cambridge, Massachusetts). To 96 well microplate,  $50\mu$ l of TAG standard (2-10nmol) or sample ( $3\mu$ l sample +  $47\mu$ l assay buffer) was added in duplicate. To each well,  $2\mu$ l of lipase enzyme was added. Plate was incubated at room temperature for 20 min to convert triglycerides to glycerol and fatty acid. This was followed by addition of 50µl of reaction mixture (buffer, triglyceride probe and enzyme mix in a ratio of 46:2:2) and incubating the plate for 60 min at room temperature in dark. After 60 min, absorbance was measured at 570nm using a plate reader (SpectraMax i3x). TAG concentration was calculated using the standard curve.

## 4.2.7.3. Low density lipoproteins

Plasma LDL concentration was measured using mouse LDL-cholesterol kit (Crystal Chem, Inc, IL, USA). To a 96 well microplate, 3µl Plasma sample or calibrator (0.53µg to 3.18µg) or blank was added in duplicate followed by addition of 225µl of reagent 1 to each well. The plate was covered and incubated at room temperature for 5 min followed by absorbance at 600nm. After this, 75µl of reagent 2 was added to each well. The plate was covered and incubated at room temperature for 5 min followed by absorbance at 600nm. After this, 75µl of reagent 2 was added to each well. The plate was covered and incubated at room temperature for 5 min followed by absorbance at 600nm.

## 4.2.7.4. High density lipoproteins

Plasma HDL concentration was measured using mouse LDL-cholesterol kit (Crystal Chem, Inc, IL, USA). To a 96 well microplate, 3µl Plasma sample or calibrator (0.29µg to 1.78µg) or blank was added in duplicate followed by addition of 225µl of reagent 1 to each well. The plate was covered and incubated at room temperature for 5 min followed by absorbance at 600nm. After this, 75µl of reagent 2 was added to each well. The plate was covered and incubated at room temperature for 5 min followed by absorbance at 600nm. After this, 75µl of reagent 2 was added to each well. The plate was covered and incubated at room temperature for 5 min followed by absorbance at 600nm.

#### 4.2.8 Homeostatic Model Assessment of Insulin Resistance (HOMA-IR)

The homeostatic model assessment of insulin resistance (HOMA-IR) was determined as fasting insulin [mU/L] x fasting glucose [mmol/L])/22.5 (Seimon et al., 2016). The plasma insulin (ng/ml) was measured as stated above and the units were converted to mU/L by dividing the values by 0.0347. Plasma glucose was measured by handheld glucometer (Abbott Freestyle Optium).

## 4.2.9 Assessment of Muscle Strength

Muscle strength was measured before and after 10-weeks of diet and/or exercise intervention using a grip strength test as previously described by Deacon (2013). Briefly, a grip test apparatus was used which consisted of a stainless steel wire ball connected to a series of steel chain links (Figure 4.1), varying from one (15g) to six (74g). Mouse was held by the base of its tail and allowed to grasp the grip test apparatus lying on a lab bench with its forepaws. Mouse was lifted until the apparatus was raised from the bench. Holding the apparatus at a specific weight for 3 sec on 5 consecutive occasions was deemed a successful lift and the mice would proceed to next heavier weight. If the mice failed to achieve the 3 sec target for 5 consecutive times, the trial was stopped and mouse was assigned the weight/time achieved. A 30 sec rest between lifts was used. The final score was calculated by the number of





Figure 4.1. Grip strength apparatus (A) mouse performing grip test (B).

links (1 to 6) in the heaviest weight held for 3 sec multiplied by time i.e. 3, plus the number of sec heaviest weight was held for less than 3 sec. To calculate absolute muscle strength, the total score was divided by the respective muscle mass. This method was adopted to measure the changes in muscle strength rather than the inverted screen test often used for testing grip strength which is affected by total body weight.

## 4.2.10 Statistical analysis

Data is presented as mean  $\pm$  standard deviation (SD). Delta changes (pre- and post-intervention) in body weight and body composition (fat mass and lean mass), fasting blood glucose, glucose tolerance and muscle strength were analysed using 4 (CON, IF, HIIT, IF+HIIT) × 2 (0 and 10-weeks) repeated measures analysis of variance (ANOVA) with Tukey's post hoc analysis. Since baseline measurements of plasma insulin, glucose and LDL, HDL and TAG levels were not taken for each intervention group, analysis of delta changes for plasma insulin, HOMA-IR and lipid panels were performed using the mean average of group 1 (HF/S 'obese' control values) as the "baseline" minus the post-intervention values of the respective intervention groups. These delta values were analysed using 4 (CON, IF, HIIT, IF+HIIT) × 2 (0 and 12-weeks) repeated measures ANOVA with Tukey's post hoc analysis. One way ANOVA with Tukey's post hoc analysis was also performed on plasma insulin, HOMA-IR, lipid panels and organ weights at 12-weeks. Baseline variables and average caloric intake between groups was analysed using an independent students' t-test. An alpha level of 0.05 was adopted throughout to reduce Type I statistical errors.

## 4.3. **RESULTS**

## 4.3.1 Effects of 12-weeks of the HF/S diet on body weight of all animals

A significant increase in body weight was observed in male and female mice following the initial 12-weeks consumption of the HF/S diet [Male – Baseline:  $22.2\pm2.5$ g, 12-week:  $36.2\pm2.1$ g (~63%<sup>†</sup>); Female- Baseline:  $17.6\pm1.7$ g, 12-week:  $28.5\pm1.5$ g (~62%<sup>†</sup>)] (Figure 4.2).



Figure 4.2. Weight gain (g) over the 12-weeks high fat/sugar diet feeding period in (A) male and (B) females. Data is presented as mean of each group and error bars represent standard deviation.

#### 4.3.2 Effects of IF and/or HIIT on body weight and composition

Body weight and composition measured before and after 10-weeks of diet and/or exercise intervention are presented in Figure 4.3 and 4.4.

## 4.3.2.1. Body weight and fat mass

Following the initial 12-weeks of HF/S diet, body weight and fat mass continued to significantly increase over time in the CON and HIIT groups for both male and female mice (p<0.05), and IF female group only (p<0.05). After 10-weeks of the diet and/or exercise intervention, male mice displayed significantly lower body weights and reduced fat mass in the IF and IF+HIIT groups compared to the CON (p<0.05) and HIIT groups (p<0.05) (Figure 4.3(A) and 4.4(A)). In the female mice, the IF + HIIT group displayed significantly lower body weights and reduced fat mass compared to CON (p<0.05), IF (p<0.05) and HIIT groups (p<0.05) (Figure 4.3(B) and 4.4(B)).



Figure 4.3. Change in body weight over the 10-weeks intervention period in (A) male and (B) females. Data is presented as mean of each group and error bars represent standard deviation.
# 4.3.2.2. Lean mass

Lean mass significantly increased over the 10-weeks diet and/or exercise intervention period in the CON group for both male and female mice (p<0.05) and in the HIIT and IF+HIIT female groups only (p<0.05) (Figure 4.4A and 4.4B). No significant differences between intervention groups were observed at the end of the diet and/or exercise intervention period.



Figure 4.4. Change in body composition (g) in males (A), Females (B), after 10-weeks of intervention. Bars indicate means and error bars represent standard deviation. \* significant change over time,  $\delta$  significantly different from CON,  $\Phi$  significantly different from HIIT,  $\Psi$  significantly different from IF,  $\xi$  significantly different from IF+HIIT.

# 4.3.3 Effects of IF on caloric intake

No significant differences were observed in average caloric intake during the intervention period between the fasting groups (CON and HIIT) and non-fasting groups (IF and IF+HIIT) in both male ( $14.63\pm1.47$  kcal/mouse/day versus  $13.23\pm1.62$  kcal/mouse/day, p<0.05) and female mice ( $11.93\pm1.42$  kcal/mouse/day n versus  $11.43\pm1.58$  kcal/mouse/day, p<0.05).



Figure 4.5. Caloric intake over the 12-weeks intervention period in (A) males and (B) females. Data is presented as mean of each group and error bars represent standard deviation.

# 4.3.4 <u>Effect of IF and/or HIIT on blood glucose</u>, plasma insulin, HOMA-IR and <u>AUC</u>

# 4.3.4.1. Fasting blood glucose

Fasting blood glucose levels significantly increased in the CON and IF male groups over the 10-weeks diet and/or exercise intervention period (p<0.05, Figure 4.6 (A)). At the end of the 10-weeks diet and/or exercise intervention period, the IF group for both male and female mice displayed significantly higher fasting glucose levels compared to the other 3 groups (p<0.05).

# 4.3.4.2. Plasma Insulin

Plasma insulin levels significantly increased over the 12-weeks diet and/or exercise intervention period in both male and female IF groups (p<0.05) and female CON (p<0.05) and IF+HIIT (p<0.01) groups only (Figure 4.6 (B)). No significant differences between groups were observed at the end of the diet and/or exercise intervention period.



Figure 4.6. Change in (A) fasting glucose and (B) insulin over the 10 or 12-weeks period in males and females. Bars represent means, error bars represent standard deviation. \* significant change over time,  $\delta$  significantly different from CON,  $\Phi$  significantly different from HIIT,  $\Psi$  significantly different from IF,  $\xi$  significantly different from IF,  $\xi$  significantly different from IF+HIIT.

# 4.3.5 Insulin Resistance (HOMA-IR) and Glucose Tolerance (AUC)

HOMA-IR, a marker of insulin resistance significantly increased over the 12weeks diet and/or exercise intervention period in the male and female IF groups (p<0.05, Figure 4.7). However, similar to changes in plasma insulin, female mice also showed a significant increase in HOMA-IR in the CON (p<0.05) and IF+HIIT groups (p<0.05). In addition, female mice demonstrated a significant increase in AUC (Figure 4.8, 4.9 and 4.10), a marker of glucose tolerance in IF and IF+HIIT groups over a 10-weeks diet and/or exercise interventions period (p<0.05, Figure 4.8, 4.9 and 4.10). No significant differences were observed at the end of the diet and/or exercise intervention period (Figure 4.10).



Figure 4.7. HOMA-IR changes from 'obese' baseline in males and females after 12-weeks of diet and/or exercise intervention. Bars represent means, error bars represent standard deviation and \* indicates significant change over time.



Figure 4.8. Glucose tolerance curves of (A) CON, (C) HIIT, (E) IF, (G) IF+HIIT group and glucose area under the curve of (B) CON, (D) HIIT, (F) IF, (H) IF+HIIT group in males before and after 10-weeks of intervention. Bars represent means, error bars represent standard deviation.



Figure 4.9. Glucose tolerance curves of (A) CON, (C) HIIT, (E) IF, (G) IF+HIIT group and glucose area under the curve of (B) CON, (D) HIIT, (F) IF, (H) IF+HIIT group in females, before and after 10-weeks of intervention. Bars represent means, error bars represent standard deviation and \* indicates significant change over time.



Figure 4.10. Change in glucose area under the curve (AUC) in males and females after 10-weeks of diet and/or exercise intervention. Bars represent means, error bars represent standard deviation and \* indicates significant change over time.

#### 4.3.6 Effect of IF and/or HIIT on lipid profiles

Plasma LDL, HDL and TAG levels measured before and after 12-weeks of diet and/or exercise intervention is presented in Table 4.1.

In female mice, plasma LDL levels significantly decreased over the 12-weeks diet and/or exercise intervention period in all groups (p<0.05). Conversely, HDL significantly increased over 12-weeks period, but only in the HIIT and IF groups (p<0.05). In male mice, LDL levels significantly decreased over the 12-weeks diet and/or exercise intervention period in the IF and IF+HIIT groups (p<0.05), whereas CON and HIIT groups significantly increased over that period (p<0.05). At the end of the intervention period, plasma LDL levels were significantly lower in the male IF and IF+HIIT groups compared to CON (p<0.05) and HIIT (p<0.05) groups. No significant differences were observed for plasma HDL and TAG levels in males (Table 4.1). In the female mice, plasma TAG levels were significantly lower in the IF+HIIT group compared to HIIT groups at the end of the 12-weeks period (p<0.05). No significant differences between groups were observed for plasma LDL and HDL levels in females (Table 4.1).

Variables		OBC	CON	HIIT	IF	IF + HIIT
TAG (mg/dL)	Males	60.1	58.9	72.1	62.7	57.0
_		± 14.6	$\pm 10.7$	± 16.5	$\pm 10.6$	± 27.7
	Females	65.9	70.9	75.6	66.5	51.3
		± 20.5	± 13.1	± 14.8 <sup>g</sup>	± 15.7	± 24.7 <sup>Φ</sup>
HDL (mg/dL)	Males	69.8	76.2	68 7	82 1	72 1
112 L (119, uL)	i i i i i i i i i i i i i i i i i i i	± 12.2	± 11.6	± 11.3	± 32.1	± 17.3
	Fomalas	51 1	65 1	61 1	59.6	50.8
	remates	$\pm 9.0$	± 23.0	61.1 ± 9.5 *	58.6 ± 9.1 *	50.8 ± 7.7
LDL (mg/dL)	Males	40.5	82.3	74.8	40.4	29.9
		± 17.7	± 36.0 *Ψξ	± 15.6 *Ψξ	± 16.9 <sup>δΦ</sup>	± 8.9 *δΦ
	Fomalos	40.5	24.8	17.8	19 5	27.0
	i chiaics	± 17.7	± 11.8 *	± 4.8 *	± 6.0 *	± 8.9 *

Table 4.1. The effect of 12-weeks of intermittent fasting and high intensity exercise combined or individual on plasma TAG, HDL and LDL in male and female mice.

Note: Values are presented as means  $\pm$ SD. \* significant change over time,  $\delta$  significantly different from CON,  $\Phi$  significantly different from HIIT,  $\Psi$  significantly different from IF,  $\xi$  significantly different from IF+HIIT.

# 4.3.7 Effect of IF and/or HIIT on muscle strength

Absolute and relative muscle strength measured before and after 10-weeks of diet and/or exercise intervention is presented in Table 4.2. Absolute muscle strength was significantly increased in the CON group over the 10-weeks period in the male mice (p<0.05). In the female mice, absolute muscle strength was decreased in the IF group over the 10-weeks period (p<0.05). Similarly, relative muscle strength (muscle strength expressed per lean muscle mass), also declined over the 10-weeks period in IF females group only. No other significant differences were observed for absolute muscle strength or relative muscle strength (Table 4.2).

		CON		HIIT		IF	IF + HIIT		
		0 Wk	10 Wk	0 Wk	10 Wk	0 Wk	10 Wk	0 Wk	10 Wk
Absolute	Males	9.4	10.5	10.1	10.6	10.7	11.2	10.5	11.1
muscle		± 1.8	± 1.7 *	± 1.2	±1.4	± 1.6	± 0.9	±1.1	±1.3
strength									
(Score) #	Females	9.7	9.8	10.8	11.3	12.1	9.4	11.8	12.2
		± 2.7	± 2.6	± 1.9	± 2.3	± 1.9	± 1.6 *	±1.9	±1.6
Relative	Males	0.33	0.35	0.37	0.38	0.37	0.39	0.37	0.39
muscle		$\pm 0.05$	$\pm 0.05$	$\pm 0.04$	$\pm 0.05$	$\pm 0.05$	$\pm 0.04$	$\pm 0.05$	$\pm 0.05$
strength									
(Score/LM) <sup>\$</sup>	Females	0.44	0.42	0.50	0.50	0.52	0.40	0.54	0.53
		$\pm 0.12$	$\pm 0.10$	$\pm 0.10$	$\pm 0.11$	$\pm 0.11$	$\pm 0.09 *$	$\pm 0.09$	$\pm 0.08$

Table 4.2. The effect of 10-weeks of intermittent fasting and high intensity exercise combined or individual on absolute and relative muscle strength in both male and female mice.

Note: Values are presented as means ± SD. <sup>#</sup> Absolute muscle strength score represents maximum muscle strength produced independent of muscle or body size; <sup>\$</sup> Relative muscle strength represents absolute muscle strength divided by lean mass (LM); <sup>\*</sup> indicates significant changes over time.

# 4.3.8 Effect of IF and/or HIIT on organ weight

Liver weights of male mice in IF+HIIT group was significantly (p<0.05) lower than control mice (Table 4.3). No significant difference in liver weight was observed between any other group in males and between any groups in females. Pancreas weight of IF+HIIT and IF group was significantly (p<0.05) lower than control in both sexes (Table 4.3). Pancreas weight of females in HIIT group was significantly (p<0.05) lower than control. However, pancreas weights of three intervention groups were nonsignificantly different from each other in both genders. Skeletal muscle (soleus and EDL), heart, kidneys, spleen weights were non-significantly different between different groups in both sexes (data not shown).

	Live	er	Pancreas		
	Males Females		Males	Females	
CON	$1.76\pm0.49^{\xi}$	$1.55\pm0.46$	$0.13\pm0.03^{\Psi\xi}$	$0.14\pm0.04^{\Phi\Psi\xi}$	
HIIT	$1.48\pm0.40$	1.44 ±0.24	$0.12\pm0.04$	$0.10\pm0.02^{\varepsilon}$	
IF	$1.40\pm0.12$	1.44 ±0.16	$0.10\pm0.02^{\varepsilon}$	$0.11\pm0.01^{\varepsilon}$	
IF+HIIT	$1.20\pm0.14^{\varepsilon}$	1.34 ±0.11	$0.09\pm0.01^{\varepsilon}$	$0.09\pm0.02^{\varepsilon}$	

Table 4.3. The effect of 12-weeks of intermittent fasting and high intensity exercise combined or individual on liver and pancreas weight in both male and female mice.

Note: Values are presented as means  $\pm$ SD. \* significant change over time,  $\delta$  significantly different from CON,  $\Phi$  significantly different from HIIT,  $\Psi$  significantly different from IF,  $\xi$  significantly different from IF+HIIT.

#### 4.4. **DISCUSSION**

Intermittent fasting with or without high intensity interval training resulted in significant prevention in weight gain in male mice despite concurrently consuming a high fat and sugar diet. The reduction in weight was predominantly due to a significant decrease in fat mass, with no loss in lean muscle mass. At the end of the diet and/or exercise intervention period, LDL levels were significantly lower in the male IF and IF + HIIT groups compared to controls, whereas female mice demonstrated significantly lower TAG levels in the IF+HIIT group compared to HIIT group. High intensity interval training alone was also able to reduce weight and fat mass gain, albeit not significantly. Interestingly, intermittent fasting appeared to have a negative impact on markers of glycaemic control, especially within the female mice, with significant elevations in fasting blood glucose, glucose tolerance and HOMA-IR observed at the end of the intervention period. The changes in body composition, glycaemic control and lipid panels seem to be gender specific with male mice demonstrating greater changes in most variables compared to the female mice, independent of the intervention. The results from

the present study suggest that IF with or without HIIT exercise can be an effective strategy to induce weight loss despite concurrently consuming a high fat and sugar diet.

# 4.4.1 Effect of IF and/or HIIT on body composition

Following an initial 12-weeks of HF/S feeding to induce obesity, a further 10weeks of HF/S diet caused a significant increase in body mass and fat mass by ~16% and ~47%, respectively in CON male mice and by ~19% and ~46%, respectively in CON female mice. These changes are comparable to those reported elsewhere using a similar diet (Gotthardt et al., 2016, Davis et al., 2017). Combining IF for 2 days per week and HIIT for 3 days per week produced the most effective changes in body weight and fat mass compared to either IF or HIIT intervention groups individually. The IF+HIIT group weighed ~9g and ~6g less and displayed ~66% and 24% lower fat mass in the males and females, respectively, compared to the CON mice. The weight and fat loss was also greater compared to the HIIT and IF groups; although not significantly different between the male IF groups. The body compositional changes observed in the combined diet and exercise group supports the primary hypothesis of the study and provides evidence of an additive affect when implemented concurrently. Studies using combined CR with exercise have reported reduced weight gain in comparison to control animals (no diet and exercise) (Huang et al., 2010, Huffman et al., 2008). Huffman and colleagues (2008) administered 24-weeks of CR (18% restriction of high fat diet) combined with five days per week of treadmill running at speeds approximately 13-15 m/min which resulted in weight gain prevention (~15g) and fat mass reductions (~45%) compared to CR only. Similarly, weight gain prevention (~15g) was also observed following 30% CR when combined with voluntary wheel running (distance not measured) for 8-weeks (Huang et al., 2010). These changes, especially in weight, were greater than the body compositional changes observed in the current study and are likely due to a larger energy deficiency created by the greater energy restriction and/or higher exercise-induced energy expenditure of the protocols used (Huang et al., 2010, Huffman et al., 2008).

Despite the differences in total energy restriction and energy expenditure, the aforementioned studies using the combination of prolonged energy restriction and lower intensity exercise did not report weight loss, but rather prevention of weight gain following the intervention period while consuming a high fat diet (Huang et al., 2010, Huffman et al., 2008). Conversely, the current study demonstrated weight and fat mass loss in male mice at the end of the intervention period, despite concurrently consuming a high fat and sugar diet. Moreover, with no significant differences in caloric intake between the "fasting" and "non-fasting" groups, it is possible that the fasting groups with 2 days less access to food increased their food intake on non-fasting days. In light of this, it could be suggested that the changes in body weight and fat mass are due to mechanisms other than energy balance (i.e. hormonal, inflammation, genetic etc.) and requires more research to understand the additive effects of both IF and HIIT interventions at a systemic and tissue-specific cellular level.

The intermittent fasting intervention also resulted in significant weight loss (~18% or ~7.3g) and fat mass reduction (~58% or ~6.9g) compared to CON group; albeit only in male mice. Previous studies using ADF regimes have also shown improvements in body composition, with reduced gains in body mass over a 4- to 6-weeks period compared to mice consuming a standard chow diet (Varady et al., 2007, Karbowska and Kochan, 2012). Using a similar model of diet-induced obesity to that of the current study, 4-weeks of ADF led to lower body mass and fat mass gains by approximately 13% and 50%, respectively, compared to high fat fed controls (Gotthardt et al., 2016). Similarly,

Higashida and colleagues (2013) also showed significantly less weight gain (~27%) and intra-abdominal fat accumulation (~39%) following 6-weeks of ADF compared to the high fat fed controls in rats (Higashida et al., 2013). The greater difference in reduced weight gain reported in the Higashida et al. (2013) study versus the present study (~27% vs. ~16%) could be due to the frequency of fasting days, as the Higashida et al. (2013) implemented their 21 days of fasting within a 6-weeks period, whereas the current study implemented 24 days of fasting within a 10-weeks period. The discrepancy in body fat changes is most likely a reflection of the location of fat mass measured. Notwithstanding, it is evident from the current study and others that IF is an effective method of reducing body weight and fat mass in mice consuming either standard chow or a high fat and sugar diet. The results are most likely a reflection of a negative energy balance created by the reduced caloric intake on fasting days.

Another strategy to create a negative energy balance is to increase energy expenditure greater than energy intake (Keating et al., 2017). In the current study, mice completing 10-weeks of HIIT, 3 times a week resulted in reduced gains in body mass and fat mass by approximately 5% and 8% and 6% and 10% in male and female mice, respectively compared to CON mice. Greater prevention of weight and fat mass gain have been observed in other studies with up to 16% and 45% prevention in weight and fat gain, respectively (Cho et al., 2015, Lund et al., 2015). However, such differences are most likely due to the training volume, with these studies undertaking 10-12 bouts of HIIT, 5 days a week for 8-10-weeks (Cho et al., 2015, Lund et al., 2015) compared to the present study of 8 bouts, 3 days per week. Furthermore, the HIIT group in the present study concurrently consumed a diet high in fat and sugar during the intervention period and thus, any change in appetite (i.e. increase) because of the exercise training would have

resulted in increased consumption. Although the caloric intake of HIIT group could not be measured separately, it can be speculated that mice in HIIT had high caloric intake in comparison to non- fasting exercise group. Consuming a standard chow diet may have attenuated weight gain or perhaps led to weight loss.

Lean mass was lower in all intervention groups compared to the CON mice in both males and females at 10-weeks. However, this was most likely a reflection of their reduced weight gain rather than loss of lean mass, with reduced fat mass to lean mass ratio demonstrated in the IF and combination groups. The increase in fat mass to lean ratio following the HIIT intervention is contradictory to other studies (Boudou et al., 2003, Trapp et al., 2008). As mentioned previously, any increase in appetite as a result of exercise would result in greater consumption of high fat and sugar, which may lead to greater fat deposition, rather than lean mass growth. We did hypothesise that the combination group would display greater lean mass preservation compared to the IF group due to the added benefit of exercise. However, lean mass was lower in the combination group compared to the IF intervention group in males in males. Previous studies have shown improved lean mass preservation when calorie restriction is combined with treadmill exercise (Huffman et al., 2008, Weiss et al., 2015, Foster-Schubert et al., 2012), indicating the role of exercise in preserving lean mass. Although lean mass was lower in the combination group compared to the IF intervention group, muscle quality may have been greater. Indeed, relative muscle strength was higher in both male and female mice in the combination group compared to IF intervention group. This could be a result of improved muscle quality, rather than muscle mass per se as the mice were required to lift an object and hold for as long as they could, rather than hang for a period of time which is highly dependent on the weight of the mouse.

# 4.4.2 Effect of IF and/or HIIT on glucose homeostasis

Glucose intolerance and insulin resistance have been noted in animal models of diet-induced obesity (Gotthardt et al., 2016, Lund et al., 2015, Montgomery et al., 2013). In the current study, high fat feeding resulted in hyperglycaemia, insulin resistance and glucose intolerance in both male and female mice, with elevated glucose levels before the start of IF and/or HIIT interventions, indicating diabetes onset. Despite previous studies demonstrating improved glycaemic control and reduced diabetes risk following even modest weight reduction, the present study was unable to confirm such findings following the diet and/or exercise interventions. The effect of IF on glucose metabolism and glycaemic control was unexpected. Both male and female IF groups exhibited significant increases in fasting glucose and plasma insulin and thus glucose tolerance and insulin resistance at 10- or 12-weeks. Although these results seem contradictory to the commonly reported view that IF lowers fasting blood glucose levels and improves glycaemic control (Barnosky et al., 2014), it has been reported that long term (32-weeks) IF, but not CR, can lead to redox imbalances, insulin receptor nitration, and thus glucose intolerance in rats (Cerqueira et al., 2011). The intervention period in the current study was only 10- to 12-weeks, however, this period is long in comparison to the total life span of mice and equates to about 20 years of human years (Dutta and Sengupta, 2016). The adverse effects from long term IF could be attributed to the overeating pattern adopted by IF animals on non-fasting days as observed in the current study and others (Thomas et al., 2010, Cerqueira et al., 2011). This frequent feeding/fasting cycle may be a risk factor for ageassociated obesity and insulin resistance (Park et al., 2014), which may be exacerbated when consuming a high fat and sugar diet.

# 4.4.3 Effect of IF and/or HIIT on plasma lipids

Elevated TAG and LDL levels have been reported in animal models of dietinduced obesity (Montgomery et al., 2013, Li et al., 2013). Likewise, in the present study, high fat feeding resulted in hyperlipidaemia. Previous IF studies in animals have reported lower LDL (Li et al., 2013) and TAG levels (Karbowska and Kochan, 2012, Li et al., 2013) in IF mice, although only male animals were included in these studies. In the present study, males and females exhibit different response. In males, no significant difference in TAG levels between groups was observed however there was a significant lower LDL level in IF and IF+HIIT group in comparison to CON and HIIT group. The difference in LDL level among intervention groups aligns well with the changes in body weight and fat mass as mentioned above, indicating that IF and IF+HIIT are effective for maintaining lower plasma LDL in diet induced obese male mice. HIIT by itself did not have any significant effect on plasma lipids. No change in plasma TAG level with HIIT has also been reported by another study on diet induced obese mice (Cho et al., 2015, Davis et al., 2017). In females, IF+HIIT group showed significantly lower TAG level in comparison to HIIT only group, indicating the effect of combining IF and HIIT on maintaining low TAG levels in the blood as reported by other IF studies (Li et al., 2013, Karbowska and Kochan, 2012). LDL levels showed no significant difference between various groups in females and surprisingly all groups (intervention and CON) showed significantly lower LDL level in comparison to obese baseline control group. The reason behind this change is unknown and needs further investigation. Taken together, in line with the previous literature, IF and its combination with HIIT is effective in lowering some plasma lipids, if not all. However, HIIT despite being claimed to be effective in lowering plasma lipids by some human studies (Ouerghi et al., 2017, Fisher et al., 2015), did not show any beneficial effect on plasma lipids in diet induced obese mice. As mentioned earlier, the intensity and frequency of HIIT employed in this study could be the reason behind no apparent benefit.

# 4.4.4 Effect of IF and/or HIIT on muscle strength

Obesity has been reported to affect muscle strength in high fat diet induced obese mice (Tam et al., 2015). However, in the present study, diet induced control group showed increase in absolute muscle strength, although when expressed as relative muscle strength, it did not show any change over 10-week intervention period. We assumed that HIIT will improve muscle strength, however muscle strength was unaffected by HIIT and/or IF in males. In females, both relative and absolute muscle strength was reduced in the IF group, which again was most likely a reflection of the decrease in muscle mass observed in that group. No other significant changes in muscle strength was noted in other groups. An observation of interest was that the females in IF+HIIT group demonstrated similar strength values to the HIIT and control group, which indicates that decline in muscle strength observed following the IF regime in the female mice was counteracted by exercise.

## 4.4.5 Effect of IF and/or HIIT on organ weight

Increase in adipose tissue mass during weight gain is often associated with ectopic accumulation of fats in other tissues like heart, muscle, liver and pancreas (O'Brien et al., 2017, Stern et al., 2016, Ruberg et al., 2010). Accumulation of ectopic fats in pancreas and liver leads to conditions like non-alcoholic fatty pancreas disease and non-alcoholic fatty liver disease respectively (Tariq et al., 2016). Lifestyle interventions like HIIT (Cho et al., 2015), IF (Baumeier et al., 2015, Antoni et al., 2017) and calorie restriction (Larson-Meyer et al., 2006) have been reported to prevent ectopic lipid accumulation. However,

in the present study, neither IF nor HIIT alone showed any significant difference on liver weight in comparison to *ad libitum* control group. Liver weights of males in IF+HIIT was lower in comparison to control males indicating the beneficial effect of this combined intervention in preventing lipid accumulation in liver in diet induced obese male mice. Unlike liver, IF with and without exercise showed significantly lower pancreas weight in comparison to CON group in both sexes. In females, even the HIIT only group showed significantly lower pancreas weight. These observations indicate that despite high fat/sugar diet, both male and female mice in IF with and without HIIT groups and females on HIIT only group maintained lower pancreas weight, demonstrating the effectiveness these interventions in preventing ectopic fat accumulation in pancreas.

#### 4.4.6 <u>Sexually dimorphic response to IF and/or HIIT</u>

An observation of interest was the gender specific responses to the same diet and/or exercise intervention. Changes in body weight and composition, markers of glucose homeostasis, lipid profiles, muscle strength and ectopic lipid accumulation were clearly different between males and females, with males demonstrating greater improvements in most, if not all markers. Previous lifestyle intervention studies and recent systematic reviews also confirm that males tend to lose more weight than females in response to lifestyle interventions (Williams et al., 2015, Baillot et al., 2015, Gallagher et al., 2017), but also acknowledge than females also lost weight. The exact mechanisms behind these reported gender differences have yet to be fully explored in detail. Male and female mice exhibit differences in adipose tissue distribution and functionality (Valencak et al., 2017), with obese female mice having higher fat mass, which was also noted in the present study and a tendency to mobilize fats more slowly in comparison to weightmatched males (Nickelson et al., 2012, Mauvais-Jarvis, 2015). Furthermore, females tend to store free fatty acids as triglycerides whereas males tend to oxidise free fatty acid under similar resting conditions (Uranga et al., 2005). Finally, hormonal differences between sexes may also influence the responses to diet and exercise with males exhibiting higher levels of circulating leptin and females higher levels of circulating adiponectin (Valencak et al., 2017). In addition, elevated estrogen levels following IF, have been linked to the development of insulin resistance in females (Kumar and Kaur, 2013) (Valencak et al., 2017). Clearly more research is needed in this area to understand the mechanisms behind these gender specific responses but also help guide diet and exercise recommendations in the future.

# 4.5. CONCLUSION

Current evidence suggests that as little as 10 minutes of high intensity exercise can improve metabolic health and aerobic capacity (Gillen et al., 2016) and alternative day fasting can reduce obesity-associated changes in body composition, fasting insulin and glucose concentrations (Varady et al., 2009, Varady et al., 2013). The current study wanted to confirm the aforementioned findings by mimicking similar protocols in animal models. Secondly, to investigate the combined effects of such regimes and its long-term impact, and thirdly, unlike human studies which typically don't change the research participant's diet, we wanted to observe the effects of these diet and exercise regimes while concurrently consuming a high fat and sugar diet. The present study is the first study to demonstrate superior effects on body composition and lipid profiles following combined IF and HIIT compared to either diet or exercise intervention alone while concurrently consuming a high fat and sugar diet. These observations are likely due to changes in energy balance, with a greater negative energy balance shift compared to the other groups. However, given no significant differences between 'fasting' and 'nonfasting' intervention groups and observations from previously published CR and exercise studies, it could also be suggested that our observations may be due to more than just energy balance differences. An observation of interest was the gender specific responses to the same diet and/or exercise intervention and could indicate potential hormonal differences influencing metabolic control/adaptation in mice. The results of the present study suggest that the combination of diet and exercise are most effective at attenuating the negative impact that high fat and sugar diet has on body weight, composition and lipid levels.

# **CHAPTER 5**

# MOLECULAR BASIS OF REDUCED FAT MASS AND ASSOCIATED BENEFITS IN ADIPOSE TISSUE INDUCED BY INTERMITTENT FASTING WITH OR WITHOUT HIGH INTENSITY INTERVAL TRAINING IN DIET-INDUCED OBESE MICE

# 5.1. INTRODUCTION

Chapter 4 investigated the effects of IF and HIIT, alone and in combination, on anthropometric and metabolic health parameters in diet-induced obese mice. The main findings of chapter 4 was that intermittent fasting with or without high intensity interval training resulted in significantly less weight gain, predominantly due to less adipose tissue accumulation, in male mice despite concurrently consuming a high fat and sugar diet for 12 weeks. Chapter 5 will investigate diet and/or exercise induced changes in adipose tissue by examining changes in microRNA and mRNA expression related to adipose tissue function, hypoxia, and inflammation.

Adipose tissue plays a central role in lipid metabolism and energy homoeostasis (Valencak et al., 2017). The combination of high energy intake and low energy expenditure results in the storage of excess calories in the form of adipose tissue, leading to adipose tissue expansion and disturbance in the systemic energy homeostasis (Rutkowski et al., 2015). Adipose tissue expansion also disrupts the normal functioning of adipocytes, leading to dysfunction in the lipolytic and secretory functions, increased inflammation, insulin resistance (Greenberg and Obin 2006, Guilherme, Virbasius et al. 2008), and adipose tissue hypoxia (van Beek, van Klinken et al. 2015). These repercussions are underpinned by changes in various signalling molecules and expression of genes involved in fine tuning the structural and functional integrity of adipose tissue

(Manteiga et al., 2013). For instance, high fat intake downregulates gene expression related to fatty acid uptake (Fatty acid binding protein 4), oxidation (Hydroxyacyl-Coenzyme-A dehydrogenase), insulin signalling (insulin receptor substrate 2) and glucose uptake (GLUT3) in mice (Choi et al., 2015). These effects may partially be responsible for the decrease in energy expenditure, fat accumulation and impaired insulin sensitivity typically observed during high fat feeding (Choi et al., 2015). High fat feeding also upregulates pro-inflammatory genes like TNF $\alpha$  (van der Heijden et al., 2015) and hypoxia related gene like hypoxia-inducible factor 1-alpha (He et al., 2011); which contribute to the pro-inflammatory state that is associated with an obesity phenotype.

Intermittent fasting alone is effective in reducing body weight, in the form of adipose tissue mass, while having minimal effects on muscle mass. Researchers have begun to identify potential drivers of such physiological changes at cell and molecular level following intermittent fasting. These include enhance expression of adipose tissue genes related to: adipogenesis such as cell death-inducing DFFA-like effector C (*CIDEC*) and CIDEA (Kim et al., 2017), peroxisome proliferator activated receptor gamma (Karbowska and Kochan, 2012), markers of white adipose tissue to brown adipose tissue conversion such as *PGC1a*, and uncoupling protein 1 (Kim et al., 2017, Li et al., 2017a) following intermittent fasting.

Similar to intermittent fasting, high intensity interval training (HIIT) alone can also enhance body composition and improve metabolic health. However, unlike intermittent fasting which reduces calorie intake, HIIT increases energy expenditure. Notwithstanding, it is clear that manipulations of energy intake and expenditure that can push the energy balance towards negative side is the key driver of these effects. Under high fat fed conditions, previous studies have demonstrated HIIT's ability to improve glucose handling as evidenced by increased 2-Deoxy-D-glucose uptake within adipose tissue, despite no change in expression of inflammatory genes like *IL6*, *TNFa* or the neutrophil chemoattractant *Kc* in mice (Marcinko et al., 2015). In addition, HIIT has shown to enhance adipose tissue angiogenesis by increasing expression of vascular endothelial growth factor (VEGF) gene expression (Ghafari Homadini et al., 2017), which is otherwise reduced in obesity (Disanzo and You, 2014).

Recent addition to the complexity of the molecular signals regulating adipose tissue and systemic homeostasis is miRNAs. Several studies have reported that miRNAs play a crucial role in maintenance of structural and functional integrity of adipose tissue under both normal and dysfunctional states (Wilson et al., 2017, Deiuliis, 2016, Price and Fernández-Hernando, 2016). For instance, miR-27a has been found to be critical in maintaining adipose tissue mass and its downregulation can lead to adipose tissue dysfunction (Kim et al., 2010). Limited research has examined changes in adipose tissue miRNA expression following either diet and/or exercise intervention. Recently, upregulation of miR-149-3p has been found to promote fasting induced switching of subcutaneous to visceral adipose tissue (Ding et al., 2016), while enhanced expression of miR-10b promotes endurance exercise training induced  $\beta$ -adrenergic lipolysis in adipose tissue (Tsiloulis et al., 2016).

The objective of the study was to evaluate the effect of IF with and without HIIT on adipose tissue expression of genes involved in adipogenesis, lipogenesis, lipolysis, lipid handling, oxidation, hypoxia, inflammation and beiging in diet-induced obese mice. Expression of microRNAs related to the genes targeted in the present study were also investigated to further elucidate their regulatory involvement. MiRNAs were selected on the basis of their relationship to the investigated genes, either reported in the literature and/or found through miRWALK software (Dweep et al., 2011). As the combination of IF+HIIT was most effective in reducing fat mass, it was hypothesized that combination of IF with HIIT will show additive or synergistic effect in altering the expression of genes and miRNAs in adipose tissue.

# 5.2. MATERIALS AND METHODS

# 5.2.1 Animals

Details of animal procurement, housing and diet are mentioned in chapter 3 section 3.2. Briefly, this study included four different groups namely: no intervention (CON; Males=7, Females=9); intermittent fasting (IF; Males=8, Females=10), high intensity intermittent exercise (HIIT; Males=8, Females=10) and a combination of the dietary and exercise intervention (IF+HIIT; Males=8, Females=10). All groups were maintained on high fat/high sugar (HF/S) diet for 12 weeks to induce obesity and then underwent respective interventions for another 12 weeks while still consuming HF/S diet. Details of interventions is given in chapter 3 section 3.4. After 24 weeks of HF/S feeding and intervention, all mice were killed and tested as described below.

#### 5.2.2 Culling and tissue collection

Mice were anaesthetised with intraperitoneal injection of pentobarbitone (60 mg/kg) and killed by cardiac puncture. The abdominal cavity was opened and epididymal and periovarian white adipose tissue (WAT) was collected, snap frozen in liquid nitrogen and stored in -80°C freezer until further analysis.

# 5.2.3 Adipose tissue gene and miRNA expression

Total RNA was extracted form adipose tissue, reverse transcribed to generate cDNA and subjected to RT-PCR for gene and miRNA expression. Procedures for mRNA and miRNA isolation, cDNA synthesis, RT-PCR analysis and calculation of fold change

are presented in detail in chapter 3, section 3.7. Beta-2 microglobulin (B2M) and RNU6 were used to normalize mRNA and miRNA expression data, respectively. The primers for genes and miRNAs were purchased from and designed by QIAGEN (Chadstone, Victoria, Australia). The list of genes and miRNAs targeted in this study are given in table 5.1 and table 5.2 respectively.

Gene Qiagen **NCBI Reference** symbol **Official full name** Catalogue sequence number number Peroxisome proliferator activated **PPAR**<sub>y</sub> NM\_001127330 PPM05108C receptor gamma Cell death-inducing DFFA-like CIDEC NM\_178373 PPM25558B effector c Sirtuin 1 NM 001159589 SIRT1 PPM05054A LIPE Lipase, hormone sensitive NM 001039507 PPM03313A FOX01 Forkhead box O 1 NM\_019739 PPM03381C Fatty acid binding protein 4, FABP4 NM\_024406 PPM04517A adipocyte HIF1A Hypoxia inducible factor 1, alpha NM 010431 PPM03799C Hydroxyacyl-Coenzyme A HADH NM\_008212 PPM35732B dehydrogenase Transient receptor potential cation TRPV4 NM\_022017 PPM36070A channel, subfamily V, member 4 LEP Leptin NM 008493 PPM03504B Uncoupling protein 1 UCP1 NM\_009463 PPM05164B B2M Beta-2 microglobulin NM\_009735 PPM03562A

Table 5.1. List of targeted genes in adipose tissue with full official names, NCBI reference sequence numbers and Qiagen catalogue number.

Table 5.2. List of miRNAs analysed in adipose tissue, miRbase accession numbers and Qiagen miScript Primer Assay Catalog Numbers.

Mature miRNA ID	miRBase Accession No.	Qiagen miScript Primer Assay Catalog No.
mmu-miR-27a-3p	MIMAT0000537	MS00001351
mmu-miR-24-3p	MIMAT0000219	MS00005922
mmu-miR-222-3p	MIMAT0000670	MS00007959
mmu-miR-143-3p	MIMAT0000247	MS00001617
mmu-miR-145a-5p	MIMAT0000157	MS00001631

# 5.2.4 <u>Statistical analysis</u>

Data is presented as mean  $\pm$  standard deviation. To evaluate the effect of interventions on mRNA expression, intervention groups namely HIIT, IF, and IF+HIIT were compared against no intervention CON group. Difference in mRNA expression and miRNA expression between groups were analysed using one-way ANOVA and Tukey's test was used as post hoc analysis. Correlation between mRNA and miRNA expression in each intervention group was calculated with Pearson correlation test for *PPAR* $\gamma$  Vs miR-27a, *FABP4* Vs miR-24, *HIFa* Vs miR-222, Lep Vs miR-143 and *FOXO1* Vs miR-145. An alpha level of 0.05 was adopted throughout to reduce Type I statistical errors.

# 5.3. **RESULTS**

# 5.3.1 <u>The effect of 12 weeks of IF and/or HIIT on WAT gene expression related</u> to structure and function

# 5.3.1.1. CIDEC

White adipose tissue *CIDEC* gene expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 5.1 (A) and (B) respectively. Gene expression of *CIDEC* was significantly lower in the IF group compared to the HIIT group in both males (p<0.01) and females (p<0.01). In female mice, the *CIDEC* expression was significantly lower in the IF group compared to the HIIT group (p<0.01). The combined exercise and IF group also showed significantly lower *CIDEC* expression, however, only compared to HIIT group (p<0.05).



Figure 5.1. Effect of IF, HIIT and IF+HIIT on *CIDEC* gene expression (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *B2M* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*) and (\*\*) brackets represent significant difference between groups at p<0.05 and p<0.01 level of significance respectively.

# 5.3.1.2. Leptin

White adipose tissue leptin gene expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 5.2 (A) and (B) respectively. In both males and females, gene expression of leptin in CON group was significantly higher in comparison to IF (p<0.01) and IF+HIIT (p<0.01) groups. In males only, gene expression of leptin was significantly higher in the HIIT group compared to the IF (p<0.05) and IF+HIIT (p<0.01) groups.



Figure 5.2. Effect of IF, HIIT and IF+HIIT on *LEP* gene expression (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *B2M* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*) and (\*\*) brackets represent significant difference between groups at p<0.05 and p<0.01 level of significance respectively.

# 5.3.1.3. *PPARγ*

White adipose tissue *PPARy* gene expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 5.3 (A) and (B) respectively. In males, IF+HIIT group showed significantly higher expression of *PPARy* in comparison to CON group (p<0.05). In females, no significant differences were observed between control and IF and/or HIIT groups at the end of 12-week intervention period.



Figure 5.3. Effect of IF, HIIT and IF+HIIT on *PPAR* $\gamma$  gene expression (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *B2M* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*) bracket represent significant difference between groups at p<0.05 level of significance.

# 5.3.1.4. FOXO1

White adipose tissue *FOXO1* gene expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 5.4 (A) and (B) respectively. In females, both IF and IF+HIIT group showed significantly lower expression of *FOXO1* relative to both CON (p<0.01) and HIIT (p<0.01) groups. In males, no significant differences in *FOXO1* gene expression between control and IF and/or HIIT groups were noted at the end of 12-week intervention period.



Figure 5.4. Effect of IF, HIIT and IF+HIIT on *FOXO1* gene expression (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *B2M* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*\*) brackets represent significant difference between groups at p<0.01 level of significance.

# 5.3.2 <u>The effect of 12 weeks of IF and/or HIIT on WAT gene expression related</u> to Fatty acid oxidation and trafficking

5.3.2.1. HADH

White adipose tissue *HADH* gene expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 5.5 (A) and (B) respectively. Gene expression of *HADH* in IF group was significantly lower in comparison to IF+HIIT group in both males (p<0.05) and females (p<0.01).



Figure 5.5. Effect of IF, HIIT and IF+HIIT on *HADH* gene expression (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *B2M* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*) and (\*\*) brackets represent significant difference between groups at p<0.05 and p<0.01 level of significance respectively.

#### 5.3.2.2. FABP4

White adipose tissue *FABP4* gene expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 5.6 (A) and (B) respectively. Gene expression of *FABP4* in IF+HIIT group was significantly higher in comparison to CON (p<0.01), HIIT (p<0.01) and IF (p<0.01) groups in males only. In females, IF group showed significantly (p<0.01) lower expression of *FABP4* compared to HIIT group at the end of 12-week intervention period.



Figure 5.6. Effect of IF, HIIT and IF+HIIT on *FABP4* gene expression (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *B2M* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*\*) brackets represent significant difference between groups at p<0.01 level of significance.

# 5.3.2.3. Hormone sensitive lipase

White adipose tissue hormone sensitive lipase (HSL) gene expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 5.7 (A) and (B) respectively. At the end of 12-week intervention period, no significant differences in HSL gene expression were observed between control and IF and/or HIIT groups in both males and females.



Figure 5.7. Effect of IF, HIIT and IF+HIIT on *HSL* gene expression (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *B2M* gene expression. Bars represent means, error bars represent standard deviation.

# 5.3.3 <u>The effect of 12 weeks of IF and/or HIIT on WAT gene expression of</u> <u>SIRT1</u>

White adipose tissue *SIRT1* gene expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 5.8 (A) and (B) respectively. In males, IF group showed significantly lower expression of *SIRT1* in comparison to IF+HIIT group (p<0.01). In females, both IF and IF+HIIT showed significantly lower expression of *SIRT1* compared to both CON (p<0.01) and HIIT (p<0.01) groups.



Figure 5.8. Effect of IF, HIIT and IF+HIIT on *SIRT1* gene expression (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *B2M* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*\*) brackets represent significant difference between groups at p<0.01 level of significance.

# 5.3.4 <u>The effect of 12 weeks of IF and/or HIIT on WAT gene expression related</u> to inflammation

# 5.3.4.1. TRPV4

White adipose tissue *TRPV4* gene expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 5.9 (A) and (B) respectively. Gene expression of *TRPV4* in the IF group was significantly lower in comparison to CON group in both males (p<0.05) and females (p<0.01) and the HIIT group in females only (p<0.01). In females, IF+HIIT group also showed significantly lower expression of *TRPV4* gene in comparison to CON group (p<0.01).



Figure 5.9. Effect of IF, HIIT and IF+HIIT on *TRPV4* gene expression (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *B2M* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*\*) brackets represent significant difference between groups at p<0.01 level of significance.
# 5.3.5 <u>The effect of 12 weeks of IF and/or HIIT on WAT gene expression related</u> to hypoxia

White adipose tissue *HIF1a* gene expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 5.10 (A) and (B) respectively. In males, IF group showed significantly lower *HIF1a* gene expression relative to CON group (p<0.01). In females, no significant differences in *HIF1a* gene expression were noted between control and IF and/or HIIT groups at the end of 12-week intervention period. The effect of 12 weeks of IF and/or HIIT on WAT gene expression related to white adipose tissue beiging.



Figure 5.10. Effect of IF, HIIT and IF+HIIT on  $HIF1\alpha$  gene expression (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to B2M gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*\*) bracket represent significant difference between groups at p<0.01 of significance.

<sup>5.3.5.1.</sup> *HIF1α* 

## 5.3.5.2. UCP1

White adipose tissue *UCP1* gene expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 5.11 (A) and (B) respectively. At the end of 12-week intervention period, no significant differences in *UCP1* gene expression were observed between CON and IF and/or HIIT groups in both males and females.



Figure 5.11. Effect of IF, HIIT and IF+HIIT on *UCP1* gene expression (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *B2M* gene expression. Bars represent means, error bars represent standard deviation.

# 5.3.6 <u>The effect of 12 weeks of IF and/or HIIT on miRNA expression relating</u> to adipogenesis and glucose handling in adipose tissue

5.3.6.1. miR-27a

White adipose tissue miR-27a expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 5.12 (A) and (B) respectively. At the end of 12-week intervention period, no significant differences in miR-27a expression were observed between CON and IF and/or HIIT groups in both males and females.



Figure 5.12. Effect of IF, HIIT and IF+HIIT on miR-27a expression (fold change) relative to CON group in (A) males and (B) females. MiRNA expression is normalized to *RNU6* expression. Bars represent means, error bars represent standard deviation.

# 5.3.6.2. miR-24

White adipose tissue miR-24 expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 5.13 (A) and (B) respectively. In males, expression of miR-24 was significantly higher in the HIIT group compared to the CON (p<0.01), IF (p<0.01) and IF+HIIT (p<0.01) groups. In females, no significant difference in miR-24 expression were noted between CON and IF and/or HIIT groups at the end of 12-week intervention period.



Figure 5.13. Effect of IF, HIIT and IF+HIIT on miR-24 expression (fold change) relative to CON group in (A) males and (B) females. MiRNA expression is normalized to *RNU6* expression. Bars represent means, error bars represent standard deviation. Asterisk (\*\*) brackets represent significant difference between groups at p<0.01 level of significance.

#### 5.3.6.3. miR-222

White adipose tissue miR-222 expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 5.14 (A) and (B) respectively. In males, expression of miR-222 was significantly higher in the HIIT group compared to the IF+HIIT group (p<0.01). In females, no significant difference in miR-222 expression were noted between CON and IF and/or HIIT groups at the end of 12-week intervention period.



Figure 5.14. Effect of IF, HIIT and IF+HIIT on miR-222 expression (fold change) relative to CON group in (A) males and (B) females. MiRNA expression is normalized to *RNU6* expression. Bars represent means, error bars represent standard deviation. Asterisk (\*\*) bracket represent significant difference between groups at p<0.01 level of significance.

# 5.3.6.4. miR-143

White adipose tissue miR-143 expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 5.15 (A) and (B) respectively. In males, expression of miR-143 was significantly higher in the HIIT group compared to the CON (p<0.01) and IF+HIIT (p<0.01) groups. In females, no significant differences in miR-143 expression were noted between CON and IF and/or HIIT groups at the end of 12-week intervention period.



Figure 5.15. Effect of IF, HIIT and IF+HIIT on miR-143 expression (fold change) relative to CON group in (A) males and (B) females. MiRNA expression is normalized to *RNU6* expression. Bars represent means, error bars represent standard deviation. Asterisk (\*\*) brackets represent significant difference between groups at p<0.01 level of significance.

# 5.3.6.5. miR-145

White adipose tissue miR-145 expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 5.16 (A) and (B) respectively. In males, HIIT group showed higher expression in comparison to CON (p<0.01) and IF+HIIT (p<0.01) groups. The IF group showed significantly higher expression compared to CON group (p<0.05). In females, expression of miR-145 in CON group was significantly lower than HIIT (p<0.01) and IF+HIIT (p<0.01) group.



Figure 5.16. Effect of IF, HIIT and IF+HIIT on miR-145 expression (fold change) relative to CON group in (A) males and (B) females. MiRNA expression is normalized to *RNU6* expression. Bars represent means, error bars represent standard deviation. Asterisk (\*) and (\*\*) brackets represent significant difference between groups at p<0.05 and p<0.01 level of significance respectively.

#### 5.3.7 Correlation analysis between mRNA and miRNA expression

Pearson correlation coefficient (R) and significance values (p) between gene and miRNA expression in males and females are given in table 5.3 and 5.4 respectively. Significant inverse correlation was found between *HIF1a* gene and miR-222a expression in males HIIT group (R=-0.68, p<0.05) (Figure 5.17 (A)) and between *FOXO1* gene and miR-145 expression in males IF+HIIT group (R=-0.81, p<0.016) (Figure 5.17 (B)).

Table 5.3. Correlation between mRNA and miRNA expression in adipose tissue in males.

	PPARy Vs	FABP4 Vs	HIF1α Vs	LEP Vs	FOXO1 Vs
	miR-27	<b>miR-24</b>	miR-222a	miR-143	miR-145
CON	0.91 (0.01)	0.68 (0.07)	-0.32 (0.27)	0.54 (0.13)	0.76 (0.04)
HIIT	0.70 (0.03)	0.29 (0.26)	<b>-0.68</b> (0.05)	-0.52 (0.11)	0.55 (0.10)
IF	0.52 (0.09)	0.15 (0.36)	-0.18 (0.34)	-0.26 (0.29)	0.26 (0.27)
IF+HIIT	-0.05 (0.45)	-0.49 (0.13)	-0.45 (0.13)	0.14 (0.37)	<b>-0.81</b> (0.01)

Note-Values represent Pearson correlation coefficient (R) and values in brackets represent p values.

Table 5.4. Correlation between mRNA and miRNA expression in adipose tissue in females.

	PPARy Vs	FABP4 Vs	HIF1α Vs	Lep Vs	FOXO1 Vs
	miR-27	miR-24	miR-222	miR-143	miR-145
CON	0.33 (0.21)	0.27 (0.26)	0.19 (0.32)	-0.21 (0.31)	-0.45 (0.13)
HIIT	0.72 (0.01)	0.51 (0.05)	0.29 (0.21)	-0.54 (0.07)	0.49 (0.08)
IF	-0.40 (0.16)	0.28 (0.25)	-0.49 (0.11)	0.39 (0.17)	0.45 (0.15)
IF+HIIT	0.19 (0.31)	-0.11 (0.39)	-0.17 (0.33)	-0.39 (0.15)	0.17 (0.33)

Note-Values represent Pearson correlation coefficient (R) and values in brackets represent p values.



Figure 5.17. Correlation between (A) HIF1 $\alpha$  and miR-222a in HIIT group (R=-0.68, p<0.05) and (B) FOXO1 and miR-145a in males from IF+HIIT group (R=-0.81, p<0.016) in males.

# 5.4. **DISCUSSION**

The major finding of chapter 5 is that IF with or without HIIT demonstrated greater changes in expression of WAT genes relating to fragmentation of unilocular lipid droplets, lipolysis, fatty acid oxidation and efflux compared to either HIIT and/or CON group. Increased expression of such genes within WAT supports the enhanced fat loss observed in chapter 4. Lower expression levels of leptin, pro-inflammatory markers such as TRPV4 and markers of hypoxia were also observed in the combined group providing further support to the anthropometric findings in chapter 4. Interestingly, the insulin resistant phenotype observed in the IF groups for both male and female mice also displayed higher levels of SIRT1 gene expression within WAT which have recently been reported to be related to systemic insulin resistance (Stefanowicz et al., 2018). Similar to Chapter 4 findings, the changes at the gene level were also gender specific.

#### 5.4.1 Adipose tissue structure and function

White adipose tissue functions as an energy reservoir where excess circulating fatty acids are transported to WAT, converted to triglycerides, and stored as unilocular lipid droplets. Accumulation of fats as unilocular lipid droplets is said to be regulated by CIDEC, also known as fat specific protein 27(Fsp27) (Karbowska and Kochan, 2012, Duncan et al., 2007). In the present study, IF resulted in significantly lower *CIDEC* expression compared to the HIIT and CON group in females and only compared to the HIIT group in males. Expression of *CIDEC* was also lower in the combined IF and HIIT group compared to CON, albeit not significantly. Lower expression of *CIDEC* indicates fragmentation of unilocular lipid droplets into smaller lipid droplets, by promoting phosphorylation of perilipin1 and translocation of adipose triglyceride lipase (ATGL) and hormone sensitive lipase (HSL) (Tsiloulis and Watt, 2015), which results in enhanced

lipolysis (Marcinkiewicz et al., 2006). Lower expression of CIDEC following calorie restriction has been reported by previous studies in humans after 12 weeks (Magnusson et al., 2008) and mice after 4 weeks of high fat diet (Wang et al., 2016). Contrary to this, IF (3 days feeding: 3 days fasting) for ~48 days showed increased expression of CIDEC (Karbowska and Kochan, 2012), suggesting that IF might be stimulating lipid deposition in adipose tissue. The Karbowska and Kochan (2012) study used lean rats and thus the effects of IF on CIDEC expression could be opposing, as the majority of studies demonstrating lower expression following reduced caloric intake are on models of dietinduced obesity. Indeed protein expression has been reported to be lower, while gene expression higher following high fat diet (Reynolds et al., 2015). In line with lower levels of CIDEC expression, leptin expression, which is increased with weight gain, was significantly lower in IF and IF +HIIT group compared to the CON group in both males and females. Lower expression of leptin could be reflection of the reduced fat mass accumulation observed in the diet and exercise intervention groups. It could also be driven by *PPAR* $\gamma$ . Leptin expression is negatively regulated by *PPAR* $\gamma$ , with higher expression of PPARy reducing leptin expression (Mousavi et al., 2015, Kersten, 2000). In the present, higher expression levels of *PPAR* were observed in all diet and/or exercise intervention groups in males only, though only statistically significant in the IF +HIIT group. Furthermore, a significant negative correlation between expression of  $PPAR\gamma$  and the leptin gene was observed (R=-0.68, p<0.05) in males. Increased PPARy expression has been linked to amelioration of obesity-related metabolic dysfunction and decreased markers of inflammation within white adipose tissue (Yang et al., 2013). Conversely, *PPAR* $\gamma$  has also been considered a thrifty gene and can mediate high fat diet induced adipocyte hypertrophy (Hara et al., 2000, Auwerx et al., 2003). Though protein expression of *PPAR* $\gamma$  was not measured, based on the improved body composition observed in chapter 4 and other studies reporting enhanced expression with IF and exercise (Karbowska and Kochan, 2012, Stanford et al., 2015), we can speculate that increased expression of *PPAR* $\gamma$  is having a positive benefit within the adipose tissue, by reducing inflammatory induced dysfunction or affecting the expression of other functional genes.

*FOXO1* is a regulator of cellular metabolism and survival in response to nutrient availability and environmental stress by acting as a direct target of insulin action (Kim et al., 2009). Exposure of preadipocytes to insulin along with other adipogenic hormones simultaneously activates cell proliferation and differentiation. It is suggested that role of FOXO1 role in preadipocytes is to help with transition from clonal expansion (i.e. cell cycle) to terminal differentiation, possibly by cooperating with PPAR $\gamma$  to induce a postmitotic growth arrest that is a prerequisite for the subsequent differentiation (Morrison and Farmer, 1999). In the current study, both IF and IF+HIIT groups in the female mice demonstrated significantly lower expression of FOXO1 compared to the CON group. Although this trend was similar in males, it was not deemed statistically significantly. Indeed, expression of *FOXO1* was found to be correlated (R=-0.81, p<0.01) with miR-145 in the male IF+HIIT group. MiR-145 has been shown to target *FOXO1* in adipose tissue and its enhanced expression attenuates lipolysis (Lin et al., 2014). Reduced FOXO1 expression would suggest a decrease in transition from clonal expansion (i.e. cell cycle) to terminal differentiation in preadipocytes.

# 5.4.2 Fatty acid oxidation and trafficking

High fat diet disrupts molecular machinery involved in triacylglycerol (TAG) breakdown and fatty acids release from adipose tissue (Gaidhu et al., 2010). Elevation in

both oxidation and mobilization of fats is necessary for normal functioning of adipose tissue (Gaidhu et al., 2010). In the present study, IF + HIIT in both males and females showed higher HADH gene expression relative to IF group, suggesting a higher  $\beta$ oxidation capacity, as HADH is an important enzyme that catalyses the penultimate and rate limiting step in β-oxidation (van Hove et al., 2006, Verhoef et al., 2013). Similarly, IF+HIIT group, but only in male mice, demonstrated an increased expression of FABP4 compared to all other groups. FABP4 is responsible for intracellular fatty acid trafficking and is an indicator of lipid handling capacity of adipocytes (Verhoef et al., 2013). Superior weight loss in the IF+HIIT group compared to either diet or exercise intervention alone could be a reflection of higher lipid oxidation and efflux. This was further supported by positive correlation found between HADH and FABP4 gene expression in IF+HIIT groups in both males (R=0.66, P<0.05) and females (R=0.87, P<0.001) in the present study. Similar correlation between HADH and FABP4 was also reported by another study in humans after 8-weeks of calorie restriction induced weight loss, which reported improved metabolic homeostasis, higher trafficking and  $\beta$ -oxidation of fatty acids (Camps et al., 2015). Further, enhanced oxidation and trafficking has been shown after caloric restriction for 5 weeks (500Kcal/day) with increased expression of HADH and FABP4 in adipose tissue of obese men and women (Bouwman et al., 2014). In contract, IF alone in the present study, rather showed lower HADH and FABP4 gene expression. Low expression indicates low fatty acid mobilisation and efflux from adipose tissue (Hertzel et al., 2002, Furuhashi et al., 2014). In females, HIIT only group also showed higher expression of FABP4 relative to IF only group. This was in line with the previous study where aerobic exercise training for 10 weeks has been reported to increase the FABP4 gene expression in adipose tissue of obese Zucker rats (Krskova et al., 2012).

# 5.4.3 Adipose tissue SIRT1 expression and systemic insulin sensitivity

Obesity is often associated with insulin resistance and type 2 diabetes (Wu and Ballantyne, 2017). Though, IF has been reported to improve insulin sensitivity (Barnosky et al., 2014), IF groups in both males and females and IF+HIIT group in females showed poor insulin sensitivity, even worse than CON group. A recent study has reported that insulin sensitivity in young obese men decreased with obesity and was positively related to *SIRT1* expression in adipose tissue but not muscle (Stefanowicz et al., 2018). *SIRT1*, an NAD+-dependent deacetylase enzyme, is an important nutrient sensor in various metabolic tissues (Chalkiadaki and Guarente, 2012). In the present study, expression of *SIRT1* was found to be suppressed in IF groups in both males and females and IF+HIIT group in females. This pattern was very similar to the insulin resistant state exhibited by these groups. Although the exact explanation behind deterioration of insulin sensitivity in IF groups needs extensive investigation, *SIRT1* suppression in adipose tissue is a contributing factor towards the insulin resistant states of these groups as hypothesised by Stefanowicz et al. (2018).

Treadmill exercise training (30m/min, 45 min/day, 5 days/week) for 6-weeks has been reported to enhance *SIRT1* activity in adipose tissue of rats (Ferrara et al., 2008). In the present study, females in HIIT group showed higher *SIRT1* expression in comparison to both IF and IF+HIIT group. Also in males, combining HIIT with IF resulted in higher expression of *SIRT1*, in comparison to IF only, suggesting that HIIT rescued *SIRT1* expression, which was otherwise downregulated by IF in males. These groups (HIIT in females and IF+HIIT in males) also showed lower insulin resistance in comparison to respective IF only group. Also, it has been shown that *SIRT1* induces FOXO1 deacetylation which promotes nuclear retention of FOXO1 leading to maintenance of FOXO1 induced signalling pathway and decreased adipose tissue mass (Li et al., 2017b). This is in accordance with the present study where enhanced expression of *SIRT1* gene in males from IF+HIIT group was accompanied by lower fat mass, which could be attributed to *SIRT1* induced deacetylation of FOXO1 suggested by Li et al. (2017b).

#### 5.4.4 Inflammation

Aerobic exercise training for 60min/day, five time/week has been reported to decrease the expression of pro-inflammatory marker *TRPV4* gene in adipose tissue of chow fed mice with lower epididymal fat mass than sedentary group (Castellani et al., 2014) and high fat diet fed obese mice with similar epididymal fat mass as sedentary group (Chen et al., 2015). *TRPV4* is a non-specific calcium ion channel that is highly expressed in adipose tissue (Chen et al., 2015) and is an important regulator of pro-inflammatory cytokines (Ye et al., 2012). In contrast to the previous reports (Castellani et al., 2014, Chen et al., 2015), HIIT group in both males and females in the present study did not show any significant changes in *TRPV4* gene expression, which could be attributed to lower exercise volume and this stimulus used in the present study. Similar changes in *TPV4* was significantly lower in the IF group compared to CON. In females, *TRPV4* was significantly lower in the IF group compared to CON and HIIT. The IF+HIIT was also significantly lower than the CON group.

To the authors knowledge, no reports related to adipose tissue *TRPV4* expression following dietary restriction such as calorie restriction or IF have been published. The lower expression of *TRPV4* in all intervention groups, albeit only significantly in IF group

131

and combined group (female only) suggests reduced inflammation and potentially enhanced oxidative metabolism and energy homeostasis within adipose tissue (Ye et al., 2012).

# 5.4.5 <u>Hypoxia</u>

High fat diet induced obesity leads to adipose tissue hypoxia with increased gene expression of HIF1 $\alpha$  (Fujisaka et al., 2013) an adipose tissue hypoxia marker gene (He et al., 2011). HIF1 $\alpha$  gene expression of IF only group in males showed significantly lower expression relative to CON group, indicating that reduction in fat mass due to IF was also accompanied by concomitant decrease in adipose tissue hypoxia. Previously, both short term (4 weeks) (Ye et al., 2007) and long term (8 months) (Higami et al., 2006) caloric restriction has been reported to decrease the expression of  $HIF1\alpha$  gene and protein content in epididymal adipose tissue of mice. Similarly, moderate intensity exercise training for 5 days per week has been reported to decrease the mRNA level of  $HIF1\alpha$  in epididymal adipose tissue of mice after 8 weeks (Ko and Kim, 2013, Ahn and Kim, 2014) and Wistar rats after 9 weeks (Sakurai et al., 2010). The HIIT only group in the present study showed a declining trend (p>0.07) relative to CON group, suggesting that HIIT was capable of minimizing hypoxia in adipose tissue despite lower intensity and frequency in comparison to the previous studies mentioned above. Also,  $HIF1\alpha$  is a direct target of miR-222 (Chartoumpekis et al., 2012). Expression of miR-222 was also higher in HIIT group relative to IF+HIIT group in males, indicating that lowering of  $HIF1\alpha$  gene expression in HIIT group could be due to inhibition by miR-222. This was further supported by significant (p<0.05) negative correlation (R=-0.68) found between HIF1 $\alpha$ and miR-222 expression in HIIT group in males. Gene expression of HIF1 $\alpha$  in IF+HIIT groups in males also showed a lower trend (p<0.07) relative to CON, indicating lower

level of hypoxia. Although individually IF and HIIT showed lower hypoxia, combination of IF with HIIT did not show any synergistic effect. Also, this effect of IF and/or HIIT was very gender specific and was only observed in males.

# 5.4.6 Adipose tissue browning

Interestingly, alternate day fasting (15 cycles of every other day fasting) in mice has recently been reported to induce white adipose tissue browning (Li et al., 2017a), however in the present study, IF with or without HIIT showed no change in adipose tissue browning, as no significant change in *UCP1* gene expression was observed in both males and females. The reason could be the pattern of fasting which was 5:2 (fast:feed day) in the present study and was less extensive (fasting period of 25% of total study period) in comparison to every other day (fasting period-50% of total study period) in the study by Li et al. (2017a). Another reason could be the difference in the WAT depot investigated which was epididymal in the present study and inguinal in the Li et al. (2017a) study. No browning of adipose tissue with exercise was not a big surprise as no change in *UCP1* expression has previously been reported by other studies after 11 weeks (Davis et al., 2017) and 5 weeks of HIIT training (Marcinko et al., 2015).

# 5.4.7 Expression of miRNAs

The effect of IF with or without HIIT appeared to have minimal impact on the expression of miRNA's investigated in the present study. In contrast, HIIT group demonstrated significantly higher expression in comparison to most groups, especially the control group. It is not readily apparent to the authors as to why the HIIT group displayed significantly higher levels of miR-24, 222, 143 and 145, which are involved in promoting adipogenesis and reducing glucose uptake and insulin sensitivity (Jin et al., 2016, Kang et al., 2013, Deiuliis, 2016). Interestingly, the body composition data in the

133

present study showed that the HIIT group actually increased their fat mass over time (not significantly), which was opposite to the both the IF and IF+HIIT groups. A confounding factor could be the continuation of high fat and sugar feeding while undertaking HIIT. Increase in appetite because of exercise could lead to higher intakes of fat and sugar. Although difference in food intake between exercise and non-exercise groups could not be determined in the present study due to ethical housing limitations, previous studies have reported higher food intake following acute HIIT in humans (Pomerleau et al., 2004) and eight weeks of HIIT in mice (Wang et al., 2017).

Another unexpected finding was the lower expression of these miRNAs in the control group compared to most other groups, albeit not significantly in most cases. This was opposite to what we would expect, given increase in fat mass within the CON group. This unexpected finding could be explained by pleiotropic effects of this class of molecules. For example, several miRNAs, including miR-143 and 222, exhibit inverse patterns of regulation during adipogenesis compared with later stages of obesity (Xie et al., 2009a), indicating that obesity leads to loss of miRNAs that characterize fully differentiated and metabolically active adipocytes. A major limitation in the current thesis is that we didn't measure the baseline expression values for each intervention group, therefore we are unable to determine the changes over time for each group. We can only determine the expression levels of each miRNA at one time point and compare to each group. The direction of change for miRNA expression may be different depending on the levels at baseline and these miRNAs are probably losing control over their stipulated function due to advancement in obesity state as reported by Xie et al. (2009a).

# 5.5. CONCLUSION

Body compositional analysis reported in chapter 4 suggested that IF with or without HIIT prevented weight gain by reducing fat mass accumulation in mice despite consuming high fat/sugar diet. Molecular investigation targeted in the present study on adipose tissue revealed that, lower fat mass accumulation observed in this group was associated with fragmentation of unilocular lipid droplets, enhanced lipolysis, fatty acid oxidation and efflux from adipose tissue. The results of the present study, suggest that combination of IF and HIIT, not only prevents fat accumulation by increasing fat oxidation and efflux but also promotes other cellular benefits like preventing inflammation and ameliorates the negative impact of high fat/high sugar diet on adipose tissue structure and functioning. This study also noticed that the insulin resistant state induced by IF in mice reported in chapter 4 was related to suppressed SIRT1 gene expression in adipose tissue of IF groups, providing a slight hint to the underpinning mechanism behind negative impact of IF on insulin sensitivity and needs an elaborate investigation. Intermittent fasting with or without HIIT does not seems to impact the expression of miRNA investigated in this study, however, HIIT induced the expression of most of them especially in males. This portion of the study needs more in-depth experimental investigation with true baseline and post intervention controls to understand the behaviour and involvement of these miRNAs on exercise induced adaptations in adipose tissue under high caloric intake.

# **CHAPTER 6**

# INTERMITTENT FASTING AND HIGH INTENSITY INTERVAL TRAINING DISPLAY OPPOSING EFFECTS ON MARKERS OF SKELETAL MUSCLE OXIDATIVE METABOLISM, INSULIN SIGNALLING, INFLAMMATION AND ATROPHY IN DIET-INDUCED OBESE MICE

# 6.1. INTRODUCTION

Both chapter 4 and 5 have described superior effects of IF when combined with HIIT at a whole body level, with significantly less weight gain, particularly in the form of adipose tissue, and at a molecular level, with greater changes in expression of genes relating to fatty acid oxidation, efflux, and fat accumulation compared to either IF or HIIT alone. Given the detrimental effects of HFD on skeletal muscle function such as development of insulin resistance and the reports that fasting may induce lean muscle mass loss, which is counterproductive to long term ability to maintain weight loss, we wanted to continue examine changes at whole muscle level and molecular level with gene and miRNA targets related to oxidative metabolism, mitochondrial function, inflammation, insulin signalling and atrophy.

High fat/high sucrose feeding for 28 days has been reported to increase adipogenesis, muscle atrophy and inflammation in muscle indicated by increased expression peroxisome proliferator activated receptor gamma (PPAR $\gamma$ ), muscle ring finger 1 (MURF1) and macrophage chemo attractant protein-1 (MCP1) (Collins et al., 2016). Intermittent fasting and HIIT have the capacity to reverse the negative effects of high calorie intake at body composition level and also in adipose tissue as mentioned in the previous chapters. In addition to this, few studies have also reported that IF in skeletal

136

muscle favours glycogen replenishment by increasing phosphorylation of glycogen synthase, and decreased muscle protein synthesis indicated by decreased phosphorylation of mammalian target of rapamycin (mTOR) (Soeters et al., 2009) and increased lipid oxidation indicated by increased expression of carnitine palmitoyltransferase I (CPT1) (Heilbronn et al., 2005a). Likewise, HIIT has been linked to enhanced skeletal muscle glucose uptake, mitochondrial biogenesis, improved insulin signalling indicated by enhanced expression of GLUT4 in obese mice (Davis et al., 2017), and diabetic humans (Little et al., 2011), increased peroxisome proliferator-activated receptor gamma coactivator 1-alpha (PGC1 $\alpha$ ) expression (Little et al., 2010), and increased insulinstimulated superoxide dismutase (SOD) activity, c-Jun N-terminal kinase (JNK), P38 mitogen-activated protein kinases (p38MAPK) and nuclear factor kappa-light-chainenhancer of activated B cells (NF-kB) phosphorylation in obese men (Parker et al., 2016). However, the effects of IF and HIIT alone or in combination on energy sensing pathways such AMPK/SIRT1, oxidative metabolism, insulin signalling and fibre type shifts require further investigation.

Relatively recently, microRNAs (miRNAs) have been reported to regulate various cellular pathways governing skeletal muscle structure and function in normal and obese state (Wilson et al., 2017). Muscle specific miRNAs like miR-1, miR-133a and miR-133b, known as myomirs are downregulated in skeletal muscle of high fat diet induced obese mice (Chen et al., 2012). In contrast, acute endurance training results in upregulation of miR-133a and miR-133b in skeletal muscle (Russell et al., 2013). Exercise induced increase in expression of miR-133a has been related to mitochondrial biogenesis (Nie et al., 2016). However, the fate of these and other muscle specific

miRNAs in response to IF and HIIT in diet induced obese mice is unknown and warrants investigation.

Thus, the objective of this study was to evaluate the effect of IF with and without HIIT on skeletal muscle expression of genes related to mitochondrial function, metabolism, muscle atrophy, insulin signalling, inflammation, expression of muscle specific miRNAs and changes in muscle mass and fibre type. We hypothesised that combination of IF and HIIT will show a synergistic effect on changing the muscle fibre composition from fast to slow-twitch fibre type and increase mitochondrial function, reestablish the co-ordinated upregulation of  $\beta$ -oxidation and downstream targets (TCA cycle, ETC and NADH shuttle system), which is generally not coordinated under obese conditions (Muoio et al., 2008) as well as will reduce inflammation and muscle atrophy in diet induced obese mice.

# 6.2. MATERIALS AND METHODS

## 6.2.1 Animals

Details of animal procurement, housing and diet are mentioned in chapter 3 section 3.2. Briefly, this study included four different groups namely: no intervention (CON; Males=7, Females=9); intermittent fasting (IF; Males=8, Females=10), high intensity intermittent exercise (HIIT; Males=8, Females=10) and a combination of the dietary and exercise intervention (IF+HIIT; Males=8, Females=10). All groups were maintained on high fat/high sugar (HF/S) diet for 12 weeks to induce obesity and then underwent respective interventions for another 12 weeks while still consuming HF/S diet. Details of interventions is given in chapter 3 section 3.4. After 12 weeks of HF/S feeding and intervention, all mice were killed and tested as described below.

# 6.2.2 <u>Culling and tissue collection</u>

Mice were anaesthetised with intraperitoneal injection of pentobarbitone (60 mg/kg). Once anaesthesia is conformed using pupillary and toe pinching reflex, skin is removed from the left and right hind legs. Left extensor digitorum longus (EDL) muscle was removed and snap frozen for RNA extraction. Right EDL and soleus muscles were removed, covered with Optimum Cutting Temperature (OCT) compound and froze in isopentane bath cooled in liquid nitrogen for fibre typing. All muscle samples were stored in -80°C freezer until further analysis.

#### 6.2.3 <u>Myosin ATPase Fibre typing</u>

EDL and soleus muscles were sliced to a thickness of 10µm in cryostat (Leica Biosystems) with chamber temperature set at -19°C and sample holder temperature set at -16°C. Tissue sections were collected on adhesive slides, air dried at room temperature and then kept at 4°C until further processing. Slides were incubated in acid pre-incubation solution (54.3mM sodium acetate and 32.6mM sodium barbital adjusted to pH 4.60) for 7 minutes. Slides were washed with a wash solution (pH 7.8) containing 18mM calcium chloride (CaCl<sub>2</sub>) and 100mM TRIS-HCl. Slides were then incubated in ATP incubation solution for 45 minutes at room temperature on a rocking platform. Following this slides were incubated in 11mM CaCl<sub>2</sub> solution for 3 minutes and this step was repeated three times. Slides were then incubated in 2% cobalt chloride (CoCl<sub>2</sub>) for 1.5 minutes and this step was repeated for another 5 times. Slides were then incubated for 30 seconds in 10mM Sodium barbital and this step was repeated for another 5 times. Slides were then incubated for 1 minute in 1% ammonium sulphide in a fume cupboard. Slides were then washed with tap water followed by 5 consecutive washings in distilled water. Slides were dehydrated in ethyl alcohol solutions of increasing concentration (70, 80, 90, 95 and

100%) by incubating the slides for 2 minutes in each alcohol solution. Slides were rinsed in xylene for 5 minutes. Coverslips were mounted on glass slides using Dibutylphthalate Polystyrene Xylene (DPX). Images were collected using an Olympus BX53 microscope with Olympus Cellsens imaging software. From each section 100 fibres (both types) were counted and percentage of type I and IIa in soleus and IIa and IIb in EDL was calculated out of them (Ranatunga and Thomas, 1990). Dark fibres were identified as type I in soleus and type IIb in EDL muscles and light fibres were identified as type IIa in both soleus and EDL muscles.

#### 6.2.4 Skeletal muscle gene and miRNA expression

Total RNA was extracted form whole left EDL muscle, reverse transcribed to generate cDNA and subjected to RT-PCR for gene and miRNA expression. Details of mRNA and miRNA isolation, cDNA synthesis, RT-PCR analysis and calculation of fold change are given in chapter 3, section 3.7. Peptidylprolyl isomerase A (*Ppia*) and RNU6 were used to normalize mRNA and miRNA expression data respectively. The primers for genes and miRNAs were purchased from and designed by QIAGEN (Australia). The list of genes and miRNAs targeted in this study are given in table 6.1 and table 6.2 respectively.

Gene	Official full name	NCBI Reference	Qiagen Catalogue
symbol	Official full hand	number	number
PGC1a	Peroxisome proliferator-activated		PPM03360I
	receptor gamma coactivator 1-alpha	NR_027710	
AMPK	Protein kinase AMP-activated		
	catalytic subunit alpha 2	NM_178143	PPM29410F
SIRT1	Sirtuin (silent mating type information		PPM05054A
	regulation 2 homolog) 1	NM_001159589	
CPT1	Carnitine palmitoyltransferase Ib	NM_009948	PPM57688A
CS	Citrate synthase	NM_026444	PPM29621B
COX-IV	Cytochrome c oxidase subunit 4		
	isoform 2	NM_053091	PPM32878E
ΤΝΓα	Tumor necrosis factor-alpha	NM_013693	PPM03113G
AS160	Akt substrate of 160 kDa	NM_001081278	PPM40317A
UCP3	Uncoupling protein 3	NM_009464	PPM25342F
MAFbx	F-box protein 32	NM_026346	PPM38061A
MuRF1	Muscle RING finger 1	NM_001039048	PPM61645B
Ppia	Peptidylprolyl isomerase A	NM_008907	PPM03717B

Table 6.1. List of targeted genes with full official names, NCBI reference sequence numbers and Qiagen catalogue number.

Table 6.2. List of miRNAs, miRbase accession numbers and Qiagen miScript Primer Assay Catalog Numbers.

Mature miRNA ID	miRBase Accession No.	Qiagen miScript Primer Assay Catalog No.
mmu-miR-696	MIMAT0003483	MS00002870
mmu-miR-133a-3p	MIMAT0000145	MS00007294
mmu-miR-133b-3p	MIMAT0000769	MS00007301

# 6.2.5 <u>Statistical analysis</u>

Data is presented as mean  $\pm$  standard deviation. To evaluate the effect of interventions, intervention groups namely HIIT, IF, and IF+HIIT were compared against no intervention CON group. Difference in mRNA expression and miRNA expression between groups were analysed using one-way ANOVA and Tukey's test was used as post hoc analysis. Correlation between mRNA and miRNA expression in each intervention group was calculated with Pearson correlation test for *PGC1a* Vs miR-696, *TNFa* Vs

miR-133a and *MURF1* Vs miR-133b. An alpha level of 0.05 was adopted throughout to reduce Type I statistical errors.

# 6.3. **RESULTS**

# 6.3.1 The effect of IF and/or HIIT on skeletal muscle mass and morphology

# 6.3.1.1. Muscle weight

Weight of soleus and EDL muscles following 12 weeks of IF and/or HIIT in both male and female mice are presented in Figure 6.1 (A&C) and (B&D) respectively. Muscle weight for both soleus and EDL muscles were similar between control and IF and/or HIIT groups at the end of the 12-week intervention period in both males and females.



Figure 6.1. Effect of IF, HIIT and IF+HIIT on weight (mg) of soleus muscle in (A) males and (B) females and EDL muscle in (C) males and (D) females. Bars represent means, error bars represent standard deviation.

# 6.3.1.2. *Muscle fibre type*

Muscle fibre type I and IIa in soleus muscle and IIa and IIb in EDL muscle following 12 weeks of IF and/or HIIT are presented as percentage (bars) in Figure 6.2 and 6.3 respectively. Representative images showing fibre type distribution in (A) soleus and (B) EDL muscle are presented in figure 6.3A and B respectively. No significant differences in fibre type distribution for both soleus and EDL muscle were noted between control and IF and/or HIIT groups at the end of the 12-week intervention period in both males and females.



Figure 6.2. Effect of IF, HIIT and IF+HIIT on soleus muscle type I fibre in (A) males and (B) females and type IIa fibre in (C) males and (D) females. Bars represent means, error bars represent standard deviation.



(B)



Figure 6.3. Representative images of (A) soleus and (B) EDL muscle showing fibre type distribution. Effect of IF, HIIT and IF+HIIT on ED muscle type IIa fibre in (C) males and (D) females and type IIb fibre in (E) males and (F) females. Bars represent means, error bars represent standard deviation.

# 6.3.2 <u>The effect of IF and/or HIIT on gene expression of AMPK-SIRT1-PGC1α</u> <u>axis</u>

6.3.2.1. AMPK

Muscle *AMPK* gene expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 6.4 A and B respectively. In males, *AMPK* gene expression in IF group was significantly lower (p<0.01) than IF+HIIT group. In females, no significant differences were observed in AMPK gene expression between control and IF and/or HIIT groups at the end of 12-week intervention period.



Figure 6.4. Effect of IF, HIIT and IF+HIIT on *AMPK* gene expression (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *ppia* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*) bracket represent significant difference between group at p<0.05 level of significance.

# 6.3.2.2. SIRT1

Muscle *SIRT1* gene expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 6.5 A and B respectively. *SIRT1* gene expression in IF group was significantly lower than both CON and HIIT groups in both males and females (p<0.01), and significantly lower than IF+HIIT group in males only (p<0.01). The IF+HIIT group displayed significantly lower expression of *SIRT1* in comparison to HIIT group in females (p<0.01) but not in males.



Figure 6.5. Effect of IF, HIIT and IF+HIIT on *SIRT1* gene expression (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *ppia* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*\*) bracket represent significant difference between group at p<0.01 level of significance.

# *6.3.2.3. PGC1α*

Muscle *PGC1a* gene expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 6.6 A and B respectively. Gene expression of *PGC1a* in IF group was significantly lower than CON group (p<0.05) in males and significantly lower than HIIT groups (p<0.05) in females. Also in females, gene expression of *PGC1a* in IF group showed lower trend (p<0.064) than CON group.



Figure 6.6. Effect of IF, HIIT and IF+HIIT on *PGC1a* gene expression (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *ppia* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*) bracket represent significant difference between group at p<0.05 level of significance.

# 6.3.3 <u>The effect of IF and/or HIIT on gene expression related to oxidative</u> <u>metabolism</u>

6.3.3.1. CPT1

Muscle *CPT1* gene expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 6.7 A and B respectively. *CPT1* gene expression in IF group was significantly (p<0.01) lower than HIIT group in both males and females and both CON (p<0.01) and IF+HIIT (p<0.01) groups in males only.



Figure 6.7. Effect of IF, HIIT and IF+HIIT on *CPT1* gene expression (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *ppia* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*\*) bracket represent significant difference between group at p<0.01 level of significance.

#### 6.3.3.2. Citrate Synthase

Muscle citrate synthase (*CS*) gene expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 6.8 A and B respectively. Citrate synthase (*CS*) gene expression in CON group was significantly (p<0.05) higher than IF group in males. In females, no significant differences were observed in *CS* gene expression between different groups at the end of 12-week intervention period.



Figure 6.8. Effect of IF, HIIT and IF+HIIT on citrate synthase (*CS*) gene expression (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *ppia* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*) bracket represent significant difference between group at p<0.05 level of significance.

# 6.3.3.3. COX-IV

Muscle *COX-IV* gene expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 6.9 A and B respectively. *COX-IV* gene expression in IF+HIIT group was significantly (p<0.05) higher than HIIT group in females. In females, IF+HIIT group showed a higher trend (p<0.064) for *COX-IV* gene expression relative to CON group. In males, no significant differences were observed in *COX-IV* gene expression between different groups at the end of 12-week intervention period.



Figure 6.9. Effect of IF, HIIT and IF+HIIT on *COX-IV* gene expression (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *ppia* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*) bracket represent significant difference between group at p<0.05 level of significance.

#### 6.3.3.4. UCP3

Muscle *UCP3* gene expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 6.10 A and B respectively. In both males and females gene expression of *UCP3* in CON group was significantly higher(p<0.01) than IF group. Also, in both males and females, *UCP3* gene expression in IF group was significantly (p<0.01) lower than both HIIT and IF+HIIT groups. *UCP3* gene expression of HIIT group was higher in comparison to CON in males (p<0.05).



Figure 6.10. Effect of IF, HIIT and IF+HIIT on *UCP3* gene expression (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *ppia* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*) and (\*\*) bracket represent significant difference between group at p<0.05 and p<0.01 level of significance respectively.
### 6.3.4 <u>The effect of IF and/or HIIT on gene expression related to Inflammation</u> 6.3.4.1. *TNF*

Muscle TNF $\alpha$  gene expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 6.11 A and B respectively. In males, *TNF* $\alpha$  gene expression in CON group was significantly (p<0.01) higher than HIIT, IF and IF+HIIT groups in males but not in females. There were no significant differences in *TNF* $\alpha$  gene expression between IF, HIIT and IF+HIIT intervention groups in both male and females at the end of 12-week intervention period.



Figure 6.11. Effect of IF, HIIT and IF+HIIT on *TNFa* gene expression (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *ppia* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*\*) bracket represent significant difference between group at p<0.01 level of significance.

## 6.3.5 <u>The Effect of IF and/or HIIT on gene expression related to Insulin</u> signalling

6.3.5.1. AS160

Muscle AS160 gene expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 6.12 A and B respectively. In males, AS160 gene expression in IF+HIIT group was significantly higher (p<0.01) than IF group. In females, no significant differences were observed in AS160 gene expression between different groups at the end of 12-week intervention period.



Figure 6.12. Effect of IF, HIIT and IF+HIIT on *AS160* gene expression (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *ppia* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*\*) bracket represent significant difference between group at p<0.01 level of significance.

# 6.3.6 <u>The effect of IF and/or HIIT on gene expression related to muscle atrophy</u>6.3.6.1. MAFbx

Muscle *MAFbx* gene expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 6.13 A and B respectively. Gene expression of *MAFbx* in CON group was significantly lower (p<0.01) than HIIT group in males. Also, HIIT group showed significantly higher expression relative to both IF (p<0.01) and IF+HIIT (p<0.01) groups. Gene expression of *MAFbx* in IF group was significantly lower (p<0.01) than IF+group was significantly lower (p<0.01) than IF+HIIT group in males but in females, no significant difference was observed in both these groups. In females, *MAFbx* gene expression was significantly higher (p<0.01) in both CON and HIIT groups in comparison to both IF and IF+HIIT groups.



Figure 6.13. Effect of IF, HIIT and IF+HIIT on *MAFbx* gene expression (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *ppia* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*\*) bracket represent significant difference between group at p<0.01 level of significance.

#### 6.3.6.2. MURF1

Muscle *MuRF1* gene expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 6.14 A and B respectively. *MuRF1* gene expression in CON group was significantly lower than HIIT group in both males (p<0.01) and females (p<0.05). In both males and females, HIIT group showed significantly higher expression of *MuRF1* gene in comparison to both IF (p<0.01) and IF+HIIT (p<0.01) groups. In females, *MuRF1* gene expression was significantly higher in CON group in comparison to IF (p<0.01) and IF+HIIT (p<0.01) groups.



Figure 6.14. Effect of IF, HIIT and IF+HIIT on *MURF1* gene expression (mRNA fold change) relative to CON group in (A) males and (B) females. Gene (mRNA) expression is normalized to *ppia* gene expression. Bars represent means, error bars represent standard deviation. Asterisk (\*) and (\*\*) bracket represent significant difference between group at p<0.05 and p<0.01 level of significance respectively.

#### 6.3.7 The effect of IF and/or HIIT on expression of myomiRs

#### 6.3.7.1. miR-696

Muscle miR-696 gene expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 6.15 A and B respectively. In males, miR-696 showed significantly higher expression in IF+HIIT group in comparison to CON (p<0.05) and HIIT (p<0.05) groups. In females, miR-696 expression was significantly higher in both IF (p<0.05) and IF+HIIT (p<0.05) groups in comparison to CON group.



Figure 6.15. Effect of IF, HIIT and IF+HIIT on miR-696 expression (fold change) relative to CON group in (A) males and (B) females. MiRNA expression is normalized to *RNU6* expression. Bars represent means, error bars represent standard deviation. Asterisk (\*) bracket represent significant difference between group at p<0.05 level of significance.

#### 6.3.7.2. miR-133a

Muscle miR-133a gene expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 6.16 A and B respectively. In females, IF+HIIT group showed significantly lower expression of miR-133a in comparison to CON (p<0.05), IF (p<0.05) and HIIT (p<0.05) groups. In males, no significant differences in miR-133a expression were observed between different groups at the end of 12-week intervention period.



Figure 6.16. Effect of IF, HIIT and IF+HIIT on miR-133a expression (fold change) relative to CON group in (A) males and (B) females. MiRNA expression is normalized to *RNU6* expression. Bars represent means, error bars represent standard deviation. Asterisk (\*) bracket represent significant difference between group at p<0.05 level of significance.

#### 6.3.7.3. miR-133b

Muscle miR-133b gene expression following 12 weeks of IF and/or HIIT in both male and female mice are presented as fold changes relative to CON group in Figure 6.17 A and B respectively. In females, IF group showed significantly higher (p<0.05) expression of miR-133a in comparison to CON (p<0.05) and IF+HIIT (p<0.05) groups. No significant differences were observed in miR-133a expression between HIIT group and CON and other intervention groups. In males, no significant differences in miR-133a expression were observed between different groups at the end of 12-week intervention period.



Figure 6.17. Effect of IF, HIIT and IF+HIIT on miR-133b expression (fold change) relative to CON group in (A) males and (B) females. MiRNA expression is normalized to *RNU6* expression. Bars represent means, error bars represent standard deviation. Asterisk (\*) bracket represent significant difference between group at p<0.05 level of significance.

#### 6.3.8 Correlation analysis between mRNA and miRNA expression

Pearson correlation coefficient (R) and significance values (p) between gene and miRNA expression in males and females are given in table 6.3 and 6.4 respectively. Gene expression and miRNA expression between each group in both males and females did not show any significant correlation in any of the compared mRNA and miRNAs at the end of 12-week intervention period.

 Table 6.3. Pearson correlation coefficient (R) between mRNA and miRNA expression in males.

	PGC1a vs	TNFα Vs	MURF1 Vs
	miR-696	miR-133a	miR-133b
CON	-0.31 (0.30)	0.44 (0.28)	0.77 (0.06)
HIIT	0.21 (0.32)	-0.25 (0.30)	0.14 (0.39)
IF	-0.28 (0.26)	-0.52 (0.12)	-0.43 (0.17)
IF+HIIT	0.31 (0.23)	-0.52 (0.10)	-0.53 (0.09)

Table 6.4. Pearson correlation coefficient (R) between mRNA and miRNA expression in females.

	<i>PGC1α</i> vs miR-696	<i>TNFa</i> Vs miR-133a	MURF1 Vs miR-133b
CON	-0.36 (0.21)	-0.20 (0.35)	-0.29 (0.29)
HIIT	0.84 (0.01)	-0.42 (0.21)	-0.21 (0.35)
IF	-0.27 (0.25)	-0.20 (0.31)	0.61 (0.06)
IF+HIIT	-0.42 (0.16)	-0.34 (0.21)	-0.39 (0.17)

#### 6.4. **DISCUSSION**

In the present study, short burst of high intensity interval training resulted in increased mitochondrial biogenesis, oxidative metabolism, fatty acid oxidation and ATP production in skeletal muscle. However, IF had an opposite effect on skeletal muscle manifested as lower mitochondrial biogenesis, oxidative metabolism, fatty acid oxidation, ATP production and insulin sensitivity suggested by the expression of marker genes related to them. The combination of IF and HIIT did not show any synergistic effect, however various marker genes especially related to oxidative metabolism (SIRT1, AMPK, CPT1, AS160) showed the expression level higher than IF only group indicating that HIIT was able to rescue their expression level and mitigating the effect of IF on lowering metabolism. However, this role is reversed in atrophic gene expression as atrophy induced by HIIT was mitigated by IF in IF+HIIT in both males and females. Although the effect was gender specific, both IF and HIIT alone or in combination showed decline in obesity induced inflammation in males. Even though no superior effect of combining IF with HIIT has been observed, combination of these two interventions can induce beneficial adaptations in skeletal muscle with either one of them compensating the unfavourable effects of other and keeping the balance.

#### 6.4.1 <u>AMPK-SIRT1-PGC1α signalling axis</u>

Activation of AMPK and SIRT1 is necessary for induction of PGC1 $\alpha$  which is a positive regulator of mitochondrial biogenesis and respiration, adaptive thermogenesis, gluconeogenesis as well as many other metabolic processes (Austin and St-Pierre, 2012). Conditions that create energy deficit like exercise and reduced caloric intake, upregulate AMPK/SIRT1/PGC1 $\alpha$  signalling axis (Canto and Auwerx, 2009). For instance, *PGC1\alpha* expression increases in skeletal muscle with acute exercise (running with swimming) in

rats (Terada and Tabata, 2004), chronic exercise training for 10-days in humans (Stepto et al., 2012) and acute HIIT in men (Gibala et al., 2009). In the present study, females in HIIT group showed higher expression of PGC1a compared to IF only group. Likewise, HIIT group showed higher expression of SIRT1 in comparison to IF in both males and females and compared to IF+HIIT group in females. Higher expression of SIRT1 has been reported previously after single bout of endurance exercise in skeletal muscle of rats (Suwa et al., 2008) and after six weeks of HIIT in humans (Gurd et al., 2010). Higher SIRT1 expression in these groups suggests energy deficit which increases NAD<sup>+</sup>/NADH ratio, which in turns triggers the activity of SIRT1 and likewise the expression level also (Canto and Auwerx, 2009). Higher expression of SIRT1 leads to deacetylation of phosphorylated PGC1a. Phosphorylated PGC1a activates its own promoter and enhances its expression (Ruderman et al., 2010), which in turn upregulates mitochondrial biogenesis and respiration, adaptive thermogenesis and gluconeogenesis (Austin and St-Pierre, 2012), increased capacity for generation of ATP, increased angiogenesis and increased glucose uptake, likely via increase in GLUT4 expression (Chan and Arany, 2014, Wende et al., 2007, Michael et al., 2001). In addition to this, expression of UCP3 in HIIT group was higher in comparison to IF only group in both males and females which may be attributed to higher triglyceride content of skeletal muscle and increased markers of de novo lipogenesis in response to exercise, which is typical following HIIT (Davis et al., 2017). UCP3 has been hypothesised to provides protection against (lipid-induced) mitochondrial dysfunction or oxidative stress (Schrauwen et al., 2001) and its higher expression in HIIT group suggests its beneficial effects on skeletal muscle.

Caloric restriction induced negative energy balance has been shown to enhance the expression of PGC1 $\alpha$  in skeletal muscle of overweight males and females after 6months (Civitarese et al., 2007) and enhanced expression of SIRT1 in high fat diet fed mice after 3 months (Chen et al., 2008a). Intermittent fasting is considered to induce body compositional changes similar to caloric restriction and transcriptional profile of calorie restriction and intermittent fasting has been found to be similar (Pearson et al., 2008), suggesting that IF should induce AMPK/SIRT1/PGC1α signalling axis. Surprisingly, all the components of this axis were found to be suppressed in IF group in the present study. Intermittent fasting group showed lower  $PGC1\alpha$  gene expression in comparison to CON in males and HIIT group in females. Along the same line, IF group showed lower expression of AMPK in comparison to IF+HIIT in males. Also, SIRT1 expression in IF group was lower in comparison to CON and HIIT groups in both males and females and also compared to IF+HIIT group in males. Downregulation of AMPK/SIRT1/PGC1a observed in IF group could be the reason behind lower glucose tolerance and insulin sensitivity observed in IF mice in the present study, as upregulation of  $PGC1\alpha$  is related to insulin sensitivity (Gastaldi et al., 2007). Previous IF studies have reported no change in PGC1 $\alpha$  expression in response to 22 days of alternate day fasting in human skeletal muscle (Heilbronn et al., 2005a). This also suggests that IF may not simply be operating at the level of energy balance and there may be other signalling pathways through which it is able to induce the beneficial effects acclaimed by several studies. Intermittent fasting in both males and females, also resulted in lower expression of UCP3 in comparison to CON, HIIT and IF+HIIT groups. Similar decline in UCP3 expression has been reported by other weight loss associated calorie restriction (Davis et al., 2017, Schrauwen et al., 2000) studies, indicating reduced resting metabolic rate and/or increased metabolic efficiency.

Contrary to our hypothesis, combination of IF and HIIT did not show any synergistic effect on AMPK/SIRT1/PGC1 $\alpha$  and UCP3 expression. However, expression of *AMPK*, *SIRT* and UCP3 was higher in IF+HIIT group in males in comparison to IF only group. In females, UCP3 expression was higher and AMPK gene expression showed a higher trend (p<0.09) in comparison to IF only group. In addition, PGC1 $\alpha$  gene expression in IF+HIIT group for both males and females, did not show any significant decline despite IF regime similar to IF group. This suggested that although combination of IF with HIIT may not show any synergistic effect, HIIT can prevent the suppression of these energy sensing molecules during IF in skeletal muscle.

#### 6.4.2 Oxidative metabolism

Mitochondrial fatty acid oxidation ( $\beta$ -oxidation) is regulated by CPT1, a mitochondrial transmembrane enzyme and is rate limiting for long-chain fatty acid entry into the mitochondria for  $\beta$ -oxidation (Bruce et al., 2009). Inhibition of CPT1 increases lipid deposition and aggravates insulin resistance in diet induced obese mice (Stephens et al., 2007, Kim et al., 2000). Intermittent fasting group in both males and females in the present study showed lower CPT1 expression in comparison to HIIT only group in both males and females and both CON and IF+HIIT groups in males. This suggested that IF supresses  $\beta$ -oxidation of fatty acids in skeletal muscle. Also, Citrate synthase (*CS*) expression was lower in IF group in comparison to CON indicating lowered oxidative capacity of IF males as *CS* is a biomarker for mitochondrial density and oxidative capacity (TCA cycle) of skeletal muscle (Vigelsø et al., 2014, Li et al., 2016). Also, the COX-IV expression, indicator of oxidative phosphorylation capacity in skeletal muscle (Larsen et al., 2012), was lower, albeit non-significantly (p<0.11) in males IF group. Indicating that IF lowers skeletal muscle fatty acid oxidation, and downstream TCA cycle and oxidative

phosphorylation. Previous IF study has reported no change in *CS* expression in response to 22 days of alternate day fasting in human skeletal muscle (Heilbronn et al., 2005a). Although no previous reports on *COX-IV* expression after IF were found in the literature, calorie restriction has been reported to cause no decline in electron transport chain (ETC) activity in SM of overweight humans (Civitarese et al., 2007, Toledo et al., 2008). The cause of decline in skeletal muscle oxidative metabolism due to IF observed in present study could be an adaptive response to conserve energy, however more elaborative study is required to understand he underpinning mechanism.

High intensity interval training for 10 weeks and 6 weeks of aerobic exercise training, increases gene expression of CPT1 in skeletal muscle of high fat diet fed rats (Shen et al., 2015) and mice (Niu et al., 2010) respectively. Likewise, in the present study, HIIT group showed higher expression of CPT1 in comparison to IF only groups in both males and females, which suggests higher  $\beta$ -oxidation of fatty acids in skeletal muscle. Also, Skeletal muscle CPT1 overexpression has been reported to protects rats from high fat diet induced insulin resistance (Bruce et al., 2009). Despite higher  $\beta$ oxidation level, no concomitant increase in CS and COX-IV gene expression in HIIT groups in both males and females were observed. Although previously, increase in CS activity after 6 weeks of voluntary wheel running has been reported in chow fed mice (Palacios et al., 2009, Röckl et al., 2007). Discrepancy in the results could be due to difference in the diet, as the mice in the present study were on HF/S diet before and during the HIIT period in comparison to standard chow diet reported by the aforementioned studies. Moreover, the total distance covered per day reported by one of these studies (Rockl et al., 2007) is 4.5km/day which is far less than the distance covered by mice in the present study (150m/day) although at higher intensity.

Combination of IF and HIIT in the present study showed higher expression of CPT1 in comparison to IF only group in males indicating that the inhibiting effect of IF on skeletal muscle β-oxidation was compensated by HIIT. Although, no enhancement in CPT1 expression in IF+HIIT group in females and CS expression in both males and females was observed, their CPT1 and CS expression did not show significant decline despite IF. This shows that combination of HIIT with IF can prevent lowering of CPT1 and CS expression in the muscle, which tends to fall with IF only. Expression of COX-IV in IF+HIIT group in females was higher in comparison to HIIT only group, indicating enhanced ETC activity and ATP production. Similar results were reported by Menshikova et al. (2005) where calorie restriction when combined with exercise increased ETC activity in skeletal muscle of humans, although exercise control group was not included in the study. One commonality between this study and the present study is lower body mass and fat mass in the combination group which could be the reason behind enhanced ETC expression or activity.

Although the exact mechanism of insulin resistance still remains elusive, in the state of positive energy,  $\beta$ -oxidation is induced, however the downstream TCA cycle and ETC remain inactive, causing accumulation of incomplete fatty acid oxidation products, hypothesised to cause insulin resistance in skeletal muscle (Muoio et al., 2008). However, lifestyle interventions (dietary and exercise) are expected to simultaneously upregulate  $\beta$ -oxidation and downstream TCA cycle and ETC. In females, higher expression of *CPT1* with concomitant higher expression of *CS* (although non-significantly) and *COX-IV* in IF+HIIT group, suggest that there was increase in fat oxidation ( $\beta$ -oxidation) with simultaneous increase in TCA cycle and ETC, which according to the Muoio et al. (2008), should have improved the insulin sensitivity. However, no improvement in insulin

sensitivity at systemic level (indicated by HOMA-IR, (chapter 4, section 4.3.5)) or at skeletal muscle level (indicated by AS160 expression) was observed which suggested that the hypothesis, proposed by Muoio et al. (2008) may not be the only mechanism through which insulin sensitivity is regulated.

#### 6.4.3 Insulin signalling

Skeletal muscle are primarily responsible for systemic insulin mediated glucose uptake and disposal (Ijuin et al., 2015). Relatively recently, Akt substrate of 160 kDa (AS160), a Rab GTPase-activating protein has been reported to regulates GLUT4 translocation in adipocytes and muscle (Kramer et al., 2006, Kane et al., 2002) under both insulin stimulated and exercise induced states (Parker et al., 2016, Parker et al., 2017). Activity of AS160 (phosphorylation) has been reported to increase after acute endurance exercise (Deshmukh et al., 2006), acute HIIT (Parker et al., 2017) and 12 weeks of endurance- or strength-oriented exercise training in humans with improvement in insulin sensitivity (Consitt et al., 2013). Although the expression and activity of AS160 is affected by insulin and exercise, this study was primarily interested in exercise-induced changes. In the present study no significant change in AS160 expression was observed in HIIT only group. However, IF+HIIT group in males showed higher expression of AS160 relative to IF only group. Higher expression of AS160 in the combination (IF+HIIT) group suggests that HIIT when combined with IF can improve GLUT4 translocation and skeletal muscle glucose homeostasis. Lower expression of AS160 in IF only group in males suggests that IF hampers AS160 mediated insulin signalling and glucose uptake and it could also be the reason behind poor glucose tolerance in IF group reported in chapter 4.

#### 6.4.4 Inflammation

Several studies have suggested that skeletal muscle inflammation occurs in obesity (Collins et al., 2016) and is evidenced by increased expression of proinflammatory cytokines like TNF $\alpha$  (Borst and Conover, 2005). In the present study, TNF $\alpha$  expression in ad libitum HF/S fed CON males was significantly (p<0.01) higher in comparison to all other intervention groups. Resistance training in human males has been reported to lower skeletal muscle TNF $\alpha$  gene and protein expression (Greiwe et al., 2001). Similarly, calorie restriction has also been reported to lower muscle TNF $\alpha$  expression (Phillips and Leeuwenburgh, 2005). These reports are in line with our findings that HIIT, IF and their combination is effective for reducing skeletal muscle inflammation in males. Although noticed earlier in other anthropometric and plasma parameters, females respond very different to males in context to TNF $\alpha$  expression with no significant difference between CON and/between any intervention group. Estrogen has been reported to affect TNF $\alpha$  expression (Rink, 1998), which could be the reason behind the difference in its expression in male and female mice observed in the present study.

#### 6.4.5 <u>Muscle atrophy</u>

Muscle atrophy has been reported as one of the repercussions of obesity (Roy et al., 2016). Muscle ring finger 1 (MURF1) and muscle atrophy F-box (MAFbx) are important regulators of protein degradation and gene expression in muscle atrophy. Expression of MURF1 has been reported to increase in skeletal muscle of diet induced obese rodents (Sishi et al., 2011, Le et al., 2014). Expression of *MURF1* and *MAFbx* has been reported to upregulate after fasting (1-2 days) (Yamamoto et al., 2015) and calorie restriction (30 months) in mice (Hepple et al., 2008). However, voluntary exercise for 6-weeks has been reported to reduce the expression of both *MURF1* and *MAFbx* in mice

(Hojman et al., 2014). Contrary to this, males in HIIT group in the present study showed higher expression of both *MURF1* and *MAFbx* in comparison to CON group. However, IF group showed lower expression of both *MURF1* and *MAFbx* in comparison to HIIT only group in both males and females and CON group in females only. The discrepancy between results reported by this study and previous studies could be due to different modes of diet restriction and exercise training. Also, IF+HIIT showed lower expression of both *MURF1* and *MAFbx* in comparison to HIIT and CON group in females. This suggests that, despite HIIT inducing muscle atrophy, IF in this group prevented muscle atrophy by keeping down the expression of both *MURF1* and *MAFbx* despite simultaneous consumption of high calorie diet.

#### 6.4.6 <u>Muscle mass and fibre type</u>

No significant difference in muscle mass was found between CON and/between intervention groups in both males and females. Although HIIT has been reported to enhance muscle mass of soleus and EDL muscle in chow fed mice after 16 weeks (Seldeen et al., 2018). No change in muscle mass could be due to difference in diet as the mice in the present study were on HF/S diet which might have hampered the effect of HIIT on muscle growth.

No significant difference in fibre type distribution was found between CON and/between intervention groups in both males and females. Although, moderate exercise training for 16 weeks has been reported to increase type I fibres (Dube et al., 2008, Pattanakuhar et al., 2016) in humans. Also, HIIT for 10 weeks (3 days per week) has been reported to reduce the proportion of type IIb fibres with no change in type I and type IIa fibres (Ross et al., 2016, Vogiatzis et al., 2005) in humans. However, in the present study, no change in fibre type distribution has been found either in type I, type IIa or IIb fibres in EDL or soleus muscle. A plausible cause could be the volume of exercise (Pattanakuhar et al., 2016), which as mentioned previously was of lower volume in comparison to the studies showing change in fibre type distribution.

#### 6.4.7 <u>MyomiR expression</u>

Skeletal muscle enriched miR-133a and 133b, regulate skeletal muscle development and differentiation (Koutsoulidou et al., 2011, Silva et al., 2017). MiR-133a and 133b has been reported to get upregulated in skeletal muscle of humans after acute endurance training (Russell et al., 2013). In the present study, HIIT female mice showed higher expression of miR-133a in comparison to IF+HIIT group. Higher expression of miR-133a has been linked to mitochondrial biogenesis in mice after six weeks of endurance training (Nie et al., 2016). This was also consistent with higher *PGC1a* expression found in female HIIT group.

Contrary to the above relationship, lower expression of miR-133a has been suggested to be beneficial as miR-133a can target members of the IGF-1/Akt pathway in vitro, including IGF-1, the IGF-1 receptor, as well as heat shock protein 70 (HSP70) (Fyfe et al., 2016, Elia et al., 2009). In females, IF+HIIT showed significantly lower expression of miR-133a in comparison to all other groups. Lower expression of miR-133a has been suggested to induce anabolic response by alleviating the repression of IGF-1/Akt signalling pathway and promoting muscle hypertrophy. Lower expression of these miRNAs in IF+HIIT in the present study suggests induction of muscle hypertrophy by combining IF with HIIT, although muscle weight was not significantly different from other groups.

MiRNA-696 has been reported as physical activity dependent miRNA and regulates the expression of PGC1 $\alpha$  (Aoi et al., 2010). Expression of miR-696 in IF+HIIT

170

group was significantly higher compared to CON group in both males and females and compared to HIIT group in males, indicating that weight loss induced by IF+HIIT upregulates the expression of miR-696. Contrary to the expectation, no correlation was found between miR-696 and PGC1 $\alpha$ . This could be attributed to translational mode of gene regulation rather than transcriptional gene regulation by miR-696. Translational regulation by miR-696 was also reported by another study which observed a correlation coefficient (R) of 0.64 between miR-696 expression and PGC1 $\alpha$  protein expression (Aoi et al., 2010).

In addition to this, miR-696 has also been reported to contribute to insulin resistance by reducing mitochondrial function and antioxidant capacity in skeletal muscle cells (C2C12) (Queiroz et al., 2013). The study on C2C12 skeletal muscle cells reported elevated expression of miR-696 with simultaneous reduction in insulin sensitivity (Queiroz et al., 2013). Likewise, in the present study, miR-696 and insulin resistance (HOMA-IR) increased in IF group and IF+HIIT in females, further confirming the previous report on C2C12 cells and indicating that upregulation of miR-696 in IF mice could be involved in the insulin resistance noticed in these animals by our study and other studies as well (Cerqueira et al., 2011). However, at the same time, increased expression of miR-696 in IF+HIIT males did not conform to above relationship between miR-696 expression and insulin sensitivity, as no significant difference in insulin resistance was noticed between IF+HIIT group and other intervention groups (chapter 4, section 4.3.6).

In silico gene target identification using miRWalk2.0 online database showed that miR-133a targets  $TNF\alpha$ , and miR-133b targets MURF1. However, no significant correlation was found between these miRNAs and their presumed target genes in the

171

present study. Just like miR-696, these miRNAs could have been regulated at translational level and need more specific studies to elaborate their function.

#### 6.5. CONCLUSION

This study was aimed to investigate the effect of IF and/or HIIT on skeletal muscle gene expression related to oxidative metabolism, inflammation, muscle growth and atrophy, fibre type distribution and miRNAs related to metabolism and muscle growth. This study suggests that HIIT induce higher mitochondrial biogenesis, oxidative metabolism, fatty acid oxidation and ATP production. However, IF alone had a very different and opposite response which lowers mitochondrial biogenesis, oxidative metabolism, fatty acid oxidation, ATP production and insulin sensitivity suggested by the expression of marker genes related to them. The combination of IF and HIIT did not show any synergistic effect on the muscle, however various marker genes especially related to metabolism (SIRT1, AMPK, CPT1, AS160) showed the expression level higher than IF only group indicating that HIIT was able to rescue their expression level and mitigating the effect of IF on metabolism. However, this role is reversed in terms of atrophy genes as induced by HIIT was mitigated by IF in IF+HIIT in both males and females. Although the effect was very gender specific, both IF and HIIT alone or in combination showed decline in obesity induced inflammation. Even though no superior effect of combining IF with HIIT has been observed, combination of these two interventions can induce beneficial adaptations in skeletal muscle with either one of them compensating the unfavourable effects of other and keeping the balance. Expression of miR-696, 133a and 133b although did not show any significant correlation with their reported or in silico identified targets, their differential expression between different groups and their gender specificity suggests their role in fine tuning the adaptive changes induced by IF and HIIT.

#### **CHAPTER 7**

#### **CONCLUSION, LIMITATIONS AND FUTURE DIRECTIONS**

The overall purpose of this thesis was to investigate the effect of two lifestyle interventions, IF and HIIT, alone or in combination, in high fat/sugar diet induced obese mice, on body composition, glucose homeostasis, plasma lipid profile (discussed in chapter 4), molecular changes underpinning adaptations in adipose tissue (discussed in chapter 5) and skeletal muscle (discussed in chapter 6). This chapter will summarise the main findings of this study in the first part of the chapter. Last two sections of the chapter will discuss limitations and will also provide future direction.

#### 7.1. KEY FINDINGS

## 7.1.1 <u>Combination of IF and HIIT prevents weight gain by less accumulation of fat mass and maintenance of lean mass</u>

Twelve weeks of IF combined with HIIT prevented weight gain in both males and females while concurrently consuming a high fat/sugar diet in mice. These mice showed no significant gain in body weight over 12-weeks, despite concurrently consuming high caloric diet. Whereas, ad libitum CON group on the same diet showed significant increase in body weight over this period. In males, IF alone also showed very similar results, with significantly no weight gain over 12-weeks intervention period. Contrary to this, females in IF group showed significant gain in body weight. Although previous studies on intermittent fasting have shown reduced weight gain (Higashida et al., 2013, Gotthardt et al., 2016), the discrepancy was due to less intensive fasting regime with few fasting days used in this study compared to the previous studies. Also, HIIT alone led to gain in body weight in both males and females. This does not agree with literature as previous HIIT studies in similar animal models have reported less gain in body weight (Cho et al., 2015, Lund et al., 2015). This was mainly due to more number of exercise training sessions performed by animals in the previous studies in comparison to the present study. Interestingly, despite low intensity of exercise and less number of fasting days, combination of these interventions led to reduced weight gain in both males and females. The change in body weight noticed in IF+HIIT group in both males and females is mainly due to low fat mass as lean mass in this group did not show any significant difference from *ad libitum* CON group. This suggests that combination of IF and HIIT can be beneficial for preventing weight gain mainly by reduced fat accumulation while maintaining lean mass. The combination of IF and HIIT also showed lower plasma TAG levels in females and lower LDL levels in both males and females.

#### 7.1.2 <u>Combination of IF with HIIT leads to less fat accumulation due enhanced</u> <u>fatty acid oxidation and efflux</u>

Molecular investigation in adipose tissue showed that IF+HIIT group displayed lower expression of genes related to the formation of unilocular lipid droplets (CIDEC) in females, suggesting fragmentation of lipid droplets and enhanced lipolysis (Marcinkiewicz et al., 2006). This was accompanied by higher expression of marker genes for fatty acid oxidation (HADH) relative to IF only group in both males and females and higher expression of fatty acid trafficking marker (FABP4) in comparison to other intervention and CON groups in males. These results suggest that reduced fat mass accumulation in adipose tissue of IF+HIIT group could be due to higher oxidation and efflux of fatty acids (Verhoef et al., 2013). Lower expression of leptin and proinflammatory markers were also observed in the combined diet and exercise group supporting the findings in chapter 4.

#### 7.1.3 <u>Intermittent fasting and high intensity exercise had opposite effect on</u> <u>skeletal muscle molecular markers</u>

In skeletal muscle, HIIT induced higher mitochondrial biogenesis, oxidative metabolism, fatty acid oxidation and ATP production. Contrary to this, IF alone lowered mitochondrial biogenesis, oxidative metabolism, fatty acid oxidation, ATP production and insulin sensitivity suggested by lower expression of marker genes related to them. Fasting group also showed lower expression of AMPK-SIRT1-PGC1 $\alpha$  axis. However, IF group showed lower expression of MURF1 and MAFbx indicating lower muscle atrophy, whereas HIIT showed higher expression of these genes. Although IF and HIIT did not show any synergistic effect on various molecular markers, combination of IF and HIIT can promote favourable adaptations in skeletal muscle with either one of them compensating the unfavourable effects of other and keeping the balance.

#### 7.1.4 Intermittent fasting perturbed the systemic glucose homeostasis

Surprisingly, after 12-weeks intervention period, IF group in both males and females and IF+HIIT group in females showed higher fasting glucose, plasma insulin, higher insulin resistance and poor glucose tolerance. This is very unexpected as most of the studies (Barnosky et al., 2014) have reported improvement in glucose homeostasis. This could be due to redox imbalances and insulin receptor nitration (Cerqueira et al., 2011) or could be due to frequent feeding/fasting cycle which has been suggested as a risk factor for age-associated obesity and insulin resistance (Park et al., 2014). Besides hampered systemic glucose homeostasis, skeletal muscle, one of the major contributing tissues towards systemic glucose disposal, also showed poor glucose handling indicated by lower expression of AS160 in IF group. Although more intensive investigations are required to back up this finding, SIRT1 expression in adipose tissue of IF group was found to be lower in both males and females. Expression of SIRT1 in adipose tissue has recently

been correlated to systemic glucose homeostasis, and lower expression of SIRT1 with poor glucose homeostasis in IF group in the present study, further confirms the previous report by Stefanowicz et al. (2018).

#### 7.1.5 <u>Sexually dimorphic response to IF and/or HIIT</u>

Males and females responded very differently to IF and/or HIIT. Males showed higher changes in body composition in comparison to females. This was also in line with some previous studies where differences between sexes has also been reported in response to lifestyle interventions (Williams et al., 2015, Baillot et al., 2015, Gallagher et al., 2017). This has been attributed to differences in adipose tissue distribution and functionality and also hormonal differences between sexes with males exhibiting higher levels of circulating leptin and females exhibiting higher levels of circulating adiponectin (Valencak et al., 2017). Also at the molecular level, several genes and miRNAs showed different trend in two genders. Especially in case of miRNAs, in females, miR-24, 222, 143 in adipose tissue and in males, muscle specific miR-133a and 133b did not show any significant differences with IF/HIIT. Translating this to humans, care must be taken while recommending these interventions to males and females as both these interventions have to be optimised separately for two sexes.

#### 7.2. LIMITATIONS

A number of limitations exist in this thesis. To determine the time effect of IF and/or HIIT, plasma and tissue samples before and after intervention period from the same animal could not be obtained due to animal ethics limitations. Therefore, plasma insulin, glucose, HOMA-IR, lipid panels, muscle fibre type distribution and expression of mRNA and miRNAs were only measured at the end of the 12-week intervention period. Moreover, due to housing constrains of ethical approval, mice from different intervention groups were housed together, more specifically, mice undergoing fasting (IF and IF + HIIT) were housed together and mice undergoing no fasting (HIIT and CON) were housed together. Thus, food intake of individual mice could not be obtained and the effect of IF or exercise on diet intake per se could not be determined.

Also, though mice were acquired from the same inbred strains, it was noticed that there was variability within the same intervention groups in terms of gain in body weight, body composition, plasma glucose and lipid profiles, expression of genes and miRNAs etc, and thus we can only assume the mice had similar genotype.

#### 7.3. FUTURE DIRECTIONS

There are several questions that arose from the findings of this thesis which will inform future research in this area. Firstly, despite several claims by most previous studies both male and female mice in IF group showed poor glucose homeostasis. This was also reported by very few previous animal studies (Cerqueira et al., 2011). It is still unknown, if it is something peculiar to animal models, diet or is age related disorder with IF. Long term intermittent fasting studies with proper time point controls are required to answer this anomaly.

Secondly, both diet and exercise interventions have been reported to influence structure and function of central nervous system (Pinilla, 2006). It would be interesting to know how IF and/or HIIT influence various functions especially in hypothalamic region of the brain, since this part of the brain is called "control centre" of energy homeostasis (O'Brien et al., 2017). This might also give a clue about insulin resistance and poor glucose handling found in IF mice in this study. Also, how combination of these two intervention affect cognitive functions is still unknown.

177

Thirdly, the caloric intake between fasting and non-fasting groups were found to be similar. This indicates that body compositional changes induced by IF may not just be induced by negative energy balance, as despite consuming same amount of diet, mice in fasting group weighed less. Future studies can be targeted to explore the reason behind this observation.

Moreover, only limited miRNAs can be investigated in the present study and their roles at the level of translational regulation were not determined due to limitations of tissue availability and resources. More robust investigations are required to determine the exact role of signature miRNAs in response to these lifestyle interventions. MiRNAs have been suggested as a potential therapeutic targets (Krützfeldt, 2016) and identification of candidate miRNAs, that can be used as therapeutic targets for treatment of obesity and associated comorbidities would be very novel.

Finally, the findings of the present study can be used as a foundation for clinical research on the combination of these interventions in obese humans. Since, the response to IF and HIIT was different in males and females, separate studies for males and females must be designed and optimised to maximize outcome.

#### REFERENCES

- AGUIRRE, V., WERNER, E. D., GIRAUD, J., LEE, Y. H., SHOELSON, S. E. & WHITE, M. F. 2002. Phosphorylation of Ser307 in Insulin Receptor Substrate-1 Blocks Interactions with the Insulin Receptor and Inhibits Insulin Action. *Journal* of Biological Chemistry, 277, 1531-1537.
- AHN, N. & KIM, K. 2014. Combined influence of dietary restriction and treadmill running on MCP-1 and the expression of oxidative stress-related mRNA in the adipose tissue in obese mice. *J Exerc Nutrition Biochem*, 18, 311-8.
- ALBU, J. B., KOVERA, A. J., ALLEN, L., WAINWRIGHT, M., BERK, E., RAJA-KHAN, N., JANUMALA, I., BURKEY, B., HESHKA, S. & GALLAGHER, D. 2005. Independent association of insulin resistance with larger amounts of intermuscular adipose tissue and a greater acute insulin response to glucose in African American than in white nondiabetic women. Am J Clin Nutr, 82, 1210-7.
- ALTINTAS, M. M., NAYER, B., WALFORD, E. C., JOHNSON, K. B., GAIDOSH, G., REISER, J., DE LA CRUZ-MUNOZ, N., ORTEGA, L. M. & NAYER, A. 2012. Leptin deficiency-induced obesity affects the density of mast cells in abdominal fat depots and lymph nodes in mice. *Lipids Health Dis*, 11, 21.
- AMERICAN COLLEGE OF SPORTS MEDICINE 2013. ACSM's guidelines for exercise testing and prescription, Lippincott Williams & Wilkins.
- ANDRIKOPOULOS, S., BLAIR, A. R., DELUCA, N., FAM, B. C. & PROIETTO, J. 2008. Evaluating the glucose tolerance test in mice. *Am J Physiol Endocrinol Metab*, 295, E1323-32.
- ANTONI, R., JOHNSTON, K. L., COLLINS, A. L. & ROBERTSON, M. D. 2017. Effects of intermittent fasting on glucose and lipid metabolism. *Proc Nutr Soc*, 76, 361-368.
- ANTUNES, F. & HAN, D. 2009. Redox regulation of NF-kappaB: from basic to clinical research. *Antioxid Redox Signal*, 11, 2055-6.
- AOI, W., ICHIKAWA, H., MUNE, K., TANIMURA, Y., MIZUSHIMA, K., NAITO, Y.
   & YOSHIKAWA, T. 2013. Muscle-enriched microRNA miR-486 decreases in circulation in response to exercise in young men. *Frontiers in Physiology*, 1-23.
- AOI, W., NAITO, Y., MIZUSHIMA, K., TAKANAMI, Y., KAWAI, Y., ICHIKAWA, H. & YOSHIKAWA, T. 2010. The microRNA miR-696 regulates PGC-1{alpha} in mouse skeletal muscle in response to physical activity. *Am J Physiol Endocrinol Metab*, 298, E799-806.
- ASSIMACOPOULOS-JEANNET, F., BRICHARD, S., RENCUREL, F., CUSIN, I. & JEANRENAUD, B. 1995. In vivo effects of hyperinsulinemia on lipogenic enzymes and glucose transporter expression in rat liver and adipose tissues. *Metabolism*, 44, 228-33.
- AUSTIN, S. & ST-PIERRE, J. 2012. PGC1alpha and mitochondrial metabolism-emerging concepts and relevance in ageing and neurodegenerative disorders. *J Cell Sci*, 125, 4963-71.
- AUWERX, J., COCK, T.-A. & KNOUFF, C. 2003. PPAR-γ: a thrifty transcription factor. *Nuclear Receptor Signaling*, 1, e006.
- AVRAM, M. M., AVRAM, A. S. & JAMES, W. D. 2007. Subcutaneous fat in normal and diseased states 3. Adipogenesis: from stem cell to fat cell. *J Am Acad Dermatol*, 56, 472-92.

- AZEVEDO, F. R. D., IKEOKA, D. & CARAMELLI, B. 2013. Effects of intermittent fasting on metabolism in men. *Revista da Associação Médica Brasileira*, 59, 167-173.
- BAGGISH, A. L., HALE, A., WEINER, R. B., LEWIS, G. D., SYSTROM, D., WANG, F., WANG, T. J. & CHAN, S. Y. 2011. Dynamic regulation of circulating microRNA during acute exhaustive exercise and sustained aerobic exercise training. J Physiol, 589, 3983-94.
- BAILLOT, A., ROMAIN, A. J., BOISVERT-VIGNEAULT, K., AUDET, M., BAILLARGEON, J. P., DIONNE, I. J., VALIQUETTE, L., CHAKRA, C. N., AVIGNON, A. & LANGLOIS, M. F. 2015. Effects of lifestyle interventions that include a physical activity component in class II and III obese individuals: a systematic review and meta-analysis. *PLoS One*, 10, e0119017.
- BAN JAE-JUN, ROBIN J RUTHENBORG, KEVIN W CHO & KIM, J.-W. 2014. Regulation of obesity and insulin resistance by hypoxia-inducible factors. *Hypoxia*, 2, 171-183.
- BARNOSKY, A. R., HODDY, K. K., UNTERMAN, T. G. & VARADY, K. A. 2014. Intermittent fasting vs daily calorie restriction for type 2 diabetes prevention: a review of human findings. *Transl Res*, 164, 302-311.
- BARTEL, D. P. 2004. MicroRNAs: Genomics, Biogenesis, Mechanism, and Function. *Cell*, 116, 281-297.
- BARTZ, R., LI, W. H., VENABLES, B., ZEHMER, J. K., ROTH, M. R., WELTI, R., ANDERSON, R. G., LIU, P. & CHAPMAN, K. D. 2007. Lipidomics reveals that adiposomes store ether lipids and mediate phospholipid traffic. *J Lipid Res*, 48, 837-47.
- BASKIN, KEDRYN K., WINDERS, BENJAMIN R. & OLSON, ERIC N. 2015. Muscle as a "Mediator" of Systemic Metabolism. *Cell Metabolism*, 21, 237-248.
- BASSEL-DUBY, R. & OLSON, E. N. 2006. Signaling pathways in skeletal muscle remodeling. *Annu Rev Biochem*, 75, 19-37.
- BASYUK, E., SUAVET, F., DOGLIO, A., BORDONNE, R. & BERTRAND, E. 2003. Human let-7 stem-loop precursors harbor features of RNase III cleavage products. *Nucleic Acids Res*, 31, 6593-7.
- BATACAN, R. B., DUNCAN, M. J., DALBO, V. J., TUCKER, P. S. & FENNING, A. S. 2017. Effects of high-intensity interval training on cardiometabolic health: a systematic review and meta-analysis of intervention studies. *British Journal of Sports Medicine*, 51, 494.
- BAUMEIER, C., KAISER, D., HEEREN, J., SCHEJA, L., JOHN, C., WEISE, C., ERAVCI, M., LAGERPUSCH, M., SCHULZE, G., JOOST, H.-G., SCHWENK, R. W. & SCHÜRMANN, A. 2015. Caloric restriction and intermittent fasting alter hepatic lipid droplet proteome and diacylglycerol species and prevent diabetes in NZO mice. *Biochimica et Biophysica Acta (BBA) - Molecular and Cell Biology* of Lipids, 1851, 566-576.
- BAXA, C. A., SHA, R. S., BUELT, M. K., SMITH, A. J., MATARESE, V., CHINANDER, L. L., BOUNDY, K. L. & BERNLOHR, D. A. 1989. Human adipocyte lipid-binding protein: purification of the protein and cloning of its complementary DNA. *Biochemistry*, 28, 8683-90.
- BEIGY, M., VAKILI, S., BERIJANI, S., AMINIZADE, M., AHMADI-DASTGERDI, M. & MESHKANI, R. 2013. Alternate-day fasting diet improves fructoseinduced insulin resistance in mice. J Anim Physiol Anim Nutr (Berl), 97, 1125-31.

- BERGERON, R., REN, J. M., CADMAN, K. S., MOORE, I. K., PERRET, P., PYPAERT, M., YOUNG, L. H., SEMENKOVICH, C. F. & SHULMAN, G. I. 2001. Chronic activation of AMP kinase results in NRF-1 activation and mitochondrial biogenesis. *Am J Physiol Endocrinol Metab*, 281, E1340-6.
- BERGLUND, E. D., VIANNA, C. R., DONATO, J., JR., KIM, M. H., CHUANG, J. C., LEE, C. E., LAUZON, D. A., LIN, P., BRULE, L. J., SCOTT, M. M., COPPARI, R. & ELMQUIST, J. K. 2012. Direct leptin action on POMC neurons regulates glucose homeostasis and hepatic insulin sensitivity in mice. *J Clin Invest*, 122, 1000-9.
- BHUTANI, S., KLEMPEL, M. C., KROEGER, C. M., TREPANOWSKI, J. F. & VARADY, K. A. 2013. Alternate day fasting and endurance exercise combine to reduce body weight and favorably alter plasma lipids in obese humans. *Obesity* (*Silver Spring*), 21, 1370-9.
- BJÖRNHOLM, M., KAWANO, Y., LEHTIHET, M. & ZIERATH, J. R. 1997. Insulin Receptor Substrate-1 Phosphorylation and Phosphatidylinositol 3-Kinase Activity in Skeletal Muscle From NIDDM Subjects After In Vivo Insulin Stimulation. *Diabetes*, 46, 524.
- BJORNTORP, P., KARLSSON, M. & PETTERSSON, P. 1982. Expansion of adipose tissue storage capacity at different ages in rats. *Metabolism*, 31, 366-73.
- BODEN, G., LEBED, B., SCHATZ, M., HOMKO, C. & LEMIEUX, S. 2001. Effects of acute changes of plasma free fatty acids on intramyocellular fat content and insulin resistance in healthy subjects. *Diabetes*, 50, 1612-7.
- BONEN, A., DOHM, G. L. & VAN LOON, L. J. 2006. Lipid metabolism, exercise and insulin action. *Essays Biochem*, 42, 47-59.
- BOON, M. R., BAKKER, L. E., HAKS, M. C., QUINTEN, E., SCHAART, G., VAN BEEK, L., WANG, Y., VAN SCHINKEL, L., VAN HARMELEN, V., MEINDERS, A. E., OTTENHOFF, T. H., VAN DIJK, K. W., GUIGAS, B., JAZET, I. M. & RENSEN, P. C. 2015. Short-term high-fat diet increases macrophage markers in skeletal muscle accompanied by impaired insulin signalling in healthy male subjects. *Clin Sci (Lond)*, 128, 143-51.
- BOOTH, A., MAGNUSON, A., FOUTS, J. & FOSTER MICHELLE, T. 2016. Adipose tissue: an endocrine organ playing a role in metabolic regulation. *Hormone Molecular Biology and Clinical Investigation*.
- BORDONE, L. & GUARENTE, L. 2005. Calorie restriction, SIRT1 and metabolism: understanding longevity. *Nat Rev Mol Cell Biol*, 6, 298-305.
- BORST, S. E. & CONOVER, C. F. 2005. High-fat diet induces increased tissue expression of TNF-α. *Life Sciences*, 77, 2156-2165.
- BOUDOU, P., SOBNGWI, E., MAUVAIS-JARVIS, F., VEXIAU, P. & GAUTIER, J. F. 2003. Absence of exercise-induced variations in adiponectin levels despite decreased abdominal adiposity and improved insulin sensitivity in type 2 diabetic men. *Eur J Endocrinol*, 149, 421-4.
- BOUWMAN, F. G., WANG, P., VAN BAAK, M., SARIS, W. H. & MARIMAN, E. C. 2014. Increased beta-oxidation with improved glucose uptake capacity in adipose tissue from obese after weight loss and maintenance. *Obesity (Silver Spring)*, 22, 819-27.
- BRASAEMLE, D. L. 2007. Thematic review series: adipocyte biology. The perilipin family of structural lipid droplet proteins: stabilization of lipid droplets and control of lipolysis. *J Lipid Res*, 48, 2547-59.

- BREHM, A., KRSSAK, M., SCHMID, A. I., NOWOTNY, P., WALDHAUSL, W. & RODEN, M. 2006. Increased lipid availability impairs insulin-stimulated ATP synthesis in human skeletal muscle. *Diabetes*, 55, 136-40.
- BROWN, J. E., MOSLEY, M. & ALDRED, S. 2013. Intermittent fasting: a dietary intervention for prevention of diabetes and cardiovascular disease? *The British Journal of Diabetes & Vascular Disease*, 13, 68-72.
- BRUCE, C. R., HOY, A. J., TURNER, N., WATT, M. J., ALLEN, T. L., CARPENTER, K., COONEY, G. J., FEBBRAIO, M. A. & KRAEGEN, E. W. 2009. Overexpression of Carnitine Palmitoyltransferase-1 in Skeletal Muscle Is Sufficient to Enhance Fatty Acid Oxidation and Improve High-Fat Diet–Induced Insulin Resistance. *Diabetes*, 58, 550-558.
- BRUUN, J. M., HELGE, J. W., RICHELSEN, B. & STALLKNECHT, B. 2006. Diet and exercise reduce low-grade inflammation and macrophage infiltration in adipose tissue but not in skeletal muscle in severely obese subjects. Am J Physiol Endocrinol Metab, 290, E961-7.
- BRUUN, J. M., LIHN, A. S., VERDICH, C., PEDERSEN, S. B., TOUBRO, S., ASTRUP, A. & RICHELSEN, B. 2003. Regulation of adiponectin by adipose tissue-derived cytokines: in vivo and in vitro investigations in humans. Am J Physiol Endocrinol Metab, 285, E527-33.
- CAMPS, S. G. J. A., VERHOEF, S. P. M., ROUMANS, N., BOUWMAN, F. G., MARIMAN, E. C. M. & WESTERTERP, K. R. 2015. Weight loss-induced changes in adipose tissue proteins associated with fatty acid and glucose metabolism correlate with adaptations in energy expenditure. *Nutrition & Metabolism*, 12, 37.
- CANTO, C. & AUWERX, J. 2009. PGC-1alpha, SIRT1 and AMPK, an energy sensing network that controls energy expenditure. *Curr Opin Lipidol*, 20, 98-105.
- CANTO, C., GERHART-HINES, Z., FEIGE, J. N., LAGOUGE, M., NORIEGA, L., MILNE, J. C., ELLIOTT, P. J., PUIGSERVER, P. & AUWERX, J. 2009. AMPK regulates energy expenditure by modulating NAD+ metabolism and SIRT1 activity. *Nature*, 458, 1056-60.
- CANTO, C., JIANG, L. Q., DESHMUKH, A. S., MATAKI, C., COSTE, A., LAGOUGE, M., ZIERATH, J. R. & AUWERX, J. 2010. Interdependence of AMPK and SIRT1 for metabolic adaptation to fasting and exercise in skeletal muscle. *Cell Metab*, 11, 213-9.
- CAREY, A. L., STEINBERG, G. R., MACAULAY, S. L., THOMAS, W. G., HOLMES, A. G., RAMM, G., PRELOVSEK, O., HOHNEN-BEHRENS, C., WATT, M. J., JAMES, D. E., KEMP, B. E., PEDERSEN, B. K. & FEBBRAIO, M. A. 2006. Interleukin-6 increases insulin-stimulated glucose disposal in humans and glucose uptake and fatty acid oxidation in vitro via AMP-activated protein kinase. *Diabetes*, 55, 2688-97.
- CARLING, D. & HARDIE, D. G. 1989. The substrate and sequence specificity of the AMP-activated protein kinase. Phosphorylation of glycogen synthase and phosphorylase kinase. *Biochim Biophys Acta*, 1012, 81-6.
- CARLING, D., SANDERS, M. J. & WOODS, A. 2008. The regulation of AMP-activated protein kinase by upstream kinases. *Int J Obes (Lond)*, 32 Suppl 4, S55-9.
- CARTEE, G. D. 2015. Mechanisms for greater insulin-stimulated glucose uptake in normal and insulin-resistant skeletal muscle after acute exercise. *American Journal of Physiology Endocrinology and Metabolism*, 309, E949-E959.

- CASSIDY, S., THOMA, C., HALLSWORTH, K., PARIKH, J., HOLLINGSWORTH, K. G., TAYLOR, R., JAKOVLJEVIC, D. G. & TRENELL, M. I. 2016. High intensity intermittent exercise improves cardiac structure and function and reduces liver fat in patients with type 2 diabetes: a randomised controlled trial. *Diabetologia*, 59, 56-66.
- CASSIDY, S., THOMA, C., HOUGHTON, D. & TRENELL, M. I. 2017. High-intensity interval training: a review of its impact on glucose control and cardiometabolic health. *Diabetologia*, 60, 7-23.
- CASTELLANI, L., ROOT-MCCAIG, J., FRENDO-CUMBO, S., BEAUDOIN, M.-S. & WRIGHT, D. C. 2014. Exercise training protects against an acute inflammatory insult in mouse epididymal adipose tissue. *Journal of Applied Physiology*, 116, 1272-1280.
- CASTELLO, L., FROIO, T., MAINA, M., CAVALLINI, G., BIASI, F., LEONARDUZZI, G., DONATI, A., BERGAMINI, E., POLI, G. & CHIARPOTTO, E. 2010. Alternate-day fasting protects the rat heart against ageinduced inflammation and fibrosis by inhibiting oxidative damage and NF-kB activation. *Free Radical Biology and Medicine*, 48, 47-54.
- CATENACCI, V. A., PAN, Z., OSTENDORF, D., BRANNON, S., GOZANSKY, W. S., MATTSON, M. P., MARTIN, B., MACLEAN, P. S., MELANSON, E. L. & TROY DONAHOO, W. 2016. A randomized pilot study comparing zero-calorie alternate-day fasting to daily caloric restriction in adults with obesity. *Obesity (Silver Spring)*, 24, 1874-83.
- CDC 2008. http://www.cdc.gov/nchs/data/hus/hus08.pdf.
- CERQUEIRA, F. M., DA CUNHA, F. M., CALDEIRA DA SILVA, C. C., CHAUSSE, B., ROMANO, R. L., GARCIA, C. C., COLEPICOLO, P., MEDEIROS, M. H. & KOWALTOWSKI, A. J. 2011. Long-term intermittent feeding, but not caloric restriction, leads to redox imbalance, insulin receptor nitration, and glucose intolerance. *Free Radic Biol Med*, 51, 1454-60.
- CHABI, B., ADHIHETTY, P. J., O'LEARY, M. F., MENZIES, K. J. & HOOD, D. A. 2009. Relationship between Sirt1 expression and mitochondrial proteins during conditions of chronic muscle use and disuse. *J Appl Physiol (1985)*, 107, 1730-5.
- CHALKIADAKI, A. & GUARENTE, L. 2012. High-fat diet triggers inflammationinduced cleavage of SIRT1 in adipose tissue to promote metabolic dysfunction. *Cell Metabolism*, 16, 180-8.
- CHAN, M. C. & ARANY, Z. 2014. The many roles of PGC-1alpha in muscle--recent developments. *Metabolism*, 63, 441-51.
- CHARTOUMPEKIS, D. V., ZIROS, P. G., ISKRENOVA, R. P., PSYROGIANNIS, A. I., KYRIAZOPOULOU, V. E., HABEOS, I. G. & ZARAVINOS, A. 2012. Differential expression of microRNAs in adipose tissue after long-term high-fat diet-induced obesity in mice. *PLoS ONE*, 7, e34872.
- CHAWLA, A., NGUYEN, K. D. & GOH, Y. P. 2011. Macrophage-mediated inflammation in metabolic disease. *Nat Rev Immunol*, 11, 738-49.
- CHEATHAM, B., VLAHOS, C. J., CHEATHAM, L., WANG, L., BLENIS, J. & KAHN, C. R. 1994. Phosphatidylinositol 3-kinase activation is required for insulin stimulation of pp70 S6 kinase, DNA synthesis, and glucose transporter translocation. *Mol Cell Biol*, 14, 4902-11.

- CHEN, D., BRUNO, J., EASLON, E., LIN, S.-J., CHENG, H.-L., ALT, F. W. & GUARENTE, L. 2008a. Tissue-specific regulation of SIRT1 by calorie restriction. *Genes & Development*, 22, 1753-1757.
- CHEN, D., BRUNO, J., EASLON, E., LIN, S. J., CHENG, H. L., ALT, F. W. & GUARENTE, L. 2008b. Tissue-specific regulation of SIRT1 by calorie restriction. *Genes Dev*, 22, 1753-7.
- CHEN, G. Q., LIAN, W. J., WANG, G. M., WANG, S., YANG, Y. Q. & ZHAO, Z. W. 2012. Altered microRNA expression in skeletal muscle results from high-fat dietinduced insulin resistance in mice. *Mol Med Rep*, 5, 1362-8.
- CHEN, N., CHENG, J., ZHOU, L., LEI, T., CHEN, L., SHEN, Q., QIN, L. & WAN, Z. 2015. Effects of treadmill running and rutin on lipolytic signaling pathways and TRPV4 protein expression in the adipose tissue of diet-induced obese mice. *Journal of Physiology and Biochemistry*, 71, 733-742.
- CHENDRIMADA, T. P., FINN, K. J., JI, X., BAILLAT, D., GREGORY, R. I., LIEBHABER, S. A., PASQUINELLI, A. E. & SHIEKHATTAR, R. 2007. MicroRNA silencing through RISC recruitment of eIF6. *Nature*, 447, 823-8.
- CHO, J., KIM, S., LEE, S. & KANG, H. 2015. Effect of Training Intensity on Nonalcoholic Fatty Liver Disease. *Med Sci Sports Exerc*, 47, 1624-34.
- CHOE, S. S., HUH, J. Y., HWANG, I. J., KIM, J. I. & KIM, J. B. 2016. Adipose Tissue Remodeling: Its Role in Energy Metabolism and Metabolic Disorders. *Frontiers in Endocrinology*, 7, 30.
- CHOI, M.-S., KIM, Y.-J., KWON, E.-Y., RYOO, J. Y., KIM, S. R. & JUNG, U. J. 2015. High-fat diet decreases energy expenditure and expression of genes controlling lipid metabolism, mitochondrial function and skeletal system development in the adipose tissue, along with increased expression of extracellular matrix remodelling- and inflammation-related genes. *British Journal of Nutrition*, 113, 867-877.
- CHOI, S. W. & FRISO, S. 2010. Epigenetics: A New Bridge between Nutrition and Health. *Adv Nutr*, 1, 8-16.
- CIARALDI, T. P., RYAN, A. J., MUDALIAR, S. R. & HENRY, R. R. 2016. Altered Myokine Secretion Is an Intrinsic Property of Skeletal Muscle in Type 2 Diabetes. *PLoS One*, 11, e0158209.
- CIVITARESE, A. E., CARLING, S., HEILBRONN, L. K., HULVER, M. H., UKROPCOVA, B., DEUTSCH, W. A., SMITH, S. R. & RAVUSSIN, E. 2007. Calorie Restriction Increases Muscle Mitochondrial Biogenesis in Healthy Humans. *PLoS Medicine*, 4, e76.
- CLARK, I. A. 2007. How TNF was recognized as a key mechanism of disease. *Cytokine Growth Factor Rev*, 18, 335-43.
- COELHO, M., OLIVEIRA, T. & FERNANDES, R. 2013. Biochemistry of adipose tissue: an endocrine organ. *Archives of Medical Science : AMS*, 9, 191-200.
- COHEN, H. Y., MILLER, C., BITTERMAN, K. J., WALL, N. R., HEKKING, B., KESSLER, B., HOWITZ, K. T., GOROSPE, M., DE CABO, R. & SINCLAIR, D. A. 2004. Calorie restriction promotes mammalian cell survival by inducing the SIRT1 deacetylase. *Science*, 305, 390-2.
- COLLINS, K. H., PAUL, H. A., HART, D. A., REIMER, R. A., SMITH, I. C., RIOS, J. L., SEERATTAN, R. A. & HERZOG, W. 2016. A High-Fat High-Sucrose Diet Rapidly Alters Muscle Integrity, Inflammation and Gut Microbiota in Male Rats. *Scientific Reports*, 6, 37278.

- COLLINS, T., READ, M. A., NEISH, A. S., WHITLEY, M. Z., THANOS, D. & MANIATIS, T. 1995. Transcriptional regulation of endothelial cell adhesion molecules: NF-kappa B and cytokine-inducible enhancers. *Faseb j*, 9, 899-909.
- CONSIDINE, R. V., SINHA, M. K., HEIMAN, M. L., KRIAUCIUNAS, A., STEPHENS, T. W., NYCE, M. R., OHANNESIAN, J. P., MARCO, C. C., MCKEE, L. J., BAUER, T. L. & ET AL. 1996. Serum immunoreactive-leptin concentrations in normal-weight and obese humans. *N Engl J Med*, 334, 292-5.
- CONSITT, L. A., VAN METER, J., NEWTON, C. A., COLLIER, D. N., DAR, M. S., WOJTASZEWSKI, J. F. P., TREEBAK, J. T., TANNER, C. J. & HOUMARD, J. A. 2013. Impairments in Site-Specific AS160 Phosphorylation and Effects of Exercise Training. *Diabetes*, 62, 3437.
- COOKE, J. P. & OKA, R. K. 2002. Does leptin cause vascular disease? *Circulation*, 106, 1904-5.
- COPPARI, R. & BJORBAEK, C. 2012. Leptin revisited: its mechanism of action and potential for treating diabetes. *Nat Rev Drug Discov*, 11, 692-708.
- CORPELEIJN, E., SARIS, W. H., JANSEN, E. H., ROEKAERTS, P. M., FESKENS, E. J. & BLAAK, E. E. 2005. Postprandial interleukin-6 release from skeletal muscle in men with impaired glucose tolerance can be reduced by weight loss. *J Clin Endocrinol Metab*, 90, 5819-24.
- COUTURIER, A., RINGSEIS, R., MOOREN, F. C., KRUGER, K., MOST, E. & EDER, K. 2013. Carnitine supplementation to obese Zucker rats prevents obesity-induced type II to type I muscle fiber transition and favors an oxidative phenotype of skeletal muscle. *Nutr Metab (Lond)*, 10, 48.
- CUI, J., ZHOU, B., ROSS, S. A. & ZEMPLENI, J. 2017. Nutrition, microRNAs, and Human Health. *Advances in Nutrition*, 8, 105-112.
- DAHLMAN, I., KAAMAN, M., JIAO, H., KERE, J., LAAKSO, M. & ARNER, P. 2005. The CIDEA gene V115F polymorphism is associated with obesity in Swedish subjects. *Diabetes*, 54, 3032-4.
- DAUSSIN, F. N., ZOLL, J., DUFOUR, S. P., PONSOT, E., LONSDORFER-WOLF, E., DOUTRELEAU, S., METTAUER, B., PIQUARD, F., GENY, B. & RICHARD, R. 2008. Effect of interval versus continuous training on cardiorespiratory and mitochondrial functions: relationship to aerobic performance improvements in sedentary subjects. *Am J Physiol Regul Integr Comp Physiol*, 295, R264-72.
- DAVIDSEN, P. K., GALLAGHER, I. J., HARTMAN, J. W., TARNOPOLSKY, M. A., DELA, F., HELGE, J. W., TIMMONS, J. A. & PHILLIPS, S. M. 2011. High responders to resistance exercise training demonstrate differential regulation of skeletal muscle microRNA expression. J Appl Physiol (1985), 110, 309-17.
- DAVIS, R. A. H., HALBROOKS, J. E., WATKINS, E. E., FISHER, G., HUNTER, G. R., NAGY, T. R. & PLAISANCE, E. P. 2017. High-intensity interval training and calorie restriction promote remodeling of glucose and lipid metabolism in diet-induced obesity. *Am J Physiol Endocrinol Metab*, 313, E243-e256.
- DAVY, B. & JAHREN, H. 2016. New markers of dietary added sugar intake. *Curr Opin Clin Nutr Metab Care*, 19, 282-8.
- DE ALVARO, C., TERUEL, T., HERNANDEZ, R. & LORENZO, M. 2004. Tumor necrosis factor alpha produces insulin resistance in skeletal muscle by activation of inhibitor kappaB kinase in a p38 MAPK-dependent manner. *J Biol Chem*, 279, 17070-8.

- DE FERRANTI, S. & MOZAFFARIAN, D. 2008. The perfect storm: obesity, adipocyte dysfunction, and metabolic consequences. *Clin Chem*, 54, 945-55.
- DEACON, R. M. J. 2013. Measuring the Strength of Mice. Journal of Visualized Experiments : JoVE, 2610.
- DEIULIIS, J. A. 2016. MicroRNAs as regulators of metabolic disease: pathophysiologic significance and emerging role as biomarkers and therapeutics. *Int J Obes*, 40, 88-101.
- DENHAM, J., MARQUES, F. Z., O'BRIEN, B. J. & CHARCHAR, F. J. 2014. Exercise: putting action into our epigenome. *Sports Med*, 44, 189-209.
- DESHMUKH, A., COFFEY, V. G., ZHONG, Z., CHIBALIN, A. V., HAWLEY, J. A. & ZIERATH, J. R. 2006. Exercise-Induced Phosphorylation of the Novel Akt Substrates AS160 and Filamin A in Human Skeletal Muscle. *Diabetes*, 55, 1776.
- DI MEO, S., IOSSA, S. & VENDITTI, P. 2017. Skeletal muscle insulin resistance: role of mitochondria and other ROS sources. *J Endocrinol*, 233, R15-r42.
- DING, H., ZHENG, S., GARCIA-RUIZ, D., HOU, D., WEI, Z., LIAO, Z., LI, L., ZHANG, Y., HAN, X., ZEN, K., ZHANG, C.-Y., LI, J. & JIANG, X. 2016. Fasting induces a subcutaneous-to-visceral fat switch mediated by microRNA-149-3p and suppression of PRDM16. *Nature Communications*, 7, 11533.
- DISANZO, B. L. & YOU, T. 2014. Effects of exercise training on indicators of adipose tissue angiogenesis and hypoxia in obese rats. *Metabolism*, 63, 452-455.
- DRUMMOND, M. J., MCCARTHY, J. J., FRY, C. S., ESSER, K. A. & RASMUSSEN, B. B. 2008. Aging differentially affects human skeletal muscle microRNA expression at rest and after an anabolic stimulus of resistance exercise and essential amino acids. *Am J Physiol Endocrinol Metab*, 295, E1333-40.
- DUBE, J. J., AMATI, F., STEFANOVIC-RACIC, M., TOLEDO, F. G., SAUERS, S. E. & GOODPASTER, B. H. 2008. Exercise-induced alterations in intramyocellular lipids and insulin resistance: the athlete's paradox revisited. *Am J Physiol Endocrinol Metab*, 294, E882-8.
- DUMORTIER, O., HINAULT, C. & VAN OBBERGHEN, E. 2013. MicroRNAs and Metabolism Crosstalk in Energy Homeostasis. *Cell Metabolism*, 18, 312-324.
- DUNCAN, R. E., AHMADIAN, M., JAWORSKI, K., SARKADI-NAGY, E. & SUL, H. S. 2007. Regulation of Lipolysis in Adipocytes. *Annual review of nutrition*, 27, 79-101.
- DUTTA, S. & SENGUPTA, P. 2016. Men and mice: Relating their ages. *Life Sci*, 152, 244-8.
- DWEEP, H., STICHT, C., PANDEY, P. & GRETZ, N. 2011. miRWalk Database: Prediction of possible miRNA binding sites by "walking" the genes of three genomes. *Journal of Biomedical Informatics*, 44, 839-847.
- EARNEST, C. P., LUPO, M., THIBODAUX, J., HOLLIER, C., BUTITTA, B., LEJEUNE, E., JOHANNSEN, N. M., GIBALA, M. J. & CHURCH, T. S. 2013. Interval training in men at risk for insulin resistance. *Int J Sports Med*, 34, 355-63.
- ECKARDT, K., GORGENS, S. W., RASCHKE, S. & ECKEL, J. 2014. Myokines in insulin resistance and type 2 diabetes. *Diabetologia*, 57, 1087-99.
- EHLERT, T., SIMON, P. & MOSER, D. A. 2013. Epigenetics in sports. Sports Med, 43, 93-110.
- EISSING, L., SCHERER, T., TODTER, K., KNIPPSCHILD, U., GREVE, J. W., BUURMAN, W. A., PINNSCHMIDT, H. O., RENSEN, S. S., WOLF, A. M.,

BARTELT, A., HEEREN, J., BUETTNER, C. & SCHEJA, L. 2013. De novo lipogenesis in human fat and liver is linked to ChREBP-beta and metabolic health. *Nat Commun*, 4, 1528.

- ELBASHIR, S. M., LENDECKEL, W. & TUSCHL, T. 2001. RNA interference is mediated by 21- and 22-nucleotide RNAs. *Genes Dev*, 15, 188-200.
- ELIA, L., CONTU, R., QUINTAVALLE, M., VARRONE, F., CHIMENTI, C., RUSSO, M. A., CIMINO, V., DE MARINIS, L., FRUSTACI, A., CATALUCCI, D. & CONDORELLI, G. 2009. Reciprocal regulation of microRNA-1 and IGF-1 signal transduction cascade in cardiac and skeletal muscle in physiological and pathological conditions. *Circulation*, 120, 2377-2385.
- ESAU, C., DAVIS, S., MURRAY, S. F., YU, X. X., PANDEY, S. K., PEAR, M., WATTS, L., BOOTEN, S. L., GRAHAM, M., MCKAY, R., SUBRAMANIAM, A., PROPP, S., LOLLO, B. A., FREIER, S., BENNETT, C. F., BHANOT, S. & MONIA, B. P. 2006. Article: miR-122 regulation of lipid metabolism revealed by in vivo antisense targeting. *Cell Metabolism*, 3, 87-98.
- ESHGHINIA, S. & MOHAMMADZADEH, F. 2013. The effects of modified alternateday fasting diet on weight loss and CAD risk factors in overweight and obese women. *J Diabetes Metab Disord*, 12, 4.
- ESSER, N., L'HOMME, L., DE ROOVER, A., KOHNEN, L., SCHEEN, A. J., MOUTSCHEN, M., PIETTE, J., LEGRAND-POELS, S. & PAQUOT, N. 2013. Obesity phenotype is related to NLRP3 inflammasome activity and immunological profile of visceral adipose tissue. *Diabetologia*, 56, 2487-97.
- ESSER, N., LEGRAND-POELS, S., PIETTE, J., SCHEEN, A. J. & PAQUOT, N. 2014. Inflammation as a link between obesity, metabolic syndrome and type 2 diabetes. *Diabetes Research and Clinical Practice*, 105, 141-150.
- FAIN, J. N., MADAN, A. K., HILER, M. L., CHEEMA, P. & BAHOUTH, S. W. 2004. Comparison of the release of adipokines by adipose tissue, adipose tissue matrix, and adipocytes from visceral and subcutaneous abdominal adipose tissues of obese humans. *Endocrinology*, 145, 2273-82.
- FEI, W., SHUI, G., ZHANG, Y., KRAHMER, N., FERGUSON, C., KAPTERIAN, T. S., LIN, R. C., DAWES, I. W., BROWN, A. J., LI, P., HUANG, X., PARTON, R. G., WENK, M. R., WALTHER, T. C. & YANG, H. 2011. A role for phosphatidic acid in the formation of "supersized" lipid droplets. *PLoS Genet*, 7, e1002201.
- FELLI, N., FONTANA, L., PELOSI, E., BOTTA, R., BONCI, D., FACCHIANO, F., LIUZZI, F., LULLI, V., MORSILLI, O., SANTORO, S., VALTIERI, M., CALIN, G. A., LIU, C. G., SORRENTINO, A., CROCE, C. M. & PESCHLE, C. 2005. MicroRNAs 221 and 222 inhibit normal erythropoiesis and erythroleukemic cell growth via kit receptor down-modulation. *Proc Natl Acad Sci U S A*, 102, 18081-6.
- FERRARA, N., RINALDI, B., CORBI, G., CONTI, V., STIUSO, P., BOCCUTI, S., RENGO, G., ROSSI, F. & FILIPPELLI, A. 2008. Exercise training promotes SIRT1 activity in aged rats. *Rejuvenation Res*, 11, 139-50.
- FERRARO, E., GIAMMARIOLI, A. M., CHIANDOTTO, S., SPOLETINI, I. & ROSANO, G. 2014. Exercise-Induced Skeletal Muscle Remodeling and Metabolic Adaptation: Redox Signaling and Role of Autophagy. *Antioxidants & Redox Signaling*, 21, 154-176.

- FERRE, P. & FOUFELLE, F. 2007. SREBP-1c transcription factor and lipid homeostasis: clinical perspective. *Horm Res*, 68, 72-82.
- FEX, A., LEDUC-GAUDET, J. P., FILION, M. E., KARELIS, A. D. & AUBERTIN-LEHEUDRE, M. 2015. Effect of Elliptical High Intensity Interval Training on Metabolic Risk Factor in Pre- and Type 2 Diabetes Patients: A Pilot Study. J Phys Act Health, 12, 942-6.
- FINK, L. N., COSTFORD, S. R., LEE, Y. S., JENSEN, T. E., BILAN, P. J., OBERBACH, A., BLUHER, M., OLEFSKY, J. M., SAMS, A. & KLIP, A. 2014. Pro-inflammatory macrophages increase in skeletal muscle of high fat-fed mice and correlate with metabolic risk markers in humans. *Obesity (Silver Spring)*, 22, 747-57.
- FINLEY, L. W., LEE, J., SOUZA, A., DESQUIRET-DUMAS, V., BULLOCK, K., ROWE, G. C., PROCACCIO, V., CLISH, C. B., ARANY, Z. & HAIGIS, M. C. 2012. Skeletal muscle transcriptional coactivator PGC-1alpha mediates mitochondrial, but not metabolic, changes during calorie restriction. *Proc Natl Acad Sci U S A*, 109, 2931-6.
- FISCHER, C. P. 2006. Interleukin-6 in acute exercise and training: what is the biological relevance? *Exerc Immunol Rev*, 12, 6-33.
- FISHER, G., BROWN, A. W., BOHAN BROWN, M. M., ALCORN, A., NOLES, C., WINWOOD, L., RESUEHR, H., GEORGE, B., JEANSONNE, M. M. & ALLISON, D. B. 2015. High Intensity Interval- vs Moderate Intensity- Training for Improving Cardiometabolic Health in Overweight or Obese Males: A Randomized Controlled Trial. *PLOS ONE*, 10, e0138853.
- FLOWERS, E., FROELICHER, E. S. & AOUIZERAT, B. E. 2013. MicroRNA regulation of lipid metabolism. *Metabolism*, 62, 12-20.
- FLOWERS, E., WON, G. Y. & FUKUOKA, Y. 2015. MicroRNAs associated with exercise and diet: a systematic review. *Physiol Genomics*, 47, 1-11.
- FOSTER-SCHUBERT, K. E., ALFANO, C. M., DUGGAN, C. R., XIAO, L., CAMPBELL, K. L., KONG, A., BAIN, C. E., WANG, C. Y., BLACKBURN, G. L. & MCTIERNAN, A. 2012. Effect of diet and exercise, alone or combined, on weight and body composition in overweight-to-obese postmenopausal women. *Obesity (Silver Spring)*, 20, 1628-38.
- FRAYN, K. N. 2002. Adipose tissue as a buffer for daily lipid flux. *Diabetologia*, 45, 1201-10.
- FRIAS FDE, T., DE MENDONCA, M., MARTINS, A. R., GINDRO, A. F., COGLIATI, B., CURI, R. & RODRIGUES, A. C. 2016. MyomiRs as Markers of Insulin Resistance and Decreased Myogenesis in Skeletal Muscle of Diet-Induced Obese Mice. *Front Endocrinol (Lausanne)*, 7, 76.
- FROST, R. J. A. & OLSON, E. N. 2011. Control of glucose homeostasis and insulin sensitivity by the Let-7 family of microRNAs. *Proceedings of the National Academy of Sciences*, 108, 21075-21080.
- FUJISAKA, S., USUI, I., IKUTANI, M., AMINUDDIN, A., TAKIKAWA, A., TSUNEYAMA, K., MAHMOOD, A., GODA, N., NAGAI, Y., TAKATSU, K. & TOBE, K. 2013. Adipose tissue hypoxia induces inflammatory M1 polarity of macrophages in an HIF-1alpha-dependent and HIF-1alpha-independent manner in obese mice. *Diabetologia*, 56, 1403-12.
- FURUHASHI, M., SAITOH, S., SHIMAMOTO, K. & MIURA, T. 2014. Fatty Acid-Binding Protein 4 (FABP4): Pathophysiological Insights and Potent Clinical
Biomarker of Metabolic and Cardiovascular Diseases. *Clinical Medicine Insights. Cardiology*, 8, 23-33.

- FYFE, J. J., BISHOP, D. J., ZACHAREWICZ, E., RUSSELL, A. P. & STEPTO, N. K. 2016. Concurrent exercise incorporating high-intensity interval or continuous training modulates mTORC1 signaling and microRNA expression in human skeletal muscle. *Am J Physiol Regul Integr Comp Physiol*, 310, R1297-311.
- GAIDHU, M. P., ANTHONY, N. M., PATEL, P., HAWKE, T. J. & CEDDIA, R. B. 2010. Dysregulation of lipolysis and lipid metabolism in visceral and subcutaneous adipocytes by high-fat diet: role of ATGL, HSL, and AMPK. Am J Physiol Cell Physiol, 298, C961-71.
- GALLAGHER, D., KELLEY, D. E., THORNTON, J., BOXT, L., PI-SUNYER, X., LIPKIN, E., NYENWE, E., JANUMALA, I. & HESHKA, S. 2017. Changes in skeletal muscle and organ size after a weight-loss intervention in overweight and obese type 2 diabetic patients. *Am J Clin Nutr*, 105, 78-84.
- GALLAGHER, I. J., SCHEELE, C., KELLER, P., NIELSEN, A. R., REMENYI, J., FISCHER, C. P., RODER, K., BABRAJ, J., WAHLESTEDT, C., HUTVAGNER, G., PEDERSEN, B. K. & TIMMONS, J. A. 2010. Integration of microRNA changes in vivo identifies novel molecular features of muscle insulin resistance in type 2 diabetes. *Genome Med*, 2, 9.
- GAO, Z., HWANG, D., BATAILLE, F., LEFEVRE, M., YORK, D., QUON, M. J. & YE, J. 2002. Serine Phosphorylation of Insulin Receptor Substrate 1 by Inhibitor κB Kinase Complex. *Journal of Biological Chemistry*, 277, 48115-48121.
- GASTALDELLI, A. 2011. Role of beta-cell dysfunction, ectopic fat accumulation and insulin resistance in the pathogenesis of type 2 diabetes mellitus. *Diabetes Res Clin Pract*, 93 Suppl 1, S60-5.
- GASTALDI, G., RUSSELL, A., GOLAY, A., GIACOBINO, J. P., HABICHT, F., BARTHASSAT, V., MUZZIN, P. & BOBBIONI-HARSCH, E. 2007. Upregulation of peroxisome proliferator-activated receptor gamma coactivator gene (PGC1A) during weight loss is related to insulin sensitivity but not to energy expenditure. *Diabetologia*, 50, 2348-2355.
- GEHART, H., KUMPF, S., ITTNER, A. & RICCI, R. 2010. MAPK signalling in cellular metabolism: stress or wellness? *EMBO Reports*, 11, 834-840.
- GERBER, T., BORG, M. L., HAYES, A. & STATHIS, C. G. 2014. High-intensity intermittent cycling increases purine loss compared with workload-matched continuous moderate intensity cycling. *European Journal of Applied Physiology*, 114, 1513-1520.
- GERTOW, K., PIETILÄINEN, K. H., YKI-JÄRVINEN, H., KAPRIO, J., RISSANEN, A., ERIKSSON, P., HAMSTEN, A. & FISHER, R. M. 2004. Expression of fattyacid-handling proteins in human adipose tissue in relation to obesity and insulin resistance. *Diabetologia*, 47, 1118-1125.
- GHAFARI HOMADINI, S., ASAD, M. R., BAZGIR, B. & RAHIMI, M. 2017. Effects of High Intensity Interval Training and Moderate-Intensity Continuous Training on VEGF Gene Expression in Visceral and Subcutaneous Adipose Tissues of Male Wistar Rats. *IJEM*, 19, 170-176.
- GIBALA, M. J., LITTLE, J. P., MACDONALD, M. J. & HAWLEY, J. A. 2012. Physiological adaptations to low-volume, high-intensity interval training in health and disease. *J Physiol*, 590, 1077-84.

- GIBALA, M. J., MCGEE, S. L., GARNHAM, A. P., HOWLETT, K. F., SNOW, R. J. & HARGREAVES, M. 2009. Brief intense interval exercise activates AMPK and p38 MAPK signaling and increases the expression of PGC-1alpha in human skeletal muscle. *J Appl Physiol (1985)*, 106, 929-34.
- GILLEN, J. B., MARTIN, B. J., MACINNIS, M. J., SKELLY, L. E., TARNOPOLSKY, M. A. & GIBALA, M. J. 2016. Twelve Weeks of Sprint Interval Training Improves Indices of Cardiometabolic Health Similar to Traditional Endurance Training despite a Five-Fold Lower Exercise Volume and Time Commitment. *PLOS ONE*, 11, e0154075.
- GILLEN, J. B., PERCIVAL, M. E., LUDZKI, A., TARNOPOLSKY, M. A. & GIBALA, M. J. 2013. Interval training in the fed or fasted state improves body composition and muscle oxidative capacity in overweight women. *Obesity (Silver Spring)*, 21, 2249-55.
- GIRALDEZ, A. J., MISHIMA, Y., RIHEL, J., GROCOCK, R. J., VAN DONGEN, S., INOUE, K., ENRIGHT, A. J. & SCHIER, A. F. 2006. Zebrafish MiR-430 promotes deadenylation and clearance of maternal mRNAs. *Science*, 312, 75-9.
- GOODPASTER, B. H. & KELLEY, D. E. 2002. Skeletal muscle triglyceride: marker or mediator of obesity-induced insulin resistance in type 2 diabetes mellitus? *Curr Diab Rep*, 2, 216-22.
- GOODPASTER, B. H., THAETE, F. L. & KELLEY, D. E. 2000. Thigh adipose tissue distribution is associated with insulin resistance in obesity and in type 2 diabetes mellitus. *Am J Clin Nutr*, 71, 885-92.
- GOODYEAR, L. J., GIORGINO, F., SHERMAN, L. A., CAREY, J., SMITH, R. J. & DOHM, G. L. 1995. Insulin receptor phosphorylation, insulin receptor substrate-1 phosphorylation, and phosphatidylinositol 3-kinase activity are decreased in intact skeletal muscle strips from obese subjects. *Journal of Clinical Investigation*, 95, 2195-2204.
- GOTTHARDT, J. D., VERPEUT, J. L., YEOMANS, B. L., YANG, J. A., YASREBI, A., ROEPKE, T. A. & BELLO, N. T. 2016. Intermittent Fasting Promotes Fat Loss With Lean Mass Retention, Increased Hypothalamic Norepinephrine Content, and Increased Neuropeptide Y Gene Expression in Diet-Induced Obese Male Mice. *Endocrinology*, 157, 679-91.
- GREEN, C. J., PEDERSEN, M., PEDERSEN, B. K. & SCHEELE, C. 2011. Elevated NF-kappaB activation is conserved in human myocytes cultured from obese type 2 diabetic patients and attenuated by AMP-activated protein kinase. *Diabetes*, 60, 2810-9.
- GREIWE, J. S., CHENG, B., RUBIN, D. C., YARASHESKI, K. E. & SEMENKOVICH, C. F. 2001. Resistance exercise decreases skeletal muscle tumor necrosis factor alpha in frail elderly humans. *Faseb j*, 15, 475-82.
- GUO, Y., WALTHER, T. C., RAO, M., STUURMAN, N., GOSHIMA, G., TERAYAMA, K., WONG, J. S., VALE, R. D., WALTER, P. & FARESE, R. V. 2008. Functional genomic screen reveals genes involved in lipid-droplet formation and utilization. *Nature*, 453, 657-61.
- GURD, B. J., PERRY, C. G., HEIGENHAUSER, G. J., SPRIET, L. L. & BONEN, A. 2010. High-intensity interval training increases SIRT1 activity in human skeletal muscle. *Appl Physiol Nutr Metab*, 35, 350-7.
- HAAM, J. H., KIM, Y. S., KOO, H. S., HAAM, J., SEO, N. K., KIM, H. Y., PARK, K. C., PARK, K. S. & KIM, M. J. 2016. Intermuscular adipose tissue is associated

with monocyte chemoattractant protein-1, independent of visceral adipose tissue. *Clin Biochem*, 49, 439-443.

- HAEMMERLE, G., LASS, A., ZIMMERMANN, R., GORKIEWICZ, G., MEYER, C., ROZMAN, J., HELDMAIER, G., MAIER, R., THEUSSL, C., EDER, S., KRATKY, D., WAGNER, E. F., KLINGENSPOR, M., HOEFLER, G. & ZECHNER, R. 2006. Defective lipolysis and altered energy metabolism in mice lacking adipose triglyceride lipase. *Science*, 312, 734-7.
- HAJER, G. R., VAN HAEFTEN, T. W. & VISSEREN, F. L. 2008. Adipose tissue dysfunction in obesity, diabetes, and vascular diseases. *Eur Heart J*, 29, 2959-71.
- HALBERG, N., KHAN, T., TRUJILLO, M. E., WERNSTEDT-ASTERHOLM, I., ATTIE, A. D., SHERWANI, S., WANG, Z. V., LANDSKRONER-EIGER, S., DINEEN, S., MAGALANG, U. J., BREKKEN, R. A. & SCHERER, P. E. 2009. Hypoxia-inducible factor 1alpha induces fibrosis and insulin resistance in white adipose tissue. *Mol Cell Biol*, 29, 4467-83.
- HANDSCHIN, C., CHIN, S., LI, P., LIU, F., MARATOS-FLIER, E., LEBRASSEUR, N. K., YAN, Z. & SPIEGELMAN, B. M. 2007. Skeletal muscle fiber-type switching, exercise intolerance, and myopathy in PGC-1alpha muscle-specific knock-out animals. *J Biol Chem*, 282, 30014-21.
- HARA, K., KUBOTA, N., TOBE, K., TERAUCHI, Y., MIKI, H., KOMEDA, K., TAMEMOTO, H., YAMAUCHI, T., HAGURA, R., ITO, C., AKANUMA, Y. & KADOWAKI, T. 2000. The role of PPARgamma as a thrifty gene both in mice and humans. *Br J Nutr*, 84 Suppl 2, S235-9.
- HARDIE, D. G. 2011. AMP-activated protein kinase: an energy sensor that regulates all aspects of cell function. *Genes Dev*, 25, 1895-908.
- HARVIE, M. & HOWELL, A. 2017. Potential Benefits and Harms of Intermittent Energy Restriction and Intermittent Fasting Amongst Obese, Overweight and Normal Weight Subjects—A Narrative Review of Human and Animal Evidence. Behavioral Sciences, 7, 4.
- HARVIE, M. N., PEGINGTON, M., MATTSON, M. P., FRYSTYK, J., DILLON, B., EVANS, G., CUZICK, J., JEBB, S. A., MARTIN, B., CUTLER, R. G., SON, T. G., MAUDSLEY, S., CARLSON, O. D., EGAN, J. M., FLYVBJERG, A. & HOWELL, A. 2011. The effects of intermittent or continuous energy restriction on weight loss and metabolic disease risk markers: a randomized trial in young overweight women. *International Journal Of Obesity* (2005), 35, 714-727.
- HAUS, J. M., KASHYAP, S. R., KASUMOV, T., ZHANG, R., KELLY, K. R., DEFRONZO, R. A. & KIRWAN, J. P. 2009. Plasma ceramides are elevated in obese subjects with type 2 diabetes and correlate with the severity of insulin resistance. *Diabetes*, 58, 337-43.
- HAYASHIDA, S., ARIMOTO, A., KURAMOTO, Y., KOZAKO, T., HONDA, S., SHIMENO, H. & SOEDA, S. 2010. Fasting promotes the expression of SIRT1, an NAD+ -dependent protein deacetylase, via activation of PPARalpha in mice. *Mol Cell Biochem*, 339, 285-92.
- HE, Q., GAO, Z., YIN, J., ZHANG, J., YUN, Z. & YE, J. 2011. Regulation of HIF-1{alpha} activity in adipose tissue by obesity-associated factors: adipogenesis, insulin, and hypoxia. *Am J Physiol Endocrinol Metab*, 300, E877-85.
- HEALY, G. N., MATTHEWS, C. E., DUNSTAN, D. W., WINKLER, E. A. & OWEN, N. 2011. Sedentary time and cardio-metabolic biomarkers in US adults: NHANES 2003-06. *Eur Heart J*, 32, 590-7.

- HEILBRONN, L. K., CIVITARESE, A. E., BOGACKA, I., SMITH, S. R., HULVER, M. & RAVUSSIN, E. 2005a. Glucose tolerance and skeletal muscle gene expression in response to alternate day fasting. *Obes Res*, 13, 574-81.
- HEILBRONN, L. K., SMITH, S. R., MARTIN, C. K., ANTON, S. D. & RAVUSSIN, E. 2005b. Alternate-day fasting in nonobese subjects: effects on body weight, body composition, and energy metabolism. *Am J Clin Nutr*, 81, 69-73.
- HENEGHAN, H. M., MILLER, N. & KERIN, M. J. 2010. Role of microRNAs in obesity and the metabolic syndrome. *Obes Rev*, 11, 354-61.
- HEPPLE, R. T., QIN, M., NAKAMOTO, H. & GOTO, S. 2008. Caloric restriction optimizes the proteasome pathway with aging in rat plantaris muscle: implications for sarcopenia. *American Journal of Physiology-Regulatory, Integrative and Comparative Physiology*, 295, R1231-R1237.
- HERMAN, M. A., PERONI, O. D., VILLORIA, J., SCHON, M. R., ABUMRAD, N. A., BLUHER, M., KLEIN, S. & KAHN, B. B. 2012. A novel ChREBP isoform in adipose tissue regulates systemic glucose metabolism. *Nature*, 484, 333-8.
- HERTZEL, A. V., BENNAARS-EIDEN, A. & BERNLOHR, D. A. 2002. Increased lipolysis in transgenic animals overexpressing the epithelial fatty acid binding protein in adipose cells. *Journal of Lipid Research*, 43, 2105-2111.
- HEYMSFIELD, S. B., CRISTINA GONZALEZ, M. C., SHEN, W., REDMAN, L. & THOMAS, D. 2014. Weight Loss Composition is One-Fourth Fat-Free Mass: A Critical Review and Critique of This Widely Cited Rule. *Obesity reviews : an official journal of the International Association for the Study of Obesity*, 15, 310-321.
- HIGAMI, Y., BARGER, J. L., PAGE, G. P., ALLISON, D. B., SMITH, S. R., PROLLA, T. A. & WEINDRUCH, R. 2006. Energy Restriction Lowers the Expression of Genes Linked to Inflammation, the Cytoskeleton, the Extracellular Matrix, and Angiogenesis in Mouse Adipose Tissue. *The Journal of Nutrition*, 136, 343-352.
- HIGASHIDA, K., FUJIMOTO, E., HIGUCHI, M. & TERADA, S. 2013. Effects of alternate-day fasting on high-fat diet-induced insulin resistance in rat skeletal muscle. *Life Sciences*, 93, 208-213.
- HILL, J. O., SCHLUNDT, D. G., SBROCCO, T., SHARP, T., POPE-CORDLE, J., STETSON, B., KALER, M. & HEIM, C. 1989. Evaluation of an alternatingcalorie diet with and without exercise in the treatment of obesity. *Am J Clin Nutr*, 50, 248-54.
- HILTON, C., NEVILLE, M. J. & KARPE, F. 2013. MicroRNAs in adipose tissue: their role in adipogenesis and obesity. *Int J Obes*, 37, 325-32.
- HIRSCH, J., FRIED, S. K., EDENS, N. K. & LEIBEL, R. L. 1989. The fat cell. *Med Clin North Am*, 73, 83-96.
- HOEHN, K. L., HOHNEN-BEHRENS, C., CEDERBERG, A., WU, L. E., TURNER, N., YUASA, T., EBINA, Y. & JAMES, D. E. 2008. IRS1-independent defects define major nodes of insulin resistance. *Cell Metab*, 7, 421-33.
- HOGGARD, N., MERCER, J. G., RAYNER, D. V., MOAR, K., TRAYHURN, P. & WILLIAMS, L. M. 1997. Localization of leptin receptor mRNA splice variants in murine peripheral tissues by RT-PCR and in situ hybridization. *Biochem Biophys Res Commun*, 232, 383-7.
- HOJMAN, P., FJELBYE, J., ZERAHN, B., CHRISTENSEN, J. F., DETHLEFSEN, C., LONKVIST, C. K., BRANDT, C., GISSEL, H., PEDERSEN, B. K. & GEHL, J.

2014. Voluntary exercise prevents cisplatin-induced muscle wasting during chemotherapy in mice. *PLoS One*, 9, e109030.

- HOLLEKIM-STRAND, S. M., BJORGAAS, M. R., ALBREKTSEN, G., TJONNA, A. E., WISLOFF, U. & INGUL, C. B. 2014. High-intensity interval exercise effectively improves cardiac function in patients with type 2 diabetes mellitus and diastolic dysfunction: a randomized controlled trial. *J Am Coll Cardiol*, 64, 1758-60.
- HOMMELBERG, P. P., LANGEN, R. C., SCHOLS, A. M., MENSINK, R. P. & PLAT, J. 2010. Inflammatory signaling in skeletal muscle insulin resistance: green signal for nutritional intervention? *Curr Opin Clin Nutr Metab Care*, 13, 647-55.
- HOTAMISLIGIL, G. S., JOHNSON, R. S., DISTEL, R. J., ELLIS, R., PAPAIOANNOU, V. E. & SPIEGELMAN, B. M. 1996. Uncoupling of obesity from insulin resistance through a targeted mutation in aP2, the adipocyte fatty acid binding protein. *Science*, 274, 1377-9.
- HOUTKOOPER, R. H., CANTO, C., WANDERS, R. J. & AUWERX, J. 2010. The secret life of NAD+: an old metabolite controlling new metabolic signaling pathways. *Endocr Rev*, 31, 194-223.
- HOWARD, J. K. & FLIER, J. S. 2006. Attenuation of leptin and insulin signaling by SOCS proteins. *Trends Endocrinol Metab*, 17, 365-71.
- HUANG, P., LI, S., SHAO, M., QI, Q., ZHAO, F., YOU, J., MAO, T., LI, W., YAN, Z. & LIU, Y. 2010. Calorie restriction and endurance exercise share potent antiinflammatory function in adipose tissues in ameliorating diet-induced obesity and insulin resistance in mice. *Nutr Metab (Lond)*, 7, 59.
- HUANG, S. & CZECH, M. P. 2007. The GLUT4 Glucose Transporter. *Cell Metabolism*, 5, 237-252.
- HUFFMAN, D. M., MOELLERING, D. R., GRIZZLE, W. E., STOCKARD, C. R., JOHNSON, M. S. & NAGY, T. R. 2008. Effect of exercise and calorie restriction on biomarkers of aging in mice. *Am J Physiol Regul Integr Comp Physiol*, 294, R1618-27.
- HUSS, J. M., KOPP, R. P. & KELLY, D. P. 2002. Peroxisome proliferator-activated receptor coactivator-1alpha (PGC-1alpha) coactivates the cardiac-enriched nuclear receptors estrogen-related receptor-alpha and -gamma. Identification of novel leucine-rich interaction motif within PGC-1alpha. *J Biol Chem*, 277, 40265-74.
- IFLAND, J. R., PREUSS, H. G., MARCUS, M. T., ROURKE, K. M., TAYLOR, W. C., BURAU, K., JACOBS, W. S., KADISH, W. & MANSO, G. 2009. Refined food addiction: a classic substance use disorder. *Med Hypotheses*, 72, 518-26.
- IJUIN, T., HATANO, N., HOSOOKA, T. & TAKENAWA, T. 2015. Regulation of insulin signaling in skeletal muscle by PIP3 phosphatase, SKIP, and endoplasmic reticulum molecular chaperone glucose-regulated protein 78. *Biochimica et Biophysica Acta (BBA) - Molecular Cell Research*, 1853, 3192-3201.
- JAGER, S., HANDSCHIN, C., ST-PIERRE, J. & SPIEGELMAN, B. M. 2007. AMPactivated protein kinase (AMPK) action in skeletal muscle via direct phosphorylation of PGC-1alpha. *Proc Natl Acad Sci U S A*, 104, 12017-22.
- JEPPESEN, P. B. & MADSEN, S. M. 2015. High Intensity Interval Training Improves Glycaemic Control and Pancreatic β Cell Function of Type 2 Diabetes Patients.

- JIN, M., WU, Y., WANG, J., CHEN, J., HUANG, Y., RAO, J. & FENG, C. 2016. MicroRNA-24 promotes 3T3-L1 adipocyte differentiation by directly targeting the MAPK7 signaling. *Biochem Biophys Res Commun*, 474, 76-82.
- JOHNSON, J. B., SUMMER, W., CUTLER, R. G., MARTIN, B., HYUN, D. H., DIXIT, V. D., PEARSON, M., NASSAR, M., TELLJOHANN, R., MAUDSLEY, S., CARLSON, O., JOHN, S., LAUB, D. R. & MATTSON, M. P. 2007. Alternate day calorie restriction improves clinical findings and reduces markers of oxidative stress and inflammation in overweight adults with moderate asthma. *Free Radic Biol Med*, 42, 665-74.
- JORDAN, S. D., KRUGER, M., WILLMES, D. M., REDEMANN, N., WUNDERLICH, F. T., BRONNEKE, H. S., MERKWIRTH, C., KASHKAR, H., OLKKONEN, V. M., BOTTGER, T., BRAUN, T., SEIBLER, J. & BRUNING, J. C. 2011. Obesityinduced overexpression of miRNA-143 inhibits insulin-stimulated AKT activation and impairs glucose metabolism. *Nat Cell Biol*, 13, 434-46.
- JOSLIN, P. M. N., BELL, R. K. & SWOAP, S. J. 2017. Obese mice on a high-fat alternate-day fasting regimen lose weight and improve glucose tolerance. *Journal of Animal Physiology and Animal Nutrition*, 101, 1036-1045.
- KADDAI, V., JAGER, J., GONZALEZ, T., NAJEM-LENDOM, R., BONNAFOUS, S., TRAN, A., LE MARCHAND-BRUSTEL, Y., GUAL, P., TANTI, J. F. & CORMONT, M. 2009. Involvement of TNF-alpha in abnormal adipocyte and muscle sortilin expression in obese mice and humans. *Diabetologia*, 52, 932-40.
- KADOWAKI, T. & YAMAUCHI, T. 2005. Adiponectin and adiponectin receptors. *Endocr Rev*, 26, 439-51.
- KAHN, B. B., ALQUIER, T., CARLING, D. & HARDIE, D. G. 2005. AMP-activated protein kinase: ancient energy gauge provides clues to modern understanding of metabolism. *Cell Metab*, 1, 15-25.
- KANE, S., SANO, H., LIU, S. C., ASARA, J. M., LANE, W. S., GARNER, C. C. & LIENHARD, G. E. 2002. A method to identify serine kinase substrates. Akt phosphorylates a novel adipocyte protein with a Rab GTPase-activating protein (GAP) domain. J Biol Chem, 277, 22115-8.
- KANEKO, S., IIDA, R. H., SUGA, T., FUKUI, T., MORITO, M. & YAMANE, A. 2011. Changes in triacylglycerol-accumulated fiber type, fiber type composition, and biogenesis in the mitochondria of the soleus muscle in obese rats. *Anat Rec* (*Hoboken*), 294, 1904-12.
- KANG, M., YAN, L. M., LI, Y. M., ZHANG, W. Y., WANG, H., TANG, A. Z. & OU,
  H. S. 2013. Inhibitory effect of microRNA-24 on fatty acid-binding protein expression on 3T3-L1 adipocyte differentiation. *Genet Mol Res*, 12, 5267-77.
- KARBOWSKA, J. & KOCHAN, Z. 2012. Intermittent fasting up-regulates Fsp27/Cidec gene expression in white adipose tissue. *Nutrition*, 28, 294-9.
- KARSTOFT, K., WINDING, K., KNUDSEN, S. H., JAMES, N. G., SCHEEL, M. M., OLESEN, J., HOLST, J. J., PEDERSEN, B. K. & SOLOMON, T. P. 2014. Mechanisms behind the superior effects of interval vs continuous training on glycaemic control in individuals with type 2 diabetes: a randomised controlled trial. *Diabetologia*, 57, 2081-93.
- KARSTOFT, K., WINDING, K., KNUDSEN, S. H., NIELSEN, J. S., THOMSEN, C., PEDERSEN, B. K. & SOLOMON, T. P. 2013. The effects of free-living intervalwalking training on glycemic control, body composition, and physical fitness in

type 2 diabetic patients: a randomized, controlled trial. *Diabetes Care*, 36, 228-36.

- KASHYAP, S., BELFORT, R., GASTALDELLI, A., PRATIPANAWATR, T., BERRIA, R., PRATIPANAWATR, W., BAJAJ, M., MANDARINO, L., DEFRONZO, R. & CUSI, K. 2003. A sustained increase in plasma free fatty acids impairs insulin secretion in nondiabetic subjects genetically predisposed to develop type 2 diabetes. *Diabetes*, 52, 2461-74.
- KEATING, S. E., JOHNSON, N. A., MIELKE, G. I. & COOMBES, J. S. 2017. A systematic review and meta-analysis of interval training versus moderateintensity continuous training on body adiposity. *Obes Rev*, 18, 943-964.
- KEATING, S. E., MACHAN, E. A., O'CONNOR, H. T., GEROFI, J. A., SAINSBURY, A., CATERSON, I. D. & JOHNSON, N. A. 2014. Continuous exercise but not high intensity interval training improves fat distribution in overweight adults. J Obes, 2014, 834865.
- KELLEY, D. E., GOODPASTER, B., WING, R. R. & SIMONEAU, J.-A. 1999. Skeletal muscle fatty acid metabolism in association with insulin resistance, obesity, and weight loss. *American Journal of Physiology - Endocrinology and Metabolism*, 277, E1130-E1141.
- KELLEY, D. E. & GOODPASTER, B. H. 2001. Skeletal muscle triglyceride. An aspect of regional adiposity and insulin resistance. *Diabetes Care*, 24, 933-41.
- KEMMLER, W., SCHARF, M., LELL, M., PETRASEK, C. & VON STENGEL, S. 2014. High versus moderate intensity running exercise to impact cardiometabolic risk factors: the randomized controlled RUSH-study. *Biomed Res Int*, 2014, 843095.
- KENDRICK, A. A., CHOUDHURY, M., RAHMAN, S. M., MCCURDY, C. E., FRIEDERICH, M., VAN HOVE, J. L., WATSON, P. A., BIRDSEY, N., BAO, J., GIUS, D., SACK, M. N., JING, E., KAHN, C. R., FRIEDMAN, J. E. & JONSCHER, K. R. 2011. Fatty liver is associated with reduced SIRT3 activity and mitochondrial protein hyperacetylation. *Biochem J*, 433, 505-14.
- KERSTEN, S. 2014. Physiological regulation of lipoprotein lipase. *Biochim Biophys Acta*, 1841, 919-33.
- KERSTEN, S. B. W. 2000. Roles of PPARs in health and disease. *Nature*, 405, 421.
- KEWALRAMANI, G., BILAN, P. J. & KLIP, A. 2010. Muscle insulin resistance: assault by lipids, cytokines and local macrophages. *Curr Opin Clin Nutr Metab Care*, 13, 382-90.
- KHAN, I. M., PERRARD, X. Y., BRUNNER, G., LUI, H., SPARKS, L. M., SMITH, S. R., WANG, X., SHI, Z. Z., LEWIS, D. E., WU, H. & BALLANTYNE, C. M. 2015. Intermuscular and perimuscular fat expansion in obesity correlates with skeletal muscle T cell and macrophage infiltration and insulin resistance. *Int J Obes*, 39, 1607-1618.
- KHAN, T., MUISE, E. S., IYENGAR, P., WANG, Z. V., CHANDALIA, M., ABATE, N., ZHANG, B. B., BONALDO, P., CHUA, S. & SCHERER, P. E. 2009. Metabolic dysregulation and adipose tissue fibrosis: role of collagen VI. *Mol Cell Biol*, 29, 1575-91.
- KHVOROVA, A., REYNOLDS, A. & JAYASENA, S. D. 2003. Functional siRNAs and miRNAs exhibit strand bias. *Cell*, 115, 209-16.
- KIM, E., KIM, E. J., SEO, S.-W., HUR, C.-G., MCGREGOR, R. A. & CHOI, M.-S. 2014a. Meta-Review of Protein Network Regulating Obesity Between Validated

Obesity Candidate Genes in the White Adipose Tissue of High-Fat Diet-Induced Obese C57BL/6J Mice. *Critical Reviews in Food Science & Nutrition*, 54, 910-923.

- KIM, H. J., KIM, J. H., NOH, S., HUR, H. J., SUNG, M. J., HWANG, J. T., PARK, J. H., YANG, H. J., KIM, M. S., KWON, D. Y. & YOON, S. H. 2011. Metabolomic analysis of livers and serum from high-fat diet induced obese mice. *J Proteome Res*, 10, 722-31.
- KIM, J. B. 2016. Dynamic cross talk between metabolic organs in obesity and metabolic diseases. *Experimental &Amp; Molecular Medicine,* 48, e214.
- KIM, J. J., LI, P., HUNTLEY, J., CHANG, J. P., ARDEN, K. C. & OLEFSKY, J. M. 2009. FoxO1 Haploinsufficiency Protects Against High-Fat Diet–Induced Insulin Resistance With Enhanced Peroxisome Proliferator–Activated Receptor γ Activation in Adipose Tissue. *Diabetes*, 58, 1275-1282.
- KIM, J. Y., HICKNER, R. C., CORTRIGHT, R. L., DOHM, G. L. & HOUMARD, J. A. 2000. Lipid oxidation is reduced in obese human skeletal muscle. *Am J Physiol Endocrinol Metab*, 279, E1039-44.
- KIM, K. H., KIM, Y. H., SON, J. E., LEE, J. H., KIM, S., CHOE, M. S., MOON, J. H., ZHONG, J., FU, K., LENGLIN, F., YOO, J. A., BILAN, P. J., KLIP, A., NAGY, A., KIM, J. R., PARK, J. G., HUSSEIN, S. M., DOH, K. O., HUI, C. C. & SUNG, H. K. 2017. Intermittent fasting promotes adipose thermogenesis and metabolic homeostasis via VEGF-mediated alternative activation of macrophage. *Cell Res*, 27, 1309-1326.
- KIM, M., NEINAST, M. D., FRANK, A. P., SUN, K., PARK, J., ZEHR, J. A., VISHVANATH, L., MORSELLI, E., AMELOTTE, M., PALMER, B. F., GUPTA, R. K., SCHERER, P. E. & CLEGG, D. J. 2014b. ERα upregulates Phd3 to ameliorate HIF-1 induced fibrosis and inflammation in adipose tissue. *Molecular Metabolism*, 3, 642-651.
- KIM, S. Y., KIM, A. Y., LEE, H. W., SON, Y. H., LEE, G. Y., LEE, J. W., LEE, Y. S. & KIM, J. B. 2010. miR-27a is a negative regulator of adipocyte differentiation via suppressing PPARgamma expression. *Biochem Biophys Res Commun*, 392, 323-8.
- KIM, Y.-B., KOTANI, K., CIARALDI, T. P., HENRY, R. R. & KAHN, B. B. 2003. Insulin-Stimulated Protein Kinase C  $\lambda/\zeta$  Activity Is Reduced in Skeletal Muscle of Humans With Obesity and Type 2 Diabetes. *Reversal With Weight Reduction*, 52, 1935-1942.
- KIM, Y.-B., NIKOULINA, S. E., CIARALDI, T. P., HENRY, R. R. & KAHN, B. B. 1999. Normal insulin-dependent activation of Akt/protein kinase B, with diminished activation of phosphoinositide 3-kinase, in muscle in type 2 diabetes. *The Journal of Clinical Investigation*, 104, 733-741.
- KIRIAKIDOU, M., TAN, G. S., LAMPRINAKI, S., DE PLANELL-SAGUER, M., NELSON, P. T. & MOURELATOS, Z. 2007. An mRNA m7G cap binding-like motif within human Ago2 represses translation. *Cell*, 129, 1141-51.
- KLEMPEL, M. C., KROEGER, C. M., BHUTANI, S., TREPANOWSKI, J. F. & VARADY, K. A. 2012. Intermittent fasting combined with calorie restriction is effective for weight loss and cardio-protection in obese women. *Nutr J*, 11, 1475-2891.

- KLEMPEL, M. C., KROEGER, C. M. & VARADY, K. A. 2013. Alternate day fasting (ADF) with a high-fat diet produces similar weight loss and cardio-protection as ADF with a low-fat diet. *Metabolism*, 62, 137-43.
- KLÖTING, N. & BLÜHER, M. 2014. Adipocyte dysfunction, inflammation and metabolic syndrome. *Reviews In Endocrine & Metabolic Disorders*, 15, 277-287.
- KO, J. & KIM, K. 2013. Effects of exercise and diet composition on expression of MCP-1 and oxidative stress-related mRNA of adipose tissue in diet-induced obese mice. *Journal of Exercise Nutrition & Biochemistry*, 17, 181-188.
- KONG, Z., FAN, X., SUN, S., SONG, L., SHI, Q. & NIE, J. 2016. Comparison of High-Intensity Interval Training and Moderate-to-Vigorous Continuous Training for Cardiometabolic Health and Exercise Enjoyment in Obese Young Women: A Randomized Controlled Trial. *PLoS One*, 11, e0158589.
- KONSTANTINIDES, S., SCHAFER, K., KOSCHNICK, S. & LOSKUTOFF, D. J. 2001. Leptin-dependent platelet aggregation and arterial thrombosis suggests a mechanism for atherothrombotic disease in obesity. *J Clin Invest*, 108, 1533-40.
- KORNFELD, J. W., BAITZEL, C., KONNER, A. C., NICHOLLS, H. T., VOGT, M. C., HERRMANNS, K., SCHEJA, L., HAUMAITRE, C., WOLF, A. M., KNIPPSCHILD, U., SEIBLER, J., CEREGHINI, S., HEEREN, J., STOFFEL, M. & BRUNING, J. C. 2013. Obesity-induced overexpression of miR-802 impairs glucose metabolism through silencing of Hnf1b. *Nature*, 494, 111-5.
- KOUTSOULIDOU, A., MASTROYIANNOPOULOS, N. P., FURLING, D., UNEY, J. B. & PHYLACTOU, L. A. 2011. Expression of miR-1, miR-133a, miR-133b and miR-206 increases during development of human skeletal muscle. *BMC Dev Biol*, 11, 34.
- KRAEMER, F. B. & SHEN, W. J. 2002. Hormone-sensitive lipase: control of intracellular tri-(di-)acylglycerol and cholesteryl ester hydrolysis. *J Lipid Res*, 43, 1585-94.
- KRAHMER, N., FARESE, R. V. & WALTHER, T. C. 2013. Balancing the fat: lipid droplets and human disease. *EMBO Molecular Medicine*, 5, 905-915.
- KRAHMER, N., GUO, Y., WILFLING, F., HILGER, M., LINGRELL, S., HEGER, K., NEWMAN, H. W., SCHMIDT-SUPPRIAN, M., VANCE, D. E., MANN, M., FARESE, R. V., JR. & WALTHER, T. C. 2011. Phosphatidylcholine synthesis for lipid droplet expansion is mediated by localized activation of CTP:phosphocholine cytidylyltransferase. *Cell Metab*, 14, 504-15.
- KRAMER, H. F., WITCZAK, C. A., TAYLOR, E. B., FUJII, N., HIRSHMAN, M. F. & GOODYEAR, L. J. 2006. AS160 regulates insulin- and contraction-stimulated glucose uptake in mouse skeletal muscle. *J Biol Chem*, 281, 31478-85.
- KRSKOVA, K., ECKERTOVA, M., KUKAN, M., KUBA, D., KEBIS, A., OLSZANECKI, R., SUSKI, M., GAJDOSECHOVA, L. & ZORAD, S. 2012. Aerobic training lasting for 10 weeks elevates the adipose tissue FABP4, Gialpha, and adiponectin expression associated by a reduced protein oxidation. *Endocr Regul*, 46, 137-46.
- KRÜTZFELDT, J. 2016. Strategies to use microRNAs as therapeutic targets. Best Practice & Research Clinical Endocrinology & Metabolism, 30, 551-561.
- KUMAR, S. & KAUR, G. 2013. Intermittent Fasting Dietary Restriction Regimen Negatively Influences Reproduction in Young Rats: A Study of Hypothalamo-Hypophysial-Gonadal Axis. *PLOS ONE*, 8, e52416.

- KURIYAMA, H., SHIMOMURA, I., KISHIDA, K., KONDO, H., FURUYAMA, N., NISHIZAWA, H., MAEDA, N., MATSUDA, M., NAGARETANI, H., KIHARA, S., NAKAMURA, T., TOCHINO, Y., FUNAHASHI, T. & MATSUZAWA, Y. 2002. Coordinated regulation of fat-specific and liverspecific glycerol channels, aquaporin adipose and aquaporin 9. *Diabetes*, 51, 2915-21.
- KUSHNER, R. F. 2014. Weight loss strategies for treatment of obesity. *Prog Cardiovasc Dis*, 56, 465-72.
- LACKEY, D. E. & OLEFSKY, J. M. 2016. Regulation of metabolism by the innate immune system. *Nat Rev Endocrinol*, 12, 15-28.
- LAFONTAN, M. 2008. Advances in adipose tissue metabolism. *International Journal of Obesity*, 32, S39-S51.
- LAFONTAN, M. & LANGIN, D. 2009. Lipolysis and lipid mobilization in human adipose tissue. *Prog Lipid Res*, 48, 275-97.
- LARSEN, S., NIELSEN, J., HANSEN, C. N., NIELSEN, L. B., WIBRAND, F., STRIDE, N., SCHRODER, H. D., BOUSHEL, R., HELGE, J. W., DELA, F. & HEY-MOGENSEN, M. 2012. Biomarkers of mitochondrial content in skeletal muscle of healthy young human subjects. *The Journal of Physiology*, 590, 3349-3360.
- LARSON-MEYER, D. E., HEILBRONN, L. K., REDMAN, L. M., NEWCOMER, B. R., FRISARD, M. I., ANTON, S., SMITH, S. R., ALFONSO, A. & RAVUSSIN, E. 2006. Effect of calorie restriction with or without exercise on insulin sensitivity, beta-cell function, fat cell size, and ectopic lipid in overweight subjects. *Diabetes Care*, 29, 1337-44.
- LE, N. H., KIM, C.-S., PARK, T., PARK, J. H. Y., SUNG, M.-K., LEE, D. G., HONG, S.-M., CHOE, S.-Y., GOTO, T., KAWADA, T. & YU, R. 2014. Quercetin Protects against Obesity-Induced Skeletal Muscle Inflammation and Atrophy. *Mediators of Inflammation*, 2014, 834294.
- LEAL, V. D. O. & MAFRA, D. 2013. Adipokines in obesity. *Clinica Chimica Acta*, 419, 87-94.
- LEE, Y., AHN, C., HAN, J., CHOI, H., KIM, J., YIM, J., LEE, J., PROVOST, P., RADMARK, O., KIM, S. & KIM, V. N. 2003. The nuclear RNase III Drosha initiates microRNA processing. *Nature*, 425, 415-9.
- LEE, Y. S., KIM, J. W., OSBORNE, O., OH, D. Y., SASIK, R., SCHENK, S., CHEN, A., CHUNG, H., MURPHY, A., WATKINS, S. M., QUEHENBERGER, O., JOHNSON, R. S. & OLEFSKY, J. M. 2014. Increased adipocyte O2 consumption triggers HIF-1alpha, causing inflammation and insulin resistance in obesity. *Cell*, 157, 1339-52.
- LEHMAN, J. J., BARGER, P. M., KOVACS, A., SAFFITZ, J. E., MEDEIROS, D. M. & KELLY, D. P. 2000. Peroxisome proliferator-activated receptor gamma coactivator-1 promotes cardiac mitochondrial biogenesis. *J Clin Invest*, 106, 847-56.
- LETEXIER, D., PINTEUR, C., LARGE, V., FRERING, V. & BEYLOT, M. 2003. Comparison of the expression and activity of the lipogenic pathway in human and rat adipose tissue. *J Lipid Res*, 44, 2127-34.
- LI, G., XIE, C., LU, S., NICHOLS, R. G., TIAN, Y., LI, L., PATEL, D., MA, Y., BROCKER, C. N., YAN, T., KRAUSZ, K. W., XIANG, R., GAVRILOVA, O., PATTERSON, A. D. & GONZALEZ, F. J. 2017a. Intermittent Fasting Promotes

White Adipose Browning and Decreases Obesity by Shaping the Gut Microbiota. *Cell Metabolism*, 26, 672-685.e4.

- LI, J., DEFEA, K. & ROTH, R. A. 1999. Modulation of Insulin Receptor Substrate-1 Tyrosine Phosphorylation by an Akt/Phosphatidylinositol 3-Kinase Pathway. *Journal of Biological Chemistry*, 274, 9351-9356.
- LI, L., WANG, Z. & ZUO, Z. 2013. Chronic intermittent fasting improves cognitive functions and brain structures in mice. *PLoS One*, 8, e66069.
- LI, X. 2013. SIRT1 and energy metabolism. *Acta Biochimica et Biophysica Sinica*, 45, 51-60.
- LI, X., HIGASHIDA, K., KAWAMURA, T. & HIGUCHI, M. 2016. Alternate-Day High-Fat Diet Induces an Increase in Mitochondrial Enzyme Activities and Protein Content in Rat Skeletal Muscle. *Nutrients*, 8, 203.
- LI, Y., MA, Z., JIANG, S., HU, W., LI, T., DI, S., WANG, D. & YANG, Y. 2017b. A global perspective on FOXO1 in lipid metabolism and lipid-related diseases. *Progress in Lipid Research*, 66, 42-49.
- LIANG, H. & WARD, W. F. 2006. PGC-1alpha: a key regulator of energy metabolism. *Adv Physiol Educ*, 30, 145-51.
- LIN, S. J., FORD, E., HAIGIS, M., LISZT, G. & GUARENTE, L. 2004. Calorie restriction extends yeast life span by lowering the level of NADH. *Genes Dev*, 18, 12-6.
- LIN, Y.-Y., CHOU, C.-F., GIOVARELLI, M., BRIATA, P., GHERZI, R. & CHEN, C.-Y. 2014. KSRP and MicroRNA 145 Are Negative Regulators of Lipolysis in White Adipose Tissue. *Molecular and Cellular Biology*, 34, 2339-2349.
- LITTLE, J. P., GILLEN, J. B., PERCIVAL, M. E., SAFDAR, A., TARNOPOLSKY, M. A., PUNTHAKEE, Z., JUNG, M. E. & GIBALA, M. J. 2011. Low-volume highintensity interval training reduces hyperglycemia and increases muscle mitochondrial capacity in patients with type 2 diabetes. J Appl Physiol (1985), 111, 1554-60.
- LITTLE, J. P., JUNG, M. E., WRIGHT, A. E., WRIGHT, W. & MANDERS, R. J. 2014. Effects of high-intensity interval exercise versus continuous moderate-intensity exercise on postprandial glycemic control assessed by continuous glucose monitoring in obese adults. *Appl Physiol Nutr Metab*, 39, 835-41.
- LITTLE, J. P., SAFDAR, A., WILKIN, G. P., TARNOPOLSKY, M. A. & GIBALA, M. J. 2010. A practical model of low-volume high-intensity interval training induces mitochondrial biogenesis in human skeletal muscle: potential mechanisms. J Physiol, 588, 1011-22.
- LIU, J., CARMELL, M. A., RIVAS, F. V., MARSDEN, C. G., THOMSON, J. M., SONG, J. J., HAMMOND, S. M., JOSHUA-TOR, L. & HANNON, G. J. 2004. Argonaute2 is the catalytic engine of mammalian RNAi. *Science*, 305, 1437-41.
- LOCKE, A. E., KAHALI, B., BERNDT, S. I., JUSTICE, A. E., PERS, T. H., DAY, F.
  R., POWELL, C., VEDANTAM, S., BUCHKOVICH, M. L., YANG, J., CROTEAU-CHONKA, D. C., ESKO, T., FALL, T., FERREIRA, T., GUSTAFSSON, S., KUTALIK, Z., LUAN, J. A., MÄGI, R., RANDALL, J. C., WINKLER, T. W., WOOD, A. R., WORKALEMAHU, T., FAUL, J. D., SMITH, J. A., HUA ZHAO, J., ZHAO, W., CHEN, J., FEHRMANN, R., HEDMAN, Å.
  K., KARJALAINEN, J., SCHMIDT, E. M., ABSHER, D., AMIN, N., ANDERSON, D., BEEKMAN, M., BOLTON, J. L., BRAGG-GRESHAM, J. L., BUYSKE, S., DEMIRKAN, A., DENG, G., EHRET, G. B., FEENSTRA, B.,

FEITOSA, M. F., FISCHER, K., GOEL, A., GONG, J., JACKSON, A. U., KANONI, S., KLEBER, M. E., KRISTIANSSON, K., LIM, U., LOTAY, V., MANGINO, M., MATEO LEACH, I., MEDINA-GOMEZ, C., MEDLAND, S. E., NALLS, M. A., PALMER, C. D., PASKO, D., PECHLIVANIS, S., PETERS, PROKOPENKO, I., SHUNGIN, D., STANČÁKOVÁ, M. J., A., STRAWBRIDGE, R. J., JU SUNG, Y., TANAKA, T., TEUMER, A., TROMPET, S., VAN DER LAAN, S. W., VAN SETTEN, J., VAN VLIET-OSTAPTCHOUK, J. V., WANG, Z., YENGO, L., ZHANG, W., ISAACS, A., ALBRECHT, E., ÄRNLÖV, J., ARSCOTT, G. M., ATTWOOD, A. P., BANDINELLI, S., BARRETT, A., BAS, I. N., BELLIS, C., BENNETT, A. J., R., BLÜHER, М., BERNE. С., BLAGIEVA, BÖHRINGER, S.. BONNYCASTLE, L. L., BÖTTCHER, Y., BOYD, H. A., BRUINENBERG, M., CASPERSEN, I. H., IDA CHEN, Y.-D., CLARKE, R., WARWICK DAW, E., DE CRAEN, A. J. M., DELGADO, G., DIMITRIOU, M., et al. 2015. Genetic studies of body mass index yield new insights for obesity biology. Nature, 518, 197.

- LOENNEKE, J. P. & LOPRINZI, P. D. 2016. Obesity is associated with insulin resistance but not skeletal muscle dysfunction or all-cause mortality. *Age*, 38, 2.
- LOFFREDA, S., YANG, S. Q., LIN, H. Z., KARP, C. L., BRENGMAN, M. L., WANG,
  D. J., KLEIN, A. S., BULKLEY, G. B., BAO, C., NOBLE, P. W., LANE, M. D.
  & DIEHL, A. M. 1998. Leptin regulates proinflammatory immune responses. *Faseb j*, 12, 57-65.
- LOLLMANN, B., GRUNINGER, S., STRICKER-KRONGRAD, A. & CHIESI, M. 1997. Detection and quantification of the leptin receptor splice variants Ob-Ra, b, and, e in different mouse tissues. *Biochem Biophys Res Commun*, 238, 648-52.
- LONG, Y. C., BARNES, B. R., MAHLAPUU, M., STEILER, T. L., MARTINSSON, S., LENG, Y., WALLBERG-HENRIKSSON, H., ANDERSSON, L. & ZIERATH, J. R. 2005. Role of AMP-activated protein kinase in the coordinated expression of genes controlling glucose and lipid metabolism in mouse white skeletal muscle. *Diabetologia*, 48, 2354-64.
- LONGO, V. D. & MATTSON, M. P. 2014. Fasting: Molecular Mechanisms and Clinical Applications. *Cell metabolism*, 19, 181-192.
- LORENZO, M., FERNANDEZ-VELEDO, S., VILA-BEDMAR, R., GARCIA-GUERRA, L., DE ALVARO, C. & NIETO-VAZQUEZ, I. 2008. Insulin resistance induced by tumor necrosis factor-alpha in myocytes and brown adipocytes. J Anim Sci, 86, E94-104.
- LUMENG, C. N., BODZIN, J. L. & SALTIEL, A. R. 2007. Obesity induces a phenotypic switch in adipose tissue macrophage polarization. *J Clin Invest*, 117, 175-84.
- LUMENG, C. N. & SALTIEL, A. R. 2011. Inflammatory links between obesity and metabolic disease. *J Clin Invest*, 121, 2111-7.
- LUND, J., HAFSTAD, A. D., BOARDMAN, N. T., ROSSVOLL, L., ROLIM, N. P., AHMED, M. S., FLORHOLMEN, G., ATTRAMADAL, H., WISLOFF, U., LARSEN, T. S. & AASUM, E. 2015. Exercise training promotes cardioprotection through oxygen-sparing action in high-fat fed mice. *Am J Physiol Heart Circ Physiol*, ajpheart.00734.2014.
- LUO, L. & LIU, M. 2016. Adipose tissue in control of metabolism. *J Endocrinol*, 231, R77-r99.

- MAEDA, K., CAO, H., KONO, K., GORGUN, C. Z., FURUHASHI, M., UYSAL, K. T., CAO, Q., ATSUMI, G., MALONE, H., KRISHNAN, B., MINOKOSHI, Y., KAHN, B. B., PARKER, R. A. & HOTAMISLIGIL, G. S. 2005. Adipocyte/macrophage fatty acid binding proteins control integrated metabolic responses in obesity and diabetes. *Cell Metab*, 1, 107-19.
- MAGNUSSON, B., GUMMESSON, A., GLAD, C. A., GOEDECKE, J. H., JERNAS, M., LYSTIG, T. C., CARLSSON, B., FAGERBERG, B., CARLSSON, L. M. & SVENSSON, P. A. 2008. Cell death-inducing DFF45-like effector C is reduced by caloric restriction and regulates adipocyte lipid metabolism. *Metabolism*, 57, 1307-13.
- MANTEIGA, S., CHOI, K., JAYARAMAN, A. & LEE, K. 2013. Systems biology of adipose tissue metabolism: regulation of growth, signaling and inflammation. *Wiley Interdisciplinary Reviews. Systems Biology And Medicine*, **5**, 425-447.
- MARCINKIEWICZ, A., GAUTHIER, D., GARCIA, A. & BRASAEMLE, D. L. 2006. The phosphorylation of serine 492 of perilipin a directs lipid droplet fragmentation and dispersion. *J Biol Chem*, 281, 11901-9.
- MARCINKO, K., SIKKEMA, S. R., SAMAAN, M. C., KEMP, B. E., FULLERTON, M. D. & STEINBERG, G. R. 2015. High intensity interval training improves liver and adipose tissue insulin sensitivity. *Molecular Metabolism*, 4, 903-915.
- MARIMAN, E. C. M. & WANG, P. 2010. Adipocyte extracellular matrix composition, dynamics and role in obesity. *Cellular and Molecular Life Sciences*, 67, 1277-1292.
- MARTINEZ-BOTAS, J., ANDERSON, J. B., TESSIER, D., LAPILLONNE, A., CHANG, B. H., QUAST, M. J., GORENSTEIN, D., CHEN, K. H. & CHAN, L. 2000. Absence of perilipin results in leanness and reverses obesity in Lepr(db/db) mice. *Nat Genet*, 26, 474-9.
- MARTINS, C., KAZAKOVA, I., LUDVIKSEN, M., MEHUS, I., WISLOFF, U., KULSENG, B., MORGAN, L. & KING, N. 2016. High-Intensity Interval Training and Isocaloric Moderate-Intensity Continuous Training Result in Similar Improvements in Body Composition and Fitness in Obese Individuals. *Int J Sport Nutr Exerc Metab*, 26, 197-204.
- MATTSON, M. P., LONGO, V. D. & HARVIE, M. 2016. Impact of intermittent fasting on health and disease processes. *Ageing Research Reviews*.
- MATTSON, M. P., MOEHL, K., GHENA, N., SCHMAEDICK, M. & CHENG, A. 2018. Intermittent metabolic switching, neuroplasticity and brain health. *Nat Rev Neurosci*, 19, 63-80.
- MAUVAIS-JARVIS, F. 2015. Sex differences in metabolic homeostasis, diabetes, and obesity. *Biology of Sex Differences*, 6, 14.
- MCDANELD, T. G. 2009. MicroRNA: mechanism of gene regulation and application to livestock. *J Anim Sci*, 87, E21-8.
- MCGREGOR, R. A., POPPITT, S. D. & CAMERON-SMITH, D. 2014. Role of microRNAs in the age-related changes in skeletal muscle and diet or exercise interventions to promote healthy aging in humans. *Ageing Research Reviews*, 17, 25-33.
- MELDRUM, D. R., MORRIS, M. A. & GAMBONE, J. C. 2017. Obesity pandemic: causes, consequences, and solutions—but do we have the will? *Fertility and Sterility*, 107, 833-839.

- MENSHIKOVA, E. V., RITOV, V. B., TOLEDO, F. G., FERRELL, R. E., GOODPASTER, B. H. & KELLEY, D. E. 2005. Effects of weight loss and physical activity on skeletal muscle mitochondrial function in obesity. Am J Physiol Endocrinol Metab, 288, E818-25.
- MENSINK, M., HESSELINK, M. K., RUSSELL, A. P., SCHAART, G., SELS, J. P. & SCHRAUWEN, P. 2007. Improved skeletal muscle oxidative enzyme activity and restoration of PGC-1 alpha and PPAR beta/delta gene expression upon rosiglitazone treatment in obese patients with type 2 diabetes mellitus. *Int J Obes* (*Lond*), 31, 1302-10.
- MERSEY, B. D., JIN, P. & DANNER, D. J. 2005. Human microRNA (miR29b) expression controls the amount of branched chain alpha-ketoacid dehydrogenase complex in a cell. *Hum Mol Genet*, 14, 3371-7.
- MICHAEL, L. F., WU, Z., CHEATHAM, R. B., PUIGSERVER, P., ADELMANT, G., LEHMAN, J. J., KELLY, D. P. & SPIEGELMAN, B. M. 2001. Restoration of insulin-sensitive glucose transporter (GLUT4) gene expression in muscle cells by the transcriptional coactivator PGC-1. *Proc Natl Acad Sci U S A*, 98, 3820-5.
- MILAGRO, F. I., MANSEGO, M. L., DE MIGUEL, C. & MARTÍNEZ, J. A. 2013. Dietary factors, epigenetic modifications and obesity outcomes: Progresses and perspectives. *Molecular Aspects of Medicine*, 34, 782-812.
- MITRANUN, W., DEEROCHANAWONG, C., TANAKA, H. & SUKSOM, D. 2014. Continuous vs interval training on glycemic control and macro- and microvascular reactivity in type 2 diabetic patients. *Scand J Med Sci Sports*, 24, e69-76.
- MITTENDORFER, B. 2011. Origins of metabolic complications in obesity: adipose tissue and free fatty acid trafficking. *Current opinion in clinical nutrition and metabolic care*, 14, 535-541.
- MONTGOMERY, M. K., HALLAHAN, N. L., BROWN, S. H., LIU, M., MITCHELL, T. W., COONEY, G. J. & TURNER, N. 2013. Mouse strain-dependent variation in obesity and glucose homeostasis in response to high-fat feeding. *Diabetologia*, 56, 1129-39.
- MOORE, S. C., PATEL, A. V., MATTHEWS, C. E., BERRINGTON DE GONZALEZ, A., PARK, Y., KATKI, H. A., LINET, M. S., WEIDERPASS, E., VISVANATHAN, K., HELZLSOUER, K. J., THUN, M., GAPSTUR, S. M., HARTGE, P. & LEE, I. M. 2012. Leisure time physical activity of moderate to vigorous intensity and mortality: a large pooled cohort analysis. *PLoS Med*, 9, e1001335.
- MOOTHA, V. K., HANDSCHIN, C., ARLOW, D., XIE, X., ST PIERRE, J., SIHAG, S., YANG, W., ALTSHULER, D., PUIGSERVER, P., PATTERSON, N., WILLY, P. J., SCHULMAN, I. G., HEYMAN, R. A., LANDER, E. S. & SPIEGELMAN, B. M. 2004. Erralpha and Gabpa/b specify PGC-1alpha-dependent oxidative phosphorylation gene expression that is altered in diabetic muscle. *Proc Natl Acad Sci U S A*, 101, 6570-5.
- MOOTHA, V. K., LINDGREN, C. M., ERIKSSON, K. F., SUBRAMANIAN, A., SIHAG, S., LEHAR, J., PUIGSERVER, P., CARLSSON, E., RIDDERSTRALE, M., LAURILA, E., HOUSTIS, N., DALY, M. J., PATTERSON, N., MESIROV, J. P., GOLUB, T. R., TAMAYO, P., SPIEGELMAN, B., LANDER, E. S., HIRSCHHORN, J. N., ALTSHULER, D. & GROOP, L. C. 2003. PGC-1alpharesponsive genes involved in oxidative phosphorylation are coordinately downregulated in human diabetes. *Nat Genet*, 34, 267-73.

- MORRISON, R. F. & FARMER, S. R. 1999. Role of PPARγ in Regulating a Cascade Expression of Cyclin-dependent Kinase Inhibitors, p18(INK4c) and p21(Waf1/Cip1), during Adipogenesis. *Journal of Biological Chemistry*, 274, 17088-17097.
- MOUSAVI, B., BATHAIE, S. Z., FADAI, F., ASHTARI, Z., ALI BEIGI, N., FARHANG, S., HASHEMPOUR, S., SHAHHAMZEI, N. & HEIDARZADEH, H. 2015. Safety evaluation of saffron stigma (Crocus sativus L.) aqueous extract and crocin in patients with schizophrenia. *Avicenna J Phytomed*, 5, 413-9.
- MUOIO, D. M., KOVES, T. R., AN, J. & NEWGARD, C. B. 2008. Metabolic Mechanisms of Muscle Insulin Resistance. In: FEINGLOS, M. N. & BETHEL, M. A. (eds.) Type 2 Diabetes Mellitus: An Evidence-Based Approach to Practical Management. Totowa, NJ: Humana Press.
- MURGIA, M., SERRANO, A. L., CALABRIA, E., PALLAFACCHINA, G., LOMO, T. & SCHIAFFINO, S. 2000. Ras is involved in nerve-activity-dependent regulation of muscle genes. *Nat Cell Biol*, 2, 142-7.
- NAYA, F. J., MERCER, B., SHELTON, J., RICHARDSON, J. A., WILLIAMS, R. S. & OLSON, E. N. 2000. Stimulation of slow skeletal muscle fiber gene expression by calcineurin in vivo. *J Biol Chem*, 275, 4545-8.
- NEMOTO, S., FERGUSSON, M. M. & FINKEL, T. 2004. Nutrient availability regulates SIRT1 through a forkhead-dependent pathway. *Science*, 306, 2105-8.
- NICKELSON, K. J., STROMSDORFER, K. L., PICKERING, R. T., LIU, T. W., ORTINAU, L. C., KEATING, A. F. & PERFIELD, J. W., 2ND 2012. A comparison of inflammatory and oxidative stress markers in adipose tissue from weight-matched obese male and female mice. *Exp Diabetes Res*, 2012, 859395.
- NIE, Y., SATO, Y., WANG, C., YUE, F., KUANG, S. & GAVIN, T. P. 2016. Impaired exercise tolerance, mitochondrial biogenesis, and muscle fiber maintenance in miR-133a-deficient mice. *Faseb j*, 30, 3745-3758.
- NIELSEN, S., SCHEELE, C., YFANTI, C., AKERSTROM, T., NIELSEN, A. R., PEDERSEN, B. K. & LAYE, M. J. 2010. Muscle specific microRNAs are regulated by endurance exercise in human skeletal muscle. *J Physiol*, 588, 4029-37.
- NISHINO, N., TAMORI, Y., TATEYA, S., KAWAGUCHI, T., SHIBAKUSA, T., MIZUNOYA, W., INOUE, K., KITAZAWA, R., KITAZAWA, S., MATSUKI, Y., HIRAMATSU, R., MASUBUCHI, S., OMACHI, A., KIMURA, K., SAITO, M., AMO, T., OHTA, S., YAMAGUCHI, T., OSUMI, T., CHENG, J., FUJIMOTO, T., NAKAO, H., NAKAO, K., AIBA, A., OKAMURA, H., FUSHIKI, T. & KASUGA, M. 2008. FSP27 contributes to efficient energy storage in murine white adipocytes by promoting the formation of unilocular lipid droplets. J Clin Invest, 118, 2808-21.
- NIU, Y., YUAN, H. & FU, L. 2010. Aerobic exercise's reversal of insulin resistance by activating AMPKalpha-ACC-CPT1 signaling in the skeletal muscle of C57BL/6 mice. *Int J Sport Nutr Exerc Metab*, 20, 370-80.
- NYBO, L., SUNDSTRUP, E., JAKOBSEN, M. D., MOHR, M., HORNSTRUP, T., SIMONSEN, L., BULOW, J., RANDERS, M. B., NIELSEN, J. J., AAGAARD, P. & KRUSTRUP, P. 2010. High-intensity training versus traditional exercise interventions for promoting health. *Med Sci Sports Exerc*, 42, 1951-8.
- O'BRIEN, P. D., HINDER, L. M., CALLAGHAN, B. C. & FELDMAN, E. L. 2017. Neurological consequences of obesity. *The Lancet Neurology*, 16, 465-477.

- O'NEILL, H. M., HOLLOWAY, G. P. & STEINBERG, G. R. 2013. AMPK regulation of fatty acid metabolism and mitochondrial biogenesis: Implications for obesity. *Molecular and Cellular Endocrinology*, 366, 135-151.
- OLEFSKY, J. M. & GLASS, C. K. 2010. Macrophages, inflammation, and insulin resistance. *Annu Rev Physiol*, 72, 219-46.
- OLIVO-MARSTON, S. E., HURSTING, S. D., PERKINS, S. N., SCHETTER, A., KHAN, M., CROCE, C., HARRIS, C. C. & LAVIGNE, J. 2014. Effects of calorie restriction and diet-induced obesity on murine colon carcinogenesis, growth and inflammatory factors, and microRNA expression. *PLoS ONE*, 9, e94765.
- ORTEGA, F. J., MORENO-NAVARRETE, J. M., PARDO, G., SABATER, M., HUMMEL, M., FERRER, A., RODRIGUEZ-HERMOSA, J. I., RUIZ, B., RICART, W., PERAL, B. & FERNANDEZ-REAL, J. M. 2010. MiRNA expression profile of human subcutaneous adipose and during adipocyte differentiation. *PLoS One*, 5, e9022.
- OSBORN, O. & OLEFSKY, J. M. 2012. The cellular and signaling networks linking the immune system and metabolism in disease. *Nat Med*, 18, 363-74.
- OUCHI, N., KIHARA, S., ARITA, Y., OKAMOTO, Y., MAEDA, K., KURIYAMA, H., HOTTA, K., NISHIDA, M., TAKAHASHI, M., MURAGUCHI, M., OHMOTO, Y., NAKAMURA, T., YAMASHITA, S., FUNAHASHI, T. & MATSUZAWA, Y. 2000. Adiponectin, an adipocyte-derived plasma protein, inhibits endothelial NF-kappaB signaling through a cAMP-dependent pathway. *Circulation*, 102, 1296-301.
- OUERGHI, N., FRADJ, M. K. B., BEZRATI, I., KHAMMASSI, M., FEKI, M., KAABACHI, N. & BOUASSIDA, A. 2017. Effects of high-intensity interval training on body composition, aerobic and anaerobic performance and plasma lipids in overweight/obese and normal-weight young men. *Biol Sport*, 34, 385-392.
- ÖZCAN, S. 2014. Minireview: MicroRNA Function in Pancreatic β Cells. *Molecular Endocrinology*, 28, 1922-1933.
- PALACIOS, O. M., CARMONA, J. J., MICHAN, S., CHEN, K. Y., MANABE, Y., III, J. L. W., GOODYEAR, L. J. & TONG, Q. 2009. Diet and exercise signals regulate SIRT3 and activate AMPK and PGC-1α in skeletal muscle. *Aging (Albany NY)*, 1, 771-783.
- PALMER, J. D., SOULE, B. P., SIMONE, B. A., ZAORSKY, N. G., JIN, L. & SIMONE, N. L. 2014. Review: MicroRNA expression altered by diet: Can food be medicinal? Ageing Research Reviews, 17, 16-24.
- PARK, M. H., KIM, D. H., LEE, E. K., KIM, N. D., IM, D. S., LEE, J., YU, B. P. & CHUNG, H. Y. 2014. Age-related inflammation and insulin resistance: a review of their intricate interdependency. *Archives of Pharmacal Research*, 37, 1507-1514.
- PARKER, L., STEPTO, N. K., SHAW, C. S., SERPIELLO, F. R., ANDERSON, M., HARE, D. L. & LEVINGER, I. 2016. Acute High-Intensity Interval Exercise-Induced Redox Signaling Is Associated with Enhanced Insulin Sensitivity in Obese Middle-Aged Men. *Frontiers in Physiology*, 7, 411.
- PARKER, L., TREWIN, A., LEVINGER, I., SHAW, C. S. & STEPTO, N. K. 2017. The effect of exercise-intensity on skeletal muscle stress kinase and insulin protein signaling. *PLOS ONE*, 12, e0171613.

- PARKER, R. & SHETH, U. 2007. P Bodies and the Control of mRNA Translation and Degradation. *Molecular Cell*, 25, 635-646.
- PASIAKOS, S. M. & MCCLUNG, J. P. 2013. miRNA Analysis for the Assessment of Exercise and Amino Acid Effects on Human Skeletal Muscle. Advances in Nutrition, 4, 412-417.
- PATSOURIS, D., CAO, J.-J., VIAL, G., BRAVARD, A., LEFAI, E., DURAND, A., DURAND, C., CHAUVIN, M.-A., LAUGERETTE, F., DEBARD, C., MICHALSKI, M.-C., LAVILLE, M., VIDAL, H. & RIEUSSET, J. 2014. Insulin Resistance is Associated with MCP1-Mediated Macrophage Accumulation in Skeletal Muscle in Mice and Humans. *PLOS ONE*, 9, e110653.
- PATTANAKUHAR, S., PONGCHAIDECHA, A., CHATTIPAKORN, N. & CHATTIPAKORN, S. C. 2016. The effect of exercise on skeletal muscle fibre type distribution in obesity: From cellular levels to clinical application. *Obesity Research & Clinical Practice*.
- PATTERSON, R. E., MARTÍNEZ, M. E., LAUGHLIN, G. A., LACROIX, A. Z., HARTMAN, S. J., NATARAJAN, L., SENGER, C. M., VILLASEÑOR, A., SEARS, D. D., MARINAC, C. & GALLO, L. C. 2015. Intermittent Fasting and Human Metabolic Health. *Journal of the Academy of Nutrition and Dietetics*.
- PATTERSON, R. E. & SEARS, D. D. 2017. Metabolic Effects of Intermittent Fasting. Annu Rev Nutr, 37, 371-393.
- PAZ, K., HEMI, R., LEROITH, D., KARASIK, A., ELHANANY, E., KANETY, H. & ZICK, Y. 1997. A Molecular Basis for Insulin Resistance: ELEVATED SERINE/THREONINE PHOSPHORYLATION OF IRS-1 AND IRS-2 INHIBITS THEIR BINDING TO THE JUXTAMEMBRANE REGION OF THE INSULIN RECEPTOR AND IMPAIRS THEIR ABILITY TO UNDERGO INSULIN-INDUCED TYROSINE PHOSPHORYLATION. Journal of Biological Chemistry, 272, 29911-29918.
- PEARSON, K. J., BAUR, J. A., LEWIS, K. N., PESHKIN, L., PRICE, N. L., LABINSKYY, N., SWINDELL, W. R., KAMARA, D., MINOR, R. K., PEREZ, E., JAMIESON, H. A., ZHANG, Y., DUNN, S. R., SHARMA, K., PLESHKO, N., WOOLLETT, L. A., CSISZAR, A., IKENO, Y., LE COUTEUR, D., ELLIOTT, P. J., BECKER, K. G., NAVAS, P., INGRAM, D. K., WOLF, N. S., UNGVARI, Z., SINCLAIR, D. A. & DE CABO, R. 2008. Resveratrol delays agerelated deterioration and mimics transcriptional aspects of dietary restriction without extending lifespan. *Cell metabolism*, 8, 157-168.
- PEDERSEN, B. K. & FEBBRAIO, M. A. 2012. Muscles, exercise and obesity: skeletal muscle as a secretory organ. *Nat Rev Endocrinol*, 8, 457-65.
- PENG, Y., YU, S., LI, H., XIANG, H., PENG, J. & JIANG, S. 2014. MicroRNAs: Emerging roles in adipogenesis and obesity. *Cellular Signalling*, 26, 1888-1896.
- PFLUGER, P. T., HERRANZ, D., VELASCO-MIGUEL, S., SERRANO, M. & TSCHOP, M. H. 2008. Sirt1 protects against high-fat diet-induced metabolic damage. *Proc Natl Acad Sci U S A*, 105, 9793-8.
- PHILLIPS, T. & LEEUWENBURGH, C. 2005. Muscle fiber specific apoptosis and TNFalpha signaling in sarcopenia are attenuated by life-long calorie restriction. *Faseb j*, 19, 668-70.
- PINILLA, F. G. 2006. The impact of diet and exercise on brain plasticity and disease. *Nutr Health*, 18, 277-84.

- POMERLEAU, M., IMBEAULT, P., PARKER, T. & DOUCET, E. 2004. Effects of exercise intensity on food intake and appetite in women. *The American Journal of Clinical Nutrition*, 80, 1230-1236.
- POPE, B. D., WARREN, C. R., PARKER, K. K. & COWAN, C. A. 2016. Microenvironmental Control of Adipocyte Fate and Function. *Trends in Cell Biology*, 26, 745-755.
- POY, M. N., HAUSSER, J., TRAJKOVSKI, M., BRAUN, M., COLLINS, S., RORSMAN, P., ZAVOLAN, M. & STOFFEL, M. 2009. miR-375 maintains normal pancreatic alpha- and beta-cell mass. *Proc Natl Acad Sci U S A*, 106, 5813-8.
- PRICE, N. L. & FERNÁNDEZ-HERNANDO, C. 2016. miRNA regulation of white and brown adipose tissue differentiation and function. *Biochimica et Biophysica Acta* (BBA) - Molecular and Cell Biology of Lipids, 1861, 2104-2110.
- PUIGSERVER, P., WU, Z., PARK, C. W., GRAVES, R., WRIGHT, M. & SPIEGELMAN, B. M. 1998. A cold-inducible coactivator of nuclear receptors linked to adaptive thermogenesis. *Cell*, 92, 829-39.
- PURI, V., RANJIT, S., KONDA, S., NICOLORO, S. M., STRAUBHAAR, J., CHAWLA, A., CHOUINARD, M., LIN, C., BURKART, A., CORVERA, S., PERUGINI, R. A. & CZECH, M. P. 2008. Cidea is associated with lipid droplets and insulin sensitivity in humans. *Proc Natl Acad Sci U S A*, 105, 7833-8.
- QUEIROZ, A. L., SOUSA, T. A., TEODORO, B. G., BOMFIM, L. M., SAMPAIO, I. H. & SILVEIRA, L. R. 2013. The miR-696 contributes to insulin resistance by reducing mitochondrial function and antioxidant capacity in skeletal muscle cells. *The FASEB Journal*, 27, 1010.6-1010.6.
- RACIL, G., BEN OUNIS, O., HAMMOUDA, O., KALLEL, A., ZOUHAL, H., CHAMARI, K. & AMRI, M. 2013. Effects of high vs. moderate exercise intensity during interval training on lipids and adiponectin levels in obese young females. *Eur J Appl Physiol*, 113, 2531-40.
- RANATUNGA, K. W. & THOMAS, P. E. 1990. Correlation between shortening velocity, force-velocity relation and histochemical fibre-type composition in rat muscles. *J Muscle Res Cell Motil*, 11, 240-50.
- REN, J. M., SEMENKOVICH, C. F., GULVE, E. A., GAO, J. & HOLLOSZY, J. O. 1994. Exercise induces rapid increases in GLUT4 expression, glucose transport capacity, and insulin-stimulated glycogen storage in muscle. *J Biol Chem*, 269, 14396-401.
- REYNA, S. M., GHOSH, S., TANTIWONG, P., MEKA, C. S., EAGAN, P., JENKINSON, C. P., CERSOSIMO, E., DEFRONZO, R. A., COLETTA, D. K., SRIWIJITKAMOL, A. & MUSI, N. 2008. Elevated toll-like receptor 4 expression and signaling in muscle from insulin-resistant subjects. *Diabetes*, 57, 2595-602.
- REYNOLDS, T. H., BANERJEE, S., SHARMA, V. M., DONOHUE, J., COULDWELL, S., SOSINSKY, A., FRULLA, A., ROBINSON, A. & PURI, V. 2015. Effects of a High Fat Diet and Voluntary Wheel Running Exercise on Cidea and Cidec Expression in Liver and Adipose Tissue of Mice. *PLoS ONE*, 10, e0130259.
- RIBEIRO, P. A. B., BOIDIN, M., JUNEAU, M., NIGAM, A. & GAYDA, M. 2017. High-intensity interval training in patients with coronary heart disease: Prescription models and perspectives. *Annals of Physical and Rehabilitation Medicine*, 60, 50-57.

- RICHARDSON, D. K., KASHYAP, S., BAJAJ, M., CUSI, K., MANDARINO, S. J., FINLAYSON, J., DEFRONZO, R. A., JENKINSON, C. P. & MANDARINO, L. J. 2005. Lipid infusion decreases the expression of nuclear encoded mitochondrial genes and increases the expression of extracellular matrix genes in human skeletal muscle. J Biol Chem, 280, 10290-7.
- RINGSEIS, R., ROSENBAUM, S., GESSNER, D. K., HERGES, L., KUBENS, J. F., MOOREN, F. C., KRUGER, K. & EDER, K. 2013. Supplementing obese Zucker rats with niacin induces the transition of glycolytic to oxidative skeletal muscle fibers. *J Nutr*, 143, 125-31.
- RINK, L. 1998. Tumor Necrosis Factor α A2 Delves, Peter J. Encyclopedia of Immunology (Second Edition). Oxford: Elsevier.
- RITOV, V. B., MENSHIKOVA, E. V., AZUMA, K., WOOD, R., TOLEDO, F. G., GOODPASTER, B. H., RUDERMAN, N. B. & KELLEY, D. E. 2010. Deficiency of electron transport chain in human skeletal muscle mitochondria in type 2 diabetes mellitus and obesity. *Am J Physiol Endocrinol Metab*, 298, E49-58.
- ROCKL, K. S., HIRSHMAN, M. F., BRANDAUER, J., FUJII, N., WITTERS, L. A. & GOODYEAR, L. J. 2007. Skeletal muscle adaptation to exercise training: AMPactivated protein kinase mediates muscle fiber type shift. *Diabetes*, 56, 2062-9.
- ROCKL, K. S., WITCZAK, C. A. & GOODYEAR, L. J. 2008. Signaling mechanisms in skeletal muscle: acute responses and chronic adaptations to exercise. *IUBMB Life*, 60, 145-53.
- RÖCKL, K. S. C., HIRSHMAN, M. F., BRANDAUER, J., FUJII, N., WITTERS, L. A. & GOODYEAR, L. J. 2007. Skeletal Muscle Adaptation to Exercise Training. *Diabetes*, 56, 2062.
- RODGERS, J. T., LERIN, C., HAAS, W., GYGI, S. P., SPIEGELMAN, B. M. & PUIGSERVER, P. 2005. Nutrient control of glucose homeostasis through a complex of PGC-1alpha and SIRT1. *Nature*, 434, 113-8.
- ROGGE, M. M. & GAUTAM, B. 2017. Biology of obesity and weight regain: Implications for clinical practice. J Am Assoc Nurse Pract, 29, S15-s29.
- ROMIEU, I., DOSSUS, L., BARQUERA, S., BLOTTIERE, H. M., FRANKS, P. W., GUNTER, M., HWALLA, N., HURSTING, S. D., LEITZMANN, M., MARGETTS, B., NISHIDA, C., POTISCHMAN, N., SEIDELL, J., STEPIEN, M., WANG, Y., WESTERTERP, K., WINICHAGOON, P., WISEMAN, M. & WILLETT, W. C. 2017. Energy balance and obesity: what are the main drivers? *Cancer Causes Control*, 28, 247-258.
- ROSS, L. M., PORTER, R. R. & DURSTINE, J. L. 2016. High-intensity interval training (HIIT) for patients with chronic diseases. *Journal of Sport and Health Science*, 5, 139-144.
- ROTHSCHILD, J., HODDY, K. K., JAMBAZIAN, P. & VARADY, K. A. 2014. Timerestricted feeding and risk of metabolic disease: a review of human and animal studies. *Nutr Rev*, 72, 308-18.
- ROTTIERS, V. & NÄÄR, A. M. 2012. MicroRNAs in metabolism and metabolic disorders. *Nature Reviews Molecular Cell Biology*, 13, 239-250.
- ROY, B., CURTIS, M. E., FEARS, L. S., NAHASHON, S. N. & FENTRESS, H. M. 2016. Molecular Mechanisms of Obesity-Induced Osteoporosis and Muscle Atrophy. *Frontiers in Physiology*, 7, 439.
- RUBERG, F. L., CHEN, Z., HUA, N., BIGORNIA, S., GUO, Z., HALLOCK, K., JARA, H., LAVALLEY, M., PHINIKARIDOU, A., QIAO, Y., VIERECK, J.,

APOVIAN, C. M. & HAMILTON, J. A. 2010. The Relationship of Ectopic Lipid Accumulation to Cardiac and Vascular Function in Obesity and Metabolic Syndrome. *Obesity (Silver Spring, Md.)*, 18, 1116-1121.

- RUDERMAN, N. B., JULIA XU, X., NELSON, L., CACICEDO, J. M., SAHA, A. K., LAN, F. & IDO, Y. 2010. AMPK and SIRT1: a long-standing partnership? *American Journal of Physiology - Endocrinology and Metabolism*, 298, E751-E760.
- RUSSELL, A. P., LAMON, S., BOON, H., WADA, S., GÜLLER, I., BROWN, E. L., CHIBALIN, A. V., ZIERATH, J. R., SNOW, R. J., STEPTO, N., WADLEY, G. D. & AKIMOTO, T. 2013. Regulation of miRNAs in human skeletal muscle following acute endurance exercise and short-term endurance training. *The Journal of Physiology*, 591, 4637-4653.
- RUTKOWSKI, J. M., STERN, J. H. & SCHERER, P. E. 2015. The cell biology of fat expansion. *The Journal of Cell Biology*, 208, 501-512.
- SAFDAR, A., ABADI, A., AKHTAR, M., HETTINGA, B. P. & TARNOPOLSKY, M. A. 2009. miRNA in the regulation of skeletal muscle adaptation to acute endurance exercise in C57Bl/6J male mice. *PLoS One*, 4, e5610.
- SAKURAI, T., ENDO, S., HATANO, D., OGASAWARA, J., KIZAKI, T., OH-ISHI, S., IZAWA, T., ISHIDA, H. & OHNO, H. 2010. Effects of exercise training on adipogenesis of stromal-vascular fraction cells in rat epididymal white adipose tissue. Acta Physiol (Oxf), 200, 325-38.
- SANDERS, MATTHEW J., GRONDIN, PASCAL O., HEGARTY, BRONWYN D., SNOWDEN, MICHAEL A. & CARLING, D. 2007. Investigating the mechanism for AMP activation of the AMP-activated protein kinase cascade. *Biochemical Journal*, 403, 139-148.
- SANTANASTO, A. J., NEWMAN, A. B., STROTMEYER, E. S., BOUDREAU, R. M., GOODPASTER, B. H. & GLYNN, N. W. 2015. Effects of Changes in Regional Body Composition on Physical Function in Older Adults: A Pilot Randomized Controlled Trial. J Nutr Health Aging, 19, 913-21.
- SAPONARO, C., GAGGINI, M., CARLI, F. & GASTALDELLI, A. 2015. The Subtle Balance between Lipolysis and Lipogenesis: A Critical Point in Metabolic Homeostasis. *Nutrients*, 7, 9453-9474.
- SARTOR, F., DE MORREE, H. M., MATSCHKE, V., MARCORA, S. M., MILOUSIS, A., THOM, J. M. & KUBIS, H. P. 2010. High-intensity exercise and carbohydrate-reduced energy-restricted diet in obese individuals. *Eur J Appl Physiol*, 110, 893-903.
- SAWYER, B. J., TUCKER, W. J., BHAMMAR, D. M., RYDER, J. R., SWEAZEA, K. L. & GAESSER, G. A. 2016. Effects of high-intensity interval training and moderate-intensity continuous training on endothelial function and cardiometabolic risk markers in obese adults. *J Appl Physiol (1985)*, 121, 279-88.
- SCHEJA, L., MAKOWSKI, L., UYSAL, K. T., WIESBROCK, S. M., SHIMSHEK, D. R., MEYERS, D. S., MORGAN, M., PARKER, R. A. & HOTAMISLIGIL, G. S. 1999. Altered insulin secretion associated with reduced lipolytic efficiency in aP2-/- mice. *Diabetes*, 48, 1987-94.
- SCHRAUWEN-HINDERLING, V. B., KOOI, M. E., HESSELINK, M. K., JENESON, J. A., BACKES, W. H., VAN ECHTELD, C. J., VAN ENGELSHOVEN, J. M., MENSINK, M. & SCHRAUWEN, P. 2007. Impaired in vivo mitochondrial

function but similar intramyocellular lipid content in patients with type 2 diabetes mellitus and BMI-matched control subjects. *Diabetologia*, 50, 113-20.

- SCHRAUWEN, P., SARIS, W. H. M. & HESSELINK, M. K. C. 2001. An alternative function for human uncoupling protein 3: Protection of mitochondria against accumulation of nonesterified fatty acids inside the mitochondrial matrix. *FASEB Journal*, 15, 2497-2502.
- SCHRAUWEN, P., SCHAART, G., SARIS, W. H., SLIEKER, L. J., GLATZ, J. F., VIDAL, H. & BLAAK, E. E. 2000. The effect of weight reduction on skeletal muscle UCP2 and UCP3 mRNA expression and UCP3 protein content in Type II diabetic subjects. *Diabetologia*, 43, 1408-16.
- SCHREIBER, S. N., KNUTTI, D., BROGLI, K., UHLMANN, T. & KRALLI, A. 2003. The transcriptional coactivator PGC-1 regulates the expression and activity of the orphan nuclear receptor estrogen-related receptor alpha (ERRalpha). *J Biol Chem*, 278, 9013-8.
- SCHULER, M., ALI, F., CHAMBON, C., DUTEIL, D., BORNERT, J.-M., TARDIVEL, A., DESVERGNE, B., WAHLI, W., CHAMBON, P. & METZGER, D. 2006a. PGC1α expression is controlled in skeletal muscles by PPARβ, whose ablation results in fiber-type switching, obesity, and type 2 diabetes. *Cell Metabolism*, 4, 407-414.
- SCHULER, M., ALI, F., CHAMBON, C., DUTEIL, D., BORNERT, J. M., TARDIVEL, A., DESVERGNE, B., WAHLI, W., CHAMBON, P. & METZGER, D. 2006b. PGC1alpha expression is controlled in skeletal muscles by PPARbeta, whose ablation results in fiber-type switching, obesity, and type 2 diabetes. *Cell Metab*, 4, 407-14.
- SCHWARZ, D. S., HUTVAGNER, G., DU, T., XU, Z., ARONIN, N. & ZAMORE, P. D. 2003. Asymmetry in the assembly of the RNAi enzyme complex. *Cell*, 115, 199-208.
- SCOTT, D., TRBOJEVIC, T., SKINNER, E., CLARK, R. A., LEVINGER, P., HAINES, T. P., SANDERS, K. M. & EBELING, P. R. 2015. Associations of calf inter- and intra-muscular adipose tissue with cardiometabolic health and physical function in community-dwelling older adults. *J Musculoskelet Neuronal Interact*, 15, 350-7.
- SCOTT, W., STEVENS, J. & BINDER-MACLEOD, S. A. 2001. Human skeletal muscle fiber type classifications. *Phys Ther*, 81, 1810-6.
- SEIMON, R. V., SHI, Y. C., SLACK, K., LEE, K., FERNANDO, H. A., NGUYEN, A. D., ZHANG, L., LIN, S., ENRIQUEZ, R. F., LAU, J., HERZOG, H. & SAINSBURY, A. 2016. Intermittent Moderate Energy Restriction Improves Weight Loss Efficiency in Diet-Induced Obese Mice. *PLoS One*, 11, e0145157.
- SELDEEN, K. L., LASKY, G., LEIKER, M. M., PANG, M., PERSONIUS, K. E. & TROEN, B. R. 2018. High Intensity Interval Training Improves Physical Performance and Frailty in Aged Mice. J Gerontol A Biol Sci Med Sci, 73, 429-437.
- SELL, H., LAURENCIKIENE, J., TAUBE, A., ECKARDT, K., CRAMER, A., HORRIGHS, A., ARNER, P. & ECKEL, J. 2009. Chemerin is a novel adipocytederived factor inducing insulin resistance in primary human skeletal muscle cells. *Diabetes*, 58, 2731-40.
- SHABAN, N., KENNO, K. A. & MILNE, K. J. 2014. The effects of a 2 week modified high intensity interval training program on the homeostatic model of insulin

resistance (HOMA-IR) in adults with type 2 diabetes. *J Sports Med Phys Fitness*, 54, 203-9.

- SHAW, C. S., CLARK, J. A. & SHEPHERD, S. O. 2013. HSL and ATGL: the movers and shakers of muscle lipolysis. *J Physiol*, 591, 6137-8.
- SHEN, Y., XU, X., YUE, K. & XU, G. 2015. Effect of different exercise protocols on metabolic profiles and fatty acid metabolism in skeletal muscle in high-fat diet-fed rats. *Obesity (Silver Spring)*, 23, 1000-6.
- SHEPHERD, S. O., WILSON, O. J., TAYLOR, A. S., THOGERSEN-NTOUMANI, C., ADLAN, A. M., WAGENMAKERS, A. J. & SHAW, C. S. 2015. Low-Volume High-Intensity Interval Training in a Gym Setting Improves Cardio-Metabolic and Psychological Health. *PLoS One*, 10, e0139056.
- SHI, J. & KANDROR, K. V. 2005. Sortilin is essential and sufficient for the formation of Glut4 storage vesicles in 3T3-L1 adipocytes. *Dev Cell*, 9, 99-108.
- SHIN, K. O., BAE, J. Y., WOO, J., JANG, K. S., KIM, K. S., PARK, J. S., KIM, I. K. & KANG, S. 2015. The effect of exercise on expression of myokine and angiogenesis mRNA in skeletal muscle of high fat diet induced obese rat. *J Exerc Nutrition Biochem*, 19, 91-8.
- SHOELSON, S. E., LEE, J. & GOLDFINE, A. B. 2006. Inflammation and insulin resistance. *J Clin Invest*, 116, 1793-801.
- SHOJAIE, M., GHANBARI, F. & SHOJAIE, N. 2017. Intermittent fasting could ameliorate cognitive function against distress by regulation of inflammatory response pathway. *J Adv Res*, 8, 697-701.
- SHORTREED, K. E., KRAUSE, M. P., HUANG, J. H., DHANANI, D., MORADI, J., CEDDIA, R. B. & HAWKE, T. J. 2009. Muscle-specific adaptations, impaired oxidative capacity and maintenance of contractile function characterize dietinduced obese mouse skeletal muscle. *PLoS One*, 4, e7293.
- SHULMAN, G. I. 2000. Cellular mechanisms of insulin resistance. *Journal of Clinical Investigation*, 106, 171-176.
- SIJIE, T., HAINAI, Y., FENGYING, Y. & JIANXIONG, W. 2012. High intensity interval exercise training in overweight young women. *J Sports Med Phys Fitness*, 52, 255-62.
- SILVA, G. J. J., BYE, A., EL AZZOUZI, H. & WISLØFF, U. 2017. MicroRNAs as Important Regulators of Exercise Adaptation. *Progress in Cardiovascular Diseases*, 60, 130-151.
- SIM, A. Y., WALLMAN, K. E., FAIRCHILD, T. J. & GUELFI, K. J. 2015. Effects of High-Intensity Intermittent Exercise Training on Appetite Regulation. *Med Sci* Sports Exerc, 47, 2441-9.
- SIMONEAU, J. A. & BOUCHARD, C. 1995. Genetic determinism of fiber type proportion in human skeletal muscle. *Faseb j*, 9, 1091-5.
- SISHI, B., LOOS, B., ELLIS, B., SMITH, W., DU TOIT, E. F. & ENGELBRECHT, A. M. 2011. Diet-induced obesity alters signalling pathways and induces atrophy and apoptosis in skeletal muscle in a prediabetic rat model. *Exp Physiol*, 96, 179-93.
- SKURK, T., ALBERTI-HUBER, C., HERDER, C. & HAUNER, H. 2007. Relationship between adipocyte size and adipokine expression and secretion. *J Clin Endocrinol Metab*, 92, 1023-33.
- SMIRNOVA, E., GOLDBERG, E. B., MAKAROVA, K. S., LIN, L., BROWN, W. J. & JACKSON, C. L. 2006. ATGL has a key role in lipid droplet/adiposome degradation in mammalian cells. *EMBO Rep*, 7, 106-13.

- SOETERS, M. R., LAMMERS, N. M., DUBBELHUIS, P. F., ACKERMANS, M., JONKERS-SCHUITEMA, C. F., FLIERS, E., SAUERWEIN, H. P., AERTS, J. M. & SERLIE, M. J. 2009. Intermittent fasting does not affect whole-body glucose, lipid, or protein metabolism. *Am J Clin Nutr*, 90, 1244-51.
- SOOD, P., KREK, A., ZAVOLAN, M., MACINO, G. & RAJEWSKY, N. 2006. Celltype-specific signatures of microRNAs on target mRNA expression. *Proc Natl Acad Sci U S A*, 103, 2746-51.
- SPIEGELMAN, B. M., FRANK, M. & GREEN, H. 1983. Molecular cloning of mRNA from 3T3 adipocytes. Regulation of mRNA content for glycerophosphate dehydrogenase and other differentiation-dependent proteins during adipocyte development. J Biol Chem, 258, 10083-9.
- ST-PIERRE, J., LIN, J., KRAUSS, S., TARR, P. T., YANG, R., NEWGARD, C. B. & SPIEGELMAN, B. M. 2003. Bioenergetic analysis of peroxisome proliferatoractivated receptor gamma coactivators 1alpha and 1beta (PGC-1alpha and PGC-1beta) in muscle cells. *J Biol Chem*, 278, 26597-603.
- STANFORD, K. I., MIDDELBEEK, R. J., TOWNSEND, K. L., LEE, M. Y., TAKAHASHI, H., SO, K., HITCHCOX, K. M., MARKAN, K. R., HELLBACH, K., HIRSHMAN, M. F., TSENG, Y. H. & GOODYEAR, L. J. 2015. A novel role for subcutaneous adipose tissue in exercise-induced improvements in glucose homeostasis. *Diabetes*, 64, 2002-14.
- STEFANOWICZ, M., NIKOŁAJUK, A., MATULEWICZ, N. & KARCZEWSKA-KUPCZEWSKA, M. 2018. Adipose tissue, but not skeletal muscle, sirtuin 1 expression is decreased in obesity and related to insulin sensitivity. *Endocrine*, 60, 263-271.
- STEPHENS, F. B., CONSTANTIN-TEODOSIU, D. & GREENHAFF, P. L. 2007. New insights concerning the role of carnitine in the regulation of fuel metabolism in skeletal muscle. *J Physiol*, 581, 431-44.
- STEPTO, N. K., BENZIANE, B., WADLEY, G. D., CHIBALIN, A. V., CANNY, B. J., EYNON, N. & MCCONELL, G. K. 2012. Short-Term Intensified Cycle Training Alters Acute and Chronic Responses of PGC1α and Cytochrome C Oxidase IV to Exercise in Human Skeletal Muscle. *PLOS ONE*, 7, e53080.
- STERN, J. H., RUTKOWSKI, J. M. & SCHERER, P. E. 2016. Adiponectin, Leptin, and Fatty Acids in the Maintenance of Metabolic Homeostasis through Adipose Tissue Crosstalk. *Cell Metab*, 23, 770-84.
- STORLIEN, L., OAKES, N. D. & KELLEY, D. E. 2004. Metabolic flexibility. *Proceedings of the Nutrition Society*, 63, 363-368.
- STRASSER, B., SPREITZER, A. & HABER, P. 2007. Fat loss depends on energy deficit only, independently of the method for weight loss. *Ann Nutr Metab*, 51, 428-32.
- STUART, C. A., SOUTH, M. A., LEE, M. L., MCCURRY, M. P., HOWELL, M. E., RAMSEY, M. W. & STONE, M. H. 2013. Insulin responsiveness in metabolic syndrome after eight weeks of cycle training. *Med Sci Sports Exerc*, 45, 2021-9.
- SUN, K., HALBERG, N., KHAN, M., MAGALANG, U. J. & SCHERER, P. E. 2013a. Selective inhibition of hypoxia-inducible factor 1alpha ameliorates adipose tissue dysfunction. *Mol Cell Biol*, 33, 904-17.
- SUN, K., KUSMINSKI, C. M. & SCHERER, P. E. 2011. Adipose tissue remodeling and obesity. *J Clin Invest*, 121, 2094-101.
- SUN, K. & SCHERER, P. E. 2010. Adipose Tissue Dysfunction: A Multistep Process. *In:* CHRISTEN, Y., CLÉMENT, K. & SPIEGELMAN, M. B. (eds.) *Novel*

Insights into Adipose Cell Functions. Berlin, Heidelberg: Springer Berlin Heidelberg.

- SUN, K., TORDJMAN, J., CLÉMENT, K. & SCHERER, PHILIPP E. 2013b. Fibrosis and Adipose Tissue Dysfunction. *Cell Metabolism*, 18, 470-477.
- SUWA, M., NAKANO, H., RADAK, Z. & KUMAGAI, S. 2008. Endurance exercise increases the SIRT1 and peroxisome proliferator-activated receptor gamma coactivator-1alpha protein expressions in rat skeletal muscle. *Metabolism*, 57, 986-98.
- SWIERCZYNSKI, J., GOYKE, E., WACH, L., PANKIEWICZ, A., KOCHAN, Z., ADAMONIS, W., SLEDZINSKI, Z. & ALEKSANDROWICZ, Z. 2000. Comparative study of the lipogenic potential of human and rat adipose tissue. *Metabolism*, 49, 594-9.
- SWINBURN, B. A. 2008. Obesity prevention: the role of policies, laws and regulations. *Australia and New Zealand Health Policy*, 5, 12.
- SZTALRYD, C., XU, G., DORWARD, H., TANSEY, J. T., CONTRERAS, J. A., KIMMEL, A. R. & LONDOS, C. 2003. Perilipin A is essential for the translocation of hormone-sensitive lipase during lipolytic activation. *J Cell Biol*, 161, 1093-103.
- SZYMANSKI, K. M., BINNS, D., BARTZ, R., GRISHIN, N. V., LI, W. P., AGARWAL, A. K., GARG, A., ANDERSON, R. G. & GOODMAN, J. M. 2007. The lipodystrophy protein seipin is found at endoplasmic reticulum lipid droplet junctions and is important for droplet morphology. *Proc Natl Acad Sci U S A*, 104, 20890-5.
- TAKANABE, R., ABE, Y., TAKAYA, T., WADA, H., HASEGAWA, K., SATOH, N., SHIMATSU, A., ONO, K., HORIE, T. & KITA, T. 2008. Up-regulated expression of microRNA-143 in association with obesity in adipose tissue of mice fed high-fat diet. *Biochemical and Biophysical Research Communications*, 376, 728-732.
- TALBOT, J. & MAVES, L. 2016. Skeletal muscle fiber type: using insights from muscle developmental biology to dissect targets for susceptibility and resistance to muscle disease. *Wiley Interdiscip Rev Dev Biol*, 5, 518-34.
- TAM, C. S., COVINGTON, J. D., BAJPEYI, S., TCHOUKALOVA, Y., BURK, D., JOHANNSEN, D. L., ZINGARETTI, C. M., CINTI, S. & RAVUSSIN, E. 2014. Weight gain reveals dramatic increases in skeletal muscle extracellular matrix remodeling. J Clin Endocrinol Metab, 99, 1749-57.
- TAM, C. S., POWER, J. E., MARKOVIC, T. P., YEE, C., MORSCH, M., MCLENNAN, S. V. & TWIGG, S. M. 2015. The effects of high-fat feeding on physical function and skeletal muscle extracellular matrix. *Nutrition & Diabetes*, 5, e187.
- TAM, C. S., SPARKS, L. M., JOHANNSEN, D. L., COVINGTON, J. D., CHURCH, T. S. & RAVUSSIN, E. 2012. Low macrophage accumulation in skeletal muscle of obese type 2 diabetics and elderly subjects. *Obesity (Silver Spring)*, 20, 1530-3.
- TANNER, C. J., BARAKAT, H. A., DOHM, G. L., PORIES, W. J., MACDONALD, K.
  G., CUNNINGHAM, P. R., SWANSON, M. S. & HOUMARD, J. A. 2002.
  Muscle fiber type is associated with obesity and weight loss. *Am J Physiol Endocrinol Metab*, 282, E1191-6.
- TANSEY, J. T., SZTALRYD, C., GRUIA-GRAY, J., ROUSH, D. L., ZEE, J. V., GAVRILOVA, O., REITMAN, M. L., DENG, C. X., LI, C., KIMMEL, A. R. & LONDOS, C. 2001. Perilipin ablation results in a lean mouse with aberrant

adipocyte lipolysis, enhanced leptin production, and resistance to diet-induced obesity. *Proc Natl Acad Sci U S A*, 98, 6494-9.

- TAO, R., WEI, D., GAO, H., LIU, Y., DEPINHO, R. A. & DONG, X. C. 2011. Hepatic FoxOs regulate lipid metabolism via modulation of expression of the nicotinamide phosphoribosyltransferase gene. J Biol Chem, 286, 14681-90.
- TARIQ, H., NAYUDU, S., AKELLA, S., GLANDT, M. & CHILIMURI, S. 2016. Non-Alcoholic Fatty Pancreatic Disease: A Review of Literature. *Gastroenterology Research*, 9, 87-91.
- TERADA, S. & TABATA, I. 2004. Effects of acute bouts of running and swimming exercise on PGC-1alpha protein expression in rat epitrochlearis and soleus muscle. *Am J Physiol Endocrinol Metab*, 286, E208-16.
- THIAM, A. R., FARESE, R. V. & WALTHER, T. C. 2013. The Biophysics and Cell Biology of Lipid Droplets. *Nature reviews. Molecular cell biology*, 14, 775-786.
- THOMAS, J. A., 2ND, ANTONELLI, J. A., LLOYD, J. C., MASKO, E. M., POULTON, S. H., PHILLIPS, T. E., POLLAK, M. & FREEDLAND, S. J. 2010. Effect of intermittent fasting on prostate cancer tumor growth in a mouse model. *Prostate Cancer Prostatic Dis*, 13, 350-5.
- TINSLEY, G. M. & LA BOUNTY, P. M. 2015. Effects of intermittent fasting on body composition and clinical health markers in humans. *Nutrition Reviews*, 73, 661-674.
- TJONNA, A. E., LEE, S. J., ROGNMO, O., STOLEN, T. O., BYE, A., HARAM, P. M., LOENNECHEN, J. P., AL-SHARE, Q. Y., SKOGVOLL, E., SLORDAHL, S. A., KEMI, O. J., NAJJAR, S. M. & WISLOFF, U. 2008. Aerobic interval training versus continuous moderate exercise as a treatment for the metabolic syndrome: a pilot study. *Circulation*, 118, 346-54.
- TJONNA, A. E., LEINAN, I. M., BARTNES, A. T., JENSSEN, B. M., GIBALA, M. J., WINETT, R. A. & WISLOFF, U. 2013. Low- and high-volume of intensive endurance training significantly improves maximal oxygen uptake after 10-weeks of training in healthy men. *PLoS One*, 8, e65382.
- TJONNA, A. E., ROGNMO, O., BYE, A., STOLEN, T. O. & WISLOFF, U. 2011. Time course of endothelial adaptation after acute and chronic exercise in patients with metabolic syndrome. *J Strength Cond Res*, 25, 2552-8.
- TOGASHI, N., URA, N., HIGASHIURA, K., MURAKAMI, H. & SHIMAMOTO, K. 2000. The contribution of skeletal muscle tumor necrosis factor-alpha to insulin resistance and hypertension in fructose-fed rats. *J Hypertens*, 18, 1605-10.
- TOLEDO, F. G., MENSHIKOVA, E. V., AZUMA, K., RADIKOVA, Z., KELLEY, C. A., RITOV, V. B. & KELLEY, D. E. 2008. Mitochondrial capacity in skeletal muscle is not stimulated by weight loss despite increases in insulin action and decreases in intramyocellular lipid content. *Diabetes*, 57, 987-94.
- TONTONOZ, P., HU, E. & SPIEGELMAN, B. M. 1994. Stimulation of adipogenesis in fibroblasts by PPAR gamma 2, a lipid-activated transcription factor. *Cell*, 79, 1147-56.
- TRAJKOVSKI, M., HAUSSER, J., SOUTSCHEK, J., BHAT, B., AKIN, A., ZAVOLAN, M., HEIM, M. H. & STOFFEL, M. 2011. MicroRNAs 103 and 107 regulate insulin sensitivity. *Nature*, 474, 649-53.
- TRAPP, E. G., CHISHOLM, D. J., FREUND, J. & BOUTCHER, S. H. 2008. The effects of high-intensity intermittent exercise training on fat loss and fasting insulin levels of young women. *Int J Obes*, 32, 684-691.

- TRAYHURN, P. 2013. Hypoxia and adipose tissue function and dysfunction in obesity. *Physiological Reviews*, 93, 1-21.
- TSILOULIS, T., PIKE, J., POWELL, D., ROSSELLO, F. J., CANNY, B. J., MEEX, R. C. R. & WATT, M. J. 2016. Impact of endurance exercise training on adipocyte microRNA expression in overweight men. *The FASEB Journal*, 31, 161-171.
- TSILOULIS, T. & WATT, M. J. 2015. Exercise and the Regulation of Adipose Tissue Metabolism. *Prog Mol Biol Transl Sci*, 135, 175-201.
- TUAZON, M. A., MCCONNELL, T. R., WILSON, G. J., ANTHONY, T. G. & HENDERSON, G. C. 2015. Intensity-dependent and sex-specific alterations in hepatic triglyceride metabolism in mice following acute exercise. *J Appl Physiol* (1985), 118, 61-70.
- TURER, A. T., KHERA, A., AYERS, C. R., TURER, C. B., GRUNDY, S. M., VEGA, G. L. & SCHERER, P. E. 2011. Adipose tissue mass and location affect circulating adiponectin levels. *Diabetologia*, 54, 2515-24.
- TÜRK, Y., THEEL, W., KASTELEYN, M. J., FRANSSEN, F. M. E., HIEMSTRA, P. S., RUDOLPHUS, A., TAUBE, C. & BRAUNSTAHL, G. J. 2017. High intensity training in obesity: a Meta-analysis. *Obesity Science & Practice*, *3*, 258-271.
- URANGA, A. P., LEVINE, J. & JENSEN, M. 2005. Isotope tracer measures of meal fatty acid metabolism: reproducibility and effects of the menstrual cycle. *Am J Physiol Endocrinol Metab*, 288, E547-55.
- UYSAL, K. T., SCHEJA, L., WIESBROCK, S. M., BONNER-WEIR, S. & HOTAMISLIGIL, G. S. 2000. Improved glucose and lipid metabolism in genetically obese mice lacking aP2. *Endocrinology*, 141, 3388-96.
- VALENCAK, T. G., OSTERRIEDER, A. & SCHULZ, T. J. 2017. Sex matters: The effects of biological sex on adipose tissue biology and energy metabolism. *Redox Biology*, 12, 806-813.
- VALENCIA-SANCHEZ, M. A., LIU, J., HANNON, G. J. & PARKER, R. 2006. Control of translation and mRNA degradation by miRNAs and siRNAs. *Genes Dev*, 20, 515-24.
- VAN DER HEIJDEN, R. A., SHEEDFAR, F., MORRISON, M. C., HOMMELBERG,
  P. P. H., KOR, D., KLOOSTERHUIS, N. J., GRUBEN, N., YOUSSEF, S. A., DE
  BRUIN, A., HOFKER, M. H., KLEEMANN, R., KOONEN, D. P. Y. &
  HEERINGA, P. 2015. High-fat diet induced obesity primes inflammation in
  adipose tissue prior to liver in C57BL/6j mice. *Aging (Albany NY)*, 7, 256-267.
- VAN DIJK, S. J., TELLAM, R. L., MORRISON, J. L., MUHLHAUSLER, B. S. & MOLLOY, P. L. 2015. Recent developments on the role of epigenetics in obesity and metabolic disease. *Clinical Epigenetics*, 7, 66.
- VAN HOVE, E. C., HANSEN, T., DEKKER, J. M., REILING, E., NIJPELS, G., JØRGENSEN, T., BORCH-JOHNSEN, K., HAMID, Y. H., HEINE, R. J., PEDERSEN, O., MAASSEN, J. A. & T HART, L. M. 2006. The HADHSC Gene Encoding Short-Chain 1-3-Hydroxyacyl-CoA Dehydrogenase (SCHAD) and Type 2 Diabetes Susceptibility. *Diabetes*, 55, 3193-3196.
- VAN VLIET-OSTAPTCHOUK, J. V., SNIEDER, H. & LAGOU, V. 2012. Gene-Lifestyle Interactions in Obesity. *Current Nutrition Reports*, 1, 184-196.
- VARADY, K. A., ALLISTER, C. A., ROOHK, D. J. & HELLERSTEIN, M. K. 2010. Improvements in body fat distribution and circulating adiponectin by alternateday fasting versus calorie restriction. *The Journal of Nutritional Biochemistry*, 21, 188-195.

- VARADY, K. A., BHUTANI, S., CHURCH, E. C. & KLEMPEL, M. C. 2009. Shortterm modified alternate-day fasting: a novel dietary strategy for weight loss and cardioprotection in obese adults. *Am J Clin Nutr*, 90, 1138-43.
- VARADY, K. A., BHUTANI, S., KLEMPEL, M. C., KROEGER, C. M., TREPANOWSKI, J. F., HAUS, J. M., HODDY, K. K. & CALVO, Y. 2013. Alternate day fasting for weight loss in normal weight and overweight subjects: a randomized controlled trial. *Nutr J*, 12, 146.
- VARADY, K. A., ROOHK, D. J., LOE, Y. C., MCEVOY-HEIN, B. K. & HELLERSTEIN, M. K. 2007. Effects of modified alternate-day fasting regimens on adipocyte size, triglyceride metabolism, and plasma adiponectin levels in mice. *J Lipid Res*, 48, 2212-9.
- VAZQUEZ-VELA, M. E., TORRES, N. & TOVAR, A. R. 2008. White adipose tissue as endocrine organ and its role in obesity. *Arch Med Res*, 39, 715-28.
- VEERKAMP, J. H. & VAN MOERKERK, H. T. 1993. Fatty acid-binding protein and its relation to fatty acid oxidation. *Mol Cell Biochem*, 123, 101-6.
- VEGA, R. B., HUSS, J. M. & KELLY, D. P. 2000. The coactivator PGC-1 cooperates with peroxisome proliferator-activated receptor alpha in transcriptional control of nuclear genes encoding mitochondrial fatty acid oxidation enzymes. *Mol Cell Biol*, 20, 1868-76.
- VERHOEF, S. P. M., CAMPS, S. G. J. A., BOUWMAN, F. G., MARIMAN, E. C. M. & WESTERTERP, K. R. 2013. Physiological Response of Adipocytes to Weight Loss and Maintenance. *PLoS ONE*, 8, e58011.
- VIDAL, H., AUBOEUF, D., DE VOS, P., STAELS, B., RIOU, J. P., AUWERX, J. & LAVILLE, M. 1996. The expression of ob gene is not acutely regulated by insulin and fasting in human abdominal subcutaneous adipose tissue. *J Clin Invest*, 98, 251-5.
- VIGELSØ, A., ANDERSEN, N. B. & DELA, F. 2014. The relationship between skeletal muscle mitochondrial citrate synthase activity and whole body oxygen uptake adaptations in response to exercise training. *International Journal of Physiology, Pathophysiology and Pharmacology,* **6**, 84-101.
- VOGIATZIS, I., TERZIS, G., NANAS, S., STRATAKOS, G., SIMOES, D. C., GEORGIADOU, O., ZAKYNTHINOS, S. & ROUSSOS, C. 2005. Skeletal muscle adaptations to interval training in patients with advanced COPD. *Chest*, 128, 3838-45.
- VONG, L., YE, C., YANG, Z., CHOI, B., CHUA, S., JR. & LOWELL, B. B. 2011. Leptin action on GABAergic neurons prevents obesity and reduces inhibitory tone to POMC neurons. *Neuron*, 71, 142-54.
- WANG, B. & TRAYHURN, P. 2006. Acute and prolonged effects of TNF-alpha on the expression and secretion of inflammation-related adipokines by human adipocytes differentiated in culture. *Pflugers Arch*, 452, 418-27.
- WANG, H. & YE, J. 2015. Regulation of Energy Balance by Inflammation: Common Theme in Physiology and Pathology. *Reviews in endocrine & metabolic disorders*, 16, 47-54.
- WANG, N., LIU, Y., MA, Y. & WEN, D. 2017. High-intensity interval versus moderateintensity continuous training: Superior metabolic benefits in diet-induced obesity mice. *Life Sciences*, 191, 122-131.
- WANG, S.-X., WEI, J.-G., CHEN, L.-L., HU, X. & KONG, W. 2016. The role of expression imbalance between adipose synthesis and storage mediated by PPAR-

 $\gamma$ /FSP27 in the formation of insulin resistance in catch up growth. *Lipids in Health and Disease*, 15, 173.

- WANG, X. L., ZHANG, L., YOUKER, K., ZHANG, M. X., WANG, J., LEMAIRE, S. A., COSELLI, J. S. & SHEN, Y. H. 2006. Free fatty acids inhibit insulin signaling-stimulated endothelial nitric oxide synthase activation through upregulating PTEN or inhibiting Akt kinase. *Diabetes*, 55, 2301-10.
- WANG, Y. X., LEE, C. H., TIEP, S., YU, R. T., HAM, J., KANG, H. & EVANS, R. M. 2003. Peroxisome-proliferator-activated receptor delta activates fat metabolism to prevent obesity. *Cell*, 113, 159-70.
- WANG, Y. X., ZHANG, C. L., YU, R. T., CHO, H. K., NELSON, M. C., BAYUGA-OCAMPO, C. R., HAM, J., KANG, H. & EVANS, R. M. 2004. Regulation of muscle fiber type and running endurance by PPARdelta. *PLoS Biol*, 2, e294.
- WARFEL, J. D., BERMUDEZ, E. M., MENDOZA, T. M., GHOSH, S., ZHANG, J., ELKS, C. M., MYNATT, R. & VANDANMAGSAR, B. 2016. Mitochondrial fat oxidation is essential for lipid-induced inflammation in skeletal muscle in mice. 6, 37941.
- WEISBERG, S. P., MCCANN, D., DESAI, M., ROSENBAUM, M., LEIBEL, R. L. & FERRANTE, A. W., JR. 2003. Obesity is associated with macrophage accumulation in adipose tissue. *J Clin Invest*, 112, 1796-808.
- WEISS, E. P., ALBERT, S. G., REEDS, D. N., KRESS, K. S., EZEKIEL, U. R., MCDANIEL, J. L., PATTERSON, B. W., KLEIN, S. & VILLAREAL, D. T. 2015. Calorie Restriction and Matched Weight Loss From Exercise: Independent and Additive Effects on Glucoregulation and the Incretin System in Overweight Women and Men. *Diabetes Care*, 38, 1253-62.
- WELLS, G. D., NOSEWORTHY, M. D., HAMILTON, J., TARNOPOLSKI, M. & TEIN, I. 2014. Skeletal Muscle Metabolic Dysfunction in Obesity and Metabolic Syndrome. *Canadian Journal of Neurological Sciences / Journal Canadien des Sciences Neurologiques*, 35, 31-40.
- WENDE, A. R., SCHAEFFER, P. J., PARKER, G. J., ZECHNER, C., HAN, D. H., CHEN, M. M., HANCOCK, C. R., LEHMAN, J. J., HUSS, J. M., MCCLAIN, D. A., HOLLOSZY, J. O. & KELLY, D. P. 2007. A role for the transcriptional coactivator PGC-1alpha in muscle refueling. *J Biol Chem*, 282, 36642-51.
- WEWEGE, M., VAN DEN BERG, R., WARD, R. E. & KEECH, A. 2017. The effects of high-intensity interval training vs. moderate-intensity continuous training on body composition in overweight and obese adults: a systematic review and metaanalysis. *Obesity Reviews*, 18, 635-646.
- WHITE, M. F. 2002. IRS proteins and the common path to diabetes. Am J Physiol Endocrinol Metab, 283, E413-22.
- WHYTE, L. J., FERGUSON, C., WILSON, J., SCOTT, R. A. & GILL, J. M. 2013. Effects of single bout of very high-intensity exercise on metabolic health biomarkers in overweight/obese sedentary men. *Metabolism*, 62, 212-9.
- WILLIAMS, R. L., WOOD, L. G., COLLINS, C. E. & CALLISTER, R. 2015. Effectiveness of weight loss interventions – is there a difference between men and women: a systematic review. *Obesity Reviews*, 16, 171-186.
- WILSON, R. A., DEASY, W., HAYES, A. & COOKE, M. B. 2017. High fat diet and associated changes in the expression of micro-RNAs in tissue: Lessons learned from animal studies. *Mol Nutr Food Res*, 61.

- WILSON, R. A., DEASY, W., STATHIS, C. G., HAYES, A. & COOKE, M. B. 2018. Intermittent Fasting with or without Exercise Prevents Weight Gain and Improves Lipids in Diet-Induced Obese Mice. *Nutrients*, 10, E346.
- WINDER, W. W. & HARDIE, D. G. 1996. Inactivation of acetyl-CoA carboxylase and activation of AMP-activated protein kinase in muscle during exercise. *Am J Physiol*, 270, E299-304.
- WOJTASZEWSKI, J. F., NIELSEN, P., HANSEN, B. F., RICHTER, E. A. & KIENS, B. 2000. Isoform-specific and exercise intensity-dependent activation of 5'-AMPactivated protein kinase in human skeletal muscle. *J Physiol*, 528 Pt 1, 221-6.
- WORLD HEALTH ORGANIZATION 2010. WHO Global Recommendations on Physical Activity for Health. Geneva. *World Health Organization*.
- WRIGHT, S. & ARONNE, L. 2012. Causes of obesity. Abdominal Imaging, 37, 730-732.
- WU, H. & BALLANTYNE, C. M. 2017. Skeletal muscle inflammation and insulin resistance in obesity. *J Clin Invest*, 127, 43-54.
- WU, H., KANATOUS, S. B., THURMOND, F. A., GALLARDO, T., ISOTANI, E., BASSEL-DUBY, R. & WILLIAMS, R. S. 2002. Regulation of mitochondrial biogenesis in skeletal muscle by CaMK. *Science*, 296, 349-52.
- WU, L., FAN, J. & BELASCO, J. G. 2006. MicroRNAs direct rapid deadenylation of mRNA. *Proc Natl Acad Sci U S A*, 103, 4034-9.
- WU, Z., PUIGSERVER, P., ANDERSSON, U., ZHANG, C., ADELMANT, G., MOOTHA, V., TROY, A., CINTI, S., LOWELL, B., SCARPULLA, R. C. & SPIEGELMAN, B. M. 1999. Mechanisms controlling mitochondrial biogenesis and respiration through the thermogenic coactivator PGC-1. *Cell*, 98, 115-24.
- XIE, H., LIM, B. & LODISH, H. F. 2009a. MicroRNAs induced during adipogenesis that accelerate fat cell development are downregulated in obesity. *Diabetes*, 58, 1050-1057.
- XIE, H., SUN, L. & LODISH, H. F. 2009b. Targeting microRNAs in obesity. *Expert Opin Ther Targets*, 13, 1227-38.
- XU, H., SETHI, J. K. & HOTAMISLIGIL, G. S. 1999. Transmembrane tumor necrosis factor (TNF)-alpha inhibits adipocyte differentiation by selectively activating TNF receptor 1. J Biol Chem, 274, 26287-95.
- YAMAMOTO, J., KAMATA, S., MIURA, A., NAGATA, T., KAINUMA, R. & ISHII, I. 2015. Differential adaptive responses to 1- or 2-day fasting in various mouse tissues revealed by quantitative PCR analysis. *FEBS Open Bio*, 5, 357-368.
- YAMAUCHI, T., KAMON, J., MINOKOSHI, Y., ITO, Y., WAKI, H., UCHIDA, S., YAMASHITA, S., NODA, M., KITA, S., UEKI, K., ETO, K., AKANUMA, Y., FROGUEL, P., FOUFELLE, F., FERRE, P., CARLING, D., KIMURA, S., NAGAI, R., KAHN, B. B. & KADOWAKI, T. 2002. Adiponectin stimulates glucose utilization and fatty-acid oxidation by activating AMP-activated protein kinase. *Nat Med*, 8, 1288-95.
- YANG, Z. H., MIYAHARA, H., IWASAKI, Y., TAKEO, J. & KATAYAMA, M. 2013. Dietary supplementation with long-chain monounsaturated fatty acids attenuates obesity-related metabolic dysfunction and increases expression of PPAR gamma in adipose tissue in type 2 diabetic KK-Ay mice. *Nutr Metab (Lond)*, 10, 16.
- YATAGAI, T., NAGASAKA, S., TANIGUCHI, A., FUKUSHIMA, M., NAKAMURA, T., KUROE, A., NAKAI, Y. & ISHIBASHI, S. 2003. Hypoadiponectinemia is associated with visceral fat accumulation and insulin resistance in Japanese men with type 2 diabetes mellitus. *Metabolism*, 52, 1274-8.

- YE, J., GAO, Z., YIN, J. & HE, Q. 2007. Hypoxia is a potential risk factor for chronic inflammation and adiponectin reduction in adipose tissue of ob/ob and dietary obese mice. *Am J Physiol Endocrinol Metab*, 293, E1118-28.
- YE, L., KLEINER, S., WU, J., SAH, R., GUPTA, R. K., BANKS, A. S., COHEN, P., KHANDEKAR, M. J., BOSTROM, P., MEPANI, R. J., LAZNIK, D., KAMENECKA, T. M., SONG, X., LIEDTKE, W., MOOTHA, V. K., PUIGSERVER, P., GRIFFIN, P. R., CLAPHAM, D. E. & SPIEGELMAN, B. M. 2012. TRPV4 is a regulator of adipose oxidative metabolism, inflammation, and energy homeostasis. *Cell*, 151, 96-110.
- YEOM, K. H., LEE, Y., HAN, J., SUH, M. R. & KIM, V. N. 2006. Characterization of DGCR8/Pasha, the essential cofactor for Drosha in primary miRNA processing. *Nucleic Acids Res*, 34, 4622-9.
- YOSHINO, J., MILLS, K. F., YOON, M. J. & IMAI, S. 2011. Nicotinamide mononucleotide, a key NAD(+) intermediate, treats the pathophysiology of dietand age-induced diabetes in mice. *Cell Metab*, 14, 528-36.
- YUN, Z., MAECKER, H. L., JOHNSON, R. S. & GIACCIA, A. J. 2002. Inhibition of PPAR gamma 2 gene expression by the HIF-1-regulated gene DEC1/Stra13: a mechanism for regulation of adipogenesis by hypoxia. *Dev Cell*, 2, 331-41.
- ZHAI, W., XU, C., LING, Y., LIU, S., DENG, J., QI, Y., LONDOS, C. & XU, G. 2010. Increased lipolysis in adipose tissues is associated with elevation of systemic free fatty acids and insulin resistance in perilipin null mice. *Horm Metab Res*, 42, 247-53.
- ZHANG, H., K TONG, T., QIU, W., WANG, J., NIE, J. & HE, Y. 2015. Effect of highintensity interval training protocol on abdominal fat reduction in overweight Chinese women: a randomized controlled trial. *Kineziologija*, 47, 57-66.
- ZHANG, L., MIYAKI, K., NAKAYAMA, T. & MURAMATSU, M. 2008. Cell deathinducing DNA fragmentation factor alpha-like effector A (CIDEA) gene V115F (G-->T) polymorphism is associated with phenotypes of metabolic syndrome in Japanese men. *Metabolism*, 57, 502-5.
- ZHANG, Y., PROENCA, R., MAFFEI, M., BARONE, M., LEOPOLD, L. & FRIEDMAN, J. M. 1994. Positional cloning of the mouse obese gene and its human homologue. *Nature*, 372, 425-32.
- ZHU, H., SHYH-CHANG, N., SEGRE, A. V., SHINODA, G., SHAH, S. P., EINHORN, W. S., TAKEUCHI, A., ENGREITZ, J. M., HAGAN, J. P., KHARAS, M. G., URBACH, A., THORNTON, J. E., TRIBOULET, R., GREGORY, R. I., ALTSHULER, D. & DALEY, G. Q. 2011. The Lin28/let-7 axis regulates glucose metabolism. *Cell*, 147, 81-94.
- ZIERATH, J. R. & HAWLEY, J. A. 2004. Skeletal muscle fiber type: influence on contractile and metabolic properties. *PLoS Biol*, 2, e348.
- ZIMMER, J. S., DYCKES, D. F., BERNLOHR, D. A. & MURPHY, R. C. 2004. Fatty acid binding proteins stabilize leukotriene A4: competition with arachidonic acid but not other lipoxygenase products. *J Lipid Res*, 45, 2138-44.